UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 8-K

CURRENT REPORT

Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report (Date of earliest event reported): May 25, 2020

AdaptHealth Corp.

(Exact name of registrant as specified in its charter)

Delaware (State or other jurisdiction of incorporation) **001-38399** (Commission File Number)

82-3677704 (IRS Employer Identification No.)

220 West Germantown Pike, Suite 250

Plymouth Meeting, PA (address of principal executive offices) 19462 (zip code)

(610) 630-6357 (Registrant's telephone number, including area code)

(Former name or former address, if changed since last report.)

Not Applicable (Former name or former address, if changed since last report).

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions:

□ Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)

□ Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)

□ Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))

□ Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

Securities registered pursuant to Section 12(b) of the Act:

Title of each class	Trading Symbol(s)	Name of each exchange on which registered
Class A Common Stock, par value \$0.0001 per share	АНСО	The Nasdaq Stock Market LLC

Indicate by check mark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Act of 1933 (§ 230.405) or Rule 12b-2 of the Securities Exchange Act of 1934 (§ 240.12b-2).

Emerging growth company \square

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act.

Item 1.01. Entry Into A Material Definitive Agreement.

Stock Purchase Agreement and Agreement and Plan of Merger

On May 25, 2020, AdaptHealth Corp., a Delaware corporation (the "Company"), AdaptHealth LLC, a Delaware limited liability company and an indirect subsidiary of the Company ("Buyer"), Eleanor Merger Sub, LLC, a Delaware limited liability company and a wholly-owned subsidiary of Buyer ("Merger Sub"), Solara Holdings, LLC, a Delaware limited liability company ("Solara"), and LCP Solara Blocker Seller, LLC, a Delaware limited liability company ("Blocker Seller"), in its capacity as Blocker Seller and as Representative (as defined in the Purchase Agreement), entered into a Stock Purchase Agreement and Agreement and Plan of Merger (the "Purchase Agreement") pursuant to which Blocker Seller agreed to sell to the Company, and the Company agreed to purchase from Blocker Seller, all of the issued and outstanding equity interests in LCP Solara Blocker Corp., a Delaware corporation ("Blocker"), and Merger Sub agreed to be merged with and into Solara, resulting in each of Blocker and Solara becoming subsidiaries of the Company (the "Acquisition"), subject to the satisfaction or waiver of certain conditions as further discussed below.

The base purchase price for the Acquisition is \$425,000,000, consisting of a combination of cash, and 3,906,250 shares of Class A Common Stock, par value \$0.0001 per share ("Class A Common Stock"), of the Company valued at \$16 consisting of a combination of cash per share, payable at the closing, subject to customary adjustments to the cash portion of such payment for cash, indebtedness, transaction expenses and net working capital (as compared to an agreed target net working capital amount), and subject to approximately \$10 million withheld in escrow to fund certain potential indemnification matters.

The consummation of the Acquisition is subject to the satisfaction or waiver of, among other customary closing conditions, the accuracy of the representations and warranties in the Purchase Agreement, the compliance by the parties with the covenants in the Purchase Agreement, the absence of any legal order barring the Acquisition and the termination or expiration of the waiting period under the Hart-Scott-Rodino Act. The parties are also provided with customary termination rights, including the right of either party to terminate the Purchase Agreement if the consummation of the Acquisition has not occurred within 120 days after signing unless the failure of the Acquisition to be consummated was substantially caused by such party; provided, however, Solara and Blocker Seller may elect to extend this date for an additional 30 days. The Acquisition is expected to close in June or July of 2020 subject to the satisfaction of the closing conditions.

Under the Purchase Agreement, Solara and Blocker Seller made representations and warranties with respect to the business of Solara and Blocker, as applicable to each, customary for transactions of a similar nature, including, as to Solara, with respect to organization and qualification, subsidiaries, authorization, capitalization, non-contravention, financial statements, the absence of undisclosed liabilities, the non-occurrence of certain events from the date of the latest balance sheet of Solara, litigation, product issues, environmental matters, title to properties, compliance with laws, labor and employment matters, employee benefit plans, tax matters, insurance, licenses and permits, affiliated transactions, material contracts, intellectual property, data privacy and security, key customers and suppliers, brokers' fees, condition and sufficiency of assets, health care compliance, and HIPAA.

Certain representations of Solara and Blocker Seller are qualified in whole or in part by a "material adverse effect" standard for purposes of determining whether a breach of such representations and warranties has occurred and by information in the accompanying disclosure schedules, which contain information that modifies, qualifies and creates exceptions to the applicable representations and warranties of Solara and Blocker Seller set forth in the Purchase Agreement. Moreover, certain representations and warranties in the Purchase Agreement were used for the purpose of allocating risk among the parties, rather than establishing matters as facts. Accordingly, investors and securityholders should not rely on the representations and warranties in the Purchase Agreement as characterizations of the actual state of facts about the parties.

Solara, Blocker Seller and Blocker made certain covenants under the Purchase Agreement, including, among others, the following:

- during the period between the signing of the Purchase Agreement and the closing of the Acquisition, Solara shall, and shall cause its subsidiaries to, use reasonable best efforts to take, or to cause to be taken, all actions, and to do, or cause to be done, all things necessary, proper or advisable to consummate and make effective as promptly as practicable the Acquisition;
- subject to limited exceptions, during the period between the signing of the Purchase Agreement and the closing of the Acquisition, Solara and Blocker Seller shall use reasonable best efforts to carry on the business of Solara, Blocker and their respective subsidiaries in the ordinary course;



- subject to certain conditions, during the period between the signing of the Purchase Agreement and the closing of the Acquisition, Solara is required to provide reasonable access at reasonable times and upon reasonable prior notice to Buyer and its representatives to Solara's and Blocker's senior executive employees and the books and records of Solara, Blocker and their respective subsidiaries to the extent relating to the transition of the business of such entities to the Company;
- during the period between the signing of the Purchase Agreement and the closing of the Acquisition, Solara, Blocker and their respective subsidiaries are required not to, and to direct their representatives not to, solicit, initiate or encourage the initiation of, participate in any discussions or negotiations regarding, or agree to any acquisition proposal by a third party;
- during the period between the signing of the Purchase Agreement and the closing of the Acquisition, provide notice to Buyer of certain material events; and
- subject to limited exceptions, during the period between the signing of the Purchase Agreement and the closing of the Acquisition, neither Solara nor Blocker Seller shall, without the prior written consent of Buyer and subject to certain exceptions, among other things, issue, sell or pledge any of its equity securities; make any non-cash distribution on its equity securities; incur certain indebtedness; sell or otherwise dispose of assets shown on the most recent balance sheet of Solara; make acquisitions; terminate or materially modify any material contract with a third party payor; transfer any material intellectual property rights other than in the ordinary course of business; amend its organizational documents; increase the salary payable to certain employees of Solara or its subsidiaries or increase the coverage or benefits available under certain compensation plans; or change its tax classification.

The Company, Buyer and Merger Sub made certain covenants under the Purchase Agreement, including, among others, the following:

- during the period between the signing of the Purchase Agreement and the closing of the Acquisition, the Company, Buyer and Merger Sub shall not, and shall cause their affiliates not to, contact any channel partner, customer, supplier, distributor, lessee, lessor, equityholder, lender, noteholder or other material business relation of Solara or its subsidiaries with respect to Solara, its subsidiaries, their businesses or the Acquisition, in each case, without the prior written consent of Solara; and
- for a period of one year following the closing of the Acquisition, to provide to each employee who is employed by Solara and its subsidiaries as of the closing of the Acquisition (a) base salary or wage rate and other compensation no less favorable than those provided immediately prior to the date of the closing and (b) employee benefits that are substantially comparable in the aggregate to the employee benefits provided to such employees immediately prior to the closing of the Acquisition.

Each party made certain mutual covenants under the Purchase Agreement, including, among others, to obtain from any governmental authority any consent, approval, authorization, declaration, waiver, license, franchise, permit, certificate or order required to be obtained or made, to avoid any action or proceeding by any governmental authority, to make all necessary filings and thereafter to make any other required submissions with respect to the Purchase Agreement required under any applicable law, and to cooperate with each other in connection with such actions, and to make any necessary filings under the Hart-Scott-Rodino Act and any other applicable antitrust laws.

Buyer will purchase a customary representations and warranties insurance policy in connection with the Acquisition. The representations and warranties policy will be the sole recourse of recovery for the Company, Buyer, Merger Sub and their affiliates for any breach of the representations and warranties made in the Purchase Agreement; however, Buyer and the surviving company of the merger will be entitled to recovery with respect to certain losses as a result of, or in connection with, certain specified litigation and other matters solely out of the indemnity escrow accounts established pursuant to the Purchase Agreement to the extent the amount of such losses has not been satisfied by recovery under the representations and warranties insurance policy. Such recovery is subject to certain customary limitations, including (i) a one-year survival period applicable to the remaining specified matters and (iii) caps in the amount of the applicable indemnity escrow account.

- 3 -

The Company intends to fund the Acquisition and the associated costs through a combination of incremental debt and newly-issued equity (each as described further below), as well as cash on hand or its current line of credit. In furtherance thereof, on May 25, 2020, (i) the Company entered into the Investment Agreement (as defined below) with OEP AHCO Investment Holdings, LLC, a Delaware limited liability company (the "OEP Purchaser"), and, solely for purposes of Section 3.10 thereof, One Equity Partners VII, L.P., a Delaware limited partnership, (ii) the Company entered into the Deerfield Letter Agreement (as defined below) with Deerfield Private Design Fund IV, L.P., a Delaware limited partnership ("Deerfield Private Design Fund IV"), and Deerfield Partners, LLC, a Delaware limited liability company ("Deerfield Partners"), and (iii) Buyer entered into a debt commitment letter with its current lenders, in each case as further described below (such equity and debt financings, excluding the financing pursuant to the Deerfield Letter Agreement, the "Financing"). In the Purchase Agreement, Buyer and Merger Sub agreed, during the period from the signing of the Purchase Agreement to the closing of the Acquisition, to use its reasonable best efforts to consummate the Financing no later than the date the closing is required to be effected under the Purchase Agreement.

Investment Agreement

Pursuant to the Investment Agreement, the OEP Purchaser has agreed, subject to the terms and conditions therein, to purchase in a private placement (the "OEP Investment") up to \$190,000,000 in the aggregate of shares of Class A Common Stock and shares of a new series of preferred stock of the Company designated as "Series A Convertible Preferred Stock," par value \$0.0001 per share ("Series A Preferred Stock"). Pursuant to the Investment Agreement, the closing of the OEP Investment will occur immediately prior to the closing of the Acquisition, subject to the expiration of the waiting period under the Hart-Scott-Rodino Act and certain other conditions described below. Each of the parties has made customary representations and warranties in the Investment Agreement for transactions of the nature of the OEP Investment, including representations and warranties by the Company as to organization, qualification and authorization, non-contravention, required filings, capitalization and voting power, reports filed with the Securities and Exchange Commission, financial statements, brokers' fees, litigation, compliance with laws, the non-occurrence of material adverse effects since the end of the prior fiscal year, Nasdaq listing, indebtedness, anti-takeover provisions under organizational documents, non-investment company status, absence of "bad actor" disqualification events, and no "plan assets," as such term is defined in 29 C.F.R. Section 2510.3-101, as modified by Section 3(42) of the Employee Retirement Income Security Act of 1974, as amended.

The rights and preferences of the Series A Preferred Stock will be designated by the Company's board of directors in a certificate of designations (the "Certificate of Designations") forming part of the Company's second amended and restated certificate of incorporation, which Certificate of Designations will be filed prior to the closing of the OEP Investment with the Delaware Secretary of State in the form attached to the Investment Agreement.

The Series A Preferred Stock will rank senior to the Class A Common Stock with respect to rights on the distribution of assets on any voluntary or involuntary liquidation, dissolution or winding up of the affairs of the Company, having a liquidation preference equal to its par value of \$0.0001 per share. The Series A Preferred Stock will participate equally and ratably on an as-converted basis with the holders of Class A Common Stock in all cash dividends paid on the Class A Common Stock.

The conversion of Series A Preferred Stock into Class A Common Stock issued to the OEP Purchaser in the OEP Investment is subject to Stockholder Approval (as defined and further described below); provided that, subject to certain exceptions in connection with a change of control, the Company or the OEP Purchaser, as applicable, may elect to exchange the Series A Preferred Stock following the six-month anniversary of the issuance thereof (and, with respect to the OEP Purchaser, following the later lapse in certain transfer restrictions pursuant to the Investment Agreement) for the cash value of such shares as calculated based on the volume-weighted average price per share of Class A Common Stock on the day immediately prior to the date of conversion, in lieu of delivery of shares of Class A Common Stock (if the shares deliverable upon conversion would otherwise violate listing rules of Nasdaq). After the Stockholder Approval is obtained, the Company or the OEP Purchaser may convert the Series A Preferred Stock will be non-voting.

Pursuant to the Investment Agreement, the Company agreed to increase the size of its board of directors by two members in order to elect to the board of directors, promptly following and in any event within five business days of the closing of the OEP Investment, one individual designated by the OEP Purchaser and one independent director nominated by the Company or the Company's board of directors. For as long as the OEP Purchaser or its affiliates hold beneficial ownership of at least 25% of the shares of Class A Common Stock and Series A Preferred Stock issued pursuant to the Investment Agreement on an as-converted basis, the OEP Purchaser will have the right to designate one director for election to the Company's board of directors. For as long as the OEP Purchaser has the right to designate one director for election, the Company may not increase the number of directors on the board of directors without the prior consent of the OEP Purchaser. In addition to the foregoing director designation right, for as long as the OEP Purchaser or its affiliates hold beneficial ownership of at least 50% of the shares of Class A Common Stock and Series A Preferred Stock issued pursuant to the Investment Agreement on an as-converted basis, the OEP Purchaser will have the right to designate a non-voting observer to the board of directors.

Under the Investment Agreement, the Company agreed to hold a meeting of stockholders at which a proposal will be considered with respect to the approval of the issuance of shares of Class A Common Stock to the OEP Purchaser in connection with the conversion of the Series A Preferred Stock into Class A Common Stock that would, absent such approval, violate Nasdaq Listing Rule 5635 as promptly as reasonably practicable after the signing of the Investment Agreement and to prepare and file a preliminary proxy statement relating to such meeting in any event no later than the first to occur of (x) the 30th day following the date on which the audited financial statements of Solara for the fiscal year ending December 31, 2019 required to be filed by the Company in connection with the Acquisition are received by the Company and (y) the 15th day following the closing of the OEP Investment.

The OEP Purchaser and its affiliates will be subject to certain standstill restrictions, including that they will be restricted from acquiring additional shares of Class A Common Stock, for as long as the OEP Purchaser holds beneficial ownership of at least 25% of the shares of Class A Common Stock and Series A Preferred Stock issued pursuant to the Investment Agreement on an as-converted basis. Subject to certain customary exceptions, the OEP Purchaser will be restricted from transferring the Class A Common Stock, Series A Preferred Stock or shares of Class A Common Stock issuable upon conversion of the Series A Preferred Stock until the earlier of (i) the 18-month anniversary of the closing of the OEP Investment and (ii) the date on which certain specified volume-weighted average price targets for the Class A Common Stock are met.

The closing of the OEP Investment is subject to the satisfaction or waiver of, among other customary closing conditions, the accuracy of the representations and warranties in the Purchase Agreement, the compliance by the parties with the covenants in the Purchase Agreement, the absence of any legal order barring the OEP Investment, the termination or expiration of the waiting period under the Hart-Scott-Rodino Act, the conditions precedent to the Company's obligation to consummate the closing of the Acquisition having been satisfied or waived and the parties to the Purchase Agreement being ready, willing and able to consummate the Acquisition immediately subsequent to the OEP Investment, and the delivery of customary ancillary documents. The obligation of the OEP Purchaser to effect the closing is subject to the satisfaction or waiver of certain additional conditions, among others, there having occurred no material adverse effect with respect to the OEP Purchaser and the Class A Common Stock issuable upon conversion of the Series A Preferred Stock having been approved for listing on Nasdaq, stockholders holding at least 45% of the voting power of the Company as of the earlier of (x) the record date of the first meeting of stockholders held to consider the Stockholder Approval or (y) the closing of the OEP Investment having entered into voting agreements obligating such person to vote in favor of the Stockholder Approval, and certain consents for the Acquisition having been obtained.

Pursuant to the Investment Agreement, at the closing of the OEP Investment, the Company, AdaptHealth Holdings LLC, a Delaware limited liability company and direct subsidiary of the Company ("AdaptHealth Holdings"), the OEP Purchaser and certain other stockholders of the Company will enter into an Amended and Restated Registration Rights Agreement (the "A&R Registration Rights Agreement") which will amend, restate and replace the registration rights agreement entered into by the Company, AdaptHealth Holdings and certain other holders on November 8, 2019 in connection with the closing of the Company's initial business combination, pursuant to which, among other things, the OEP Purchaser and its affiliates will be provided with customary registration rights with respect to the shares of Class A Common Stock issued to the OEP Purchaser in the OEP Investment and all shares of Class A Common Stock issuable upon conversion of the Series A Preferred Stock, including that the Company will agree to file a registration statement under the Securities Act registering the issuance and resale of all such shares.

The foregoing summary of the Investment Agreement is qualified in its entirety by the full text thereof, which is filed as Exhibit 10.1 hereto and incorporated by reference herein.

Put/Call Option and Consent Agreement

On May 25, 2020, the Company and AdaptHealth Holdings entered into a Put/Call Option and Consent Agreement (the "Put/Call Agreement") with BlueMountain Foinaven Master Fund LP, a Cayman Islands exempted partnership, BMSB LP, a Delaware limited partnership, BlueMountain Fursan Fund LP, a Cayman Islands exempted partnership, and BlueMountain Summit Opportunities Fund II (US) LP, a Delaware limited partnership (collectively, the "Option Parties"), pursuant to which certain put and call rights were granted to the parties with respect to shares of Class A Common Stock, shares of Class B Common Stock, par value \$0.0001 per share of the Company ("Class B Common Stock") and Common Units of AdaptHealth Holdings (each such Common Unit, together with one share of Class B Common Stock, a "Consideration Unit") held by the Option Parties.

Pursuant to the Put/Call Agreement, during the period from the closing of the Acquisition until October 31, 2020 (the "Option Period"), the Option Parties may require the Company to purchase up to 1,898,967 shares of Class A Common Stock and/or Consideration Units held by the Option Parties (such shares of Class A Common Stock and Consideration Units, collectively, "Interests") at a price per share of Class A Common Stock or per Consideration Unit equal to the greater of (x) \$14.50 and (y) 85% of the 30-day volume-weighted average price per share of the Class A Common Stock on the date the exercise notice is delivered. During the Option Period (including within two calendar days of the delivery of an exercise notice by the Option Parties), the Company may also require the Option Parties to sell up to 1,898,967 of the Interests held by the Option Parties to the Company at a price per share of Class A Common Stock or per Consideration Unit of \$15.76.

In addition, under the Put/Call Agreement, the Option Parties waived certain consent rights under certain Promissory Notes, dated as of November 8, 2019, and Amended and Restated Promissory Notes, dated as of March 20, 2019, issued by AdaptHealth Holdings in favor of each of the Option Parties (collectively, the "BM Notes"), and the Company irrevocably agreed to pay all PIK interest payable under the BM Notes following the closing of the Acquisition in cash rather than through an increase in the principal amount of the notes.

The foregoing summary of the Put/Call Agreement is qualified in its entirety by the full text thereof, which is filed as Exhibit 10.2 hereto and incorporated by reference herein.

Deerfield Letter Agreement

On May 25, 2020, the Company entered into a letter agreement (the "Deerfield Letter Agreement") with Deerfield Private Design Fund IV and Deerfield Partners pursuant to which, within 30 days following the execution of the Deerfield Letter Agreement, Deerfield Private Design Fund IV and the Company agreed to negotiate, enter into and consummate an exchange (the "Exchange"), pursuant to an exchange agreement (the "Exchange Agreement"), which would provide, among other things, for the exchange of all of the Class A Common Stock held by Deerfield Private Design Fund IV, other than such number of shares of Class A Common Stock representing 4.5% of the outstanding shares of Class A Common Stock, for a newly-created class of convertible, non-voting preferred stock ("Series B-1 Preferred Stock") that would be substantially equivalent to common stock except that it would have no voting rights, subject to limited exceptions, would be subject to a beneficial ownership limitation of 4.9% of the Company's outstanding Class A Common Stock and would have a liquidation preference of \$0.0001 per share of Series B-1 Preferred Stock. In connection with the execution of the Exchange Agreement, the registration rights agreement to which Deerfield Private Design Fund IV is a party would be modified to limit its, or any related fund's, obligations in respect of lockup agreements in connection with underwritten public offerings.

The Deerfield Letter Agreement also provides that the Company and Deerfield Partners would, within 30 days following the execution of the Deerfield Letter Agreement, negotiate and enter into an agreement for the purchase and sale of \$35,000,000 of a separate, newly-created, class of convertible, non-voting preferred stock ("Series B-2 Preferred Stock"), which would be convertible into Series B-1 Preferred Stock upon the obtaining of the Stockholder Approval and would be subject to a beneficial ownership limitation of 4.9% of the Company's outstanding Class A Common Stock, on terms that are substantially equivalent to the terms of the OEP Investment, subject to the foregoing and certain exceptions, including that transfer restrictions only survive for 60 days following closing of the investment, that the investment shall be conditioned on the consummation of the Exchange and that no board or observer seat shall be provided.

The definitive documentation governing the Exchange and the purchase and sale of the Series B-2 Preferred Stock has not been finalized, and, accordingly, the actual terms may differ from the description of such terms in the Deerfield Letter Agreement. The foregoing summary of the Deerfield Letter Agreement is qualified in its entirety by the full text thereof, which is filed as Exhibit 10.3 hereto and incorporated by reference herein.

Debt Commitment Letter

In connection with the signing of the Purchase Agreement, Buyer entered into a debt commitment letter (the "Commitment Letter"), dated as of May 25, 2020, with Regions Bank ("Regions Bank"), Regions Capital Markets, a division of Regions Bank ("RCM"), Citizens Bank, N.A. ("Citizens"), Deutsche Bank AG New York Branch ("DBNY"), Deutsche Bank Securities Inc. ("DBSI"), Royal Bank of Canada ("Royal Bank") and RBC Capital Markets, LLC ("RBCCM"), pursuant to which Regions Bank, Citizens, DBNY and Royal Bank have committed to provide an incremental secured term loan facility to Buyer in an aggregate principal amount of up to \$240,600,000 (the "Incremental Term Facility"). The Incremental Term Facility will be issued under Buyer's existing Third Amended and Restated Credit and Guaranty Agreement, dated as of March 20, 2019, as amended, among Buyer, the guarantors party thereto, the lenders from time to time party thereto and CIT Finance LLC, as administrative agent. The Incremental Term Facility is available (i) to finance the Acquisition, in part, and (ii) to pay fees and expenses incurred in connection therewith. Under the terms of the Commitment Letter, RCM, Citizens, DBSI and RBCCM will serve as joint lead arrangers and joint book runners for the Incremental Term Facility.

The funding of the Incremental Term Facility provided for in the Commitment Letter is contingent on the satisfaction of customary conditions, including, among other things, (i) the execution and delivery of definitive documentation in accordance with the terms sets forth in the Commitment Letter and (ii) the consummation of the Acquisition in accordance with the terms of the Purchase Agreement. The definitive documentation governing such debt financing has not been finalized, and, accordingly, the actual terms may differ from the description of such terms in the Commitment Letter.

Item 2.03. Creation of a Direct Financial Obligation or an Obligation Under an Off-Balance Sheet Arrangement of a Registrant.

The information provided under the heading "Put/Call Option and Consent Agreement" in Item 1.01 is incorporated by reference into this Item 2.03.

Item 3.02. Unregistered Sale of Equity Securities.

The consideration to be issued pursuant to the Investment Agreement and the documents to be entered into pursuant to the Deerfield Letter Agreement, and a portion of the consideration to be issued pursuant to the Purchase Agreement, each as described in Item 1.01 of this report, which descriptions are incorporated by reference into this Item 3.02, will consist of unregistered shares of Class A Common Stock and/or Series A Preferred Stock and other preferred stock, as applicable. Such shares will be issued in private placements exempt from registration under 4(2) of the Securities Act of 1933, as amended (the "Securities Act"), because the offer and sale of such securities does not involve a "public offering," as defined in Section 4(2) of the Securities Act, and other applicable requirements will be met.

Nasdaq Listing and Stockholder Approval; Registration

The Company's Class A Common Stock is currently listed on The Nasdaq Capital Market, and the Company is subject to the Nasdaq Listing Rules (the "Nasdaq Rules"). Pursuant to the terms of the Investment Agreement and the Deerfield Letter Agreement, the Company has covenanted to seek approval of its stockholders (the "Stockholder Approval") as may be required by the Nasdaq Rules, including pursuant to Nasdaq Rule 5635(a), applicable in connection with the removal of the conversion restrictions as will be applicable to the Series A Preferred Stock and Series B-2 Preferred Stock to prevent conversions of such Series A Preferred Stock or Series B-2 Preferred Stock, as applicable, into Class A Common Stock or Series B-1 Preferred Stock, as applicable.

The Company plans to hold a meeting of stockholders at which a proposal to obtain the Stockholder Approval is considered as promptly as reasonably practicable. In addition, on May 25, 2020, certain stockholders of the Company, including the Option Parties, entered into agreements with the OEP Purchaser to vote all shares of common stock of the Company owned by such persons as of the applicable record date over which such persons have voting power as of the date hereof, amounting to, in the aggregate when taken together with the Company's voting agreement with Deerfield Private Design Fund IV, greater than 50% of all current voting power of the Company, (i) in favor of the Stockholder Approval for the removal of the conversion restrictions applicable to the Series A Preferred Stock and (ii) against matters which would result in a breach by the Company of its agreement with the OEP Purchaser or otherwise be expected to impede the transactions contemplated by such agreement (collectively, the "Voting Agreements"). On May 25, 2020, the Company entered into a voting agreement with Deerfield Private Design Fund IV (i) in favor of the Stockholder Approval to remove the restrictions applicable to the Series A Preferred Stock and (ii) against matters which would result in a breach by the Company of its agreement with the OEP Purchaser or otherwise be expected to impede the transactions contemplated by such agreement (remove the restrictions applicable to the Series A Preferred Stock and (ii) against matters which would result in a breach by the Company of its agreement with the OEP Purchaser or otherwise be expected to impede the transactions contemplated by such agreement. The Company also covenanted in the Deerfield Letter Agreement to use reasonable best efforts to enter into further voting agreements with stockholders of the Company representing a majority of the outstanding common stock of the Company in the form of the Voting Agreements to support the Stockholder Approval for the removal of the conversion restrict

Cautionary Statement Regarding Forward Looking Statements

This Current Report on Form 8-K includes certain statements that are not historical facts but are forward-looking statements for purposes of the safe harbor provisions under the United States Private Securities Litigation Reform Act of 1995. Forward-looking statements generally are accompanied by words such as "believe," "may," "will," "estimate," "continue," "anticipate," "intend," "expect," "should," "uplan," "predict," "potential," "seem," "seek," "future," "outlook," and similar expressions that predict or indicate future events or trends or that are not statements of historical matters. These forward-looking statements include, but are not limited to, statements regarding projections, estimates and forecasts of revenue and other financial and performance metrics and projections of market opportunity and expectations. These statements are based on various assumptions and on the current expectations of Company management and are not predictions of actual performance. These forward-looking statements are provided for illustrative purposes only and are not intended to serve as, and must not be relied on, by any investor as, a guarantee, an assurance, a prediction or a definitive statement of fact or probability. Actual events and circumstances are difficult or impossible to predict and will differ from assumptions. Many actual events and circumstances are beyond the control of the Company.

These forward-looking statements are subject to a number of risks and uncertainties, including the outcome of judicial and administrative proceedings to which the Company may become a party or governmental investigations to which the Company may become subject that could interrupt or limit the Company's operations, result in adverse judgments, settlements or fines and create negative publicity; changes in the Company's response to it. A further description of such risks and uncertainties can be found in the Company's filings with the Securities and Exchange Commission. If the risks materialize or assumptions prove incorrect, actual results could differ materially from the results implied by these forward-looking statements. There may be additional risks that the Company presently knows or that the Company currently believes are immaterial that could also cause actual results to differ from those contained in the forward-looking statements. In addition, forward-looking statements reflect the Company will cause the Company is response to change. However, while the Company may elect to update these forward-looking statements at some point in the future, the Company specifically disclaims any obligation to do so. These forward-looking statements should not be relied upon as representing the Company's assessments as of any date subsequent to the date of this press release. Accordingly, undue reliance should not be placed upon the forward-looking statements.

Item 9.01. Financial Statements and Exhibits.

(d) Exhibits

Exhibit No.	Description of Exhibit
<u>10.1</u>	Investment Agreement, dated as of May 25, 2020, by and among the Company, OEP AHCO Investment Holdings, LLC and, solely for purposes of Section 3.10, One Equity Partners VII, L.P.
<u>10.2</u>	Put/Call Agreement and Consent Agreement, dated as of May 25, 2020, by and among the Company, AdaptHealth Holdings LLC, BlueMountain Foinaven Master Fund L.P., BMSB L.P., BlueMountain Fursan Fund L.P. and BlueMountain Summit Opportunities Fund II (US) L.P.
<u>10.3</u>	Letter Agreement, dated as of May 25, 2020, by and among the Company, Deerfield Private Design Fund IV, L.P. and Deerfield Partners, LLC.

- 7 -

SIGNATURES

Pursuant to the requirement of the Securities Exchange Act of 1934, as amended, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

Dated: May 29, 2020

ADAPTHEALTH CORP.

By:

/s/ Gregg Holst Name: Gregg Holst Title: Chief Financial Officer

Execution Version

INVESTMENT AGREEMENT

dated as of May 25, 2020

by and between

ADAPTHEALTH CORP.,

OEP AHCO INVESTMENT HOLDINGS, LLC

AND,

SOLELY FOR PURPOSES OF SECTION 3.10,

ONE EQUITY PARTNERS VII, L.P.

ARTICLE I PURCHASE; CLOSING	1
1.1 Purchase	1
1.2 <u>Closing</u>	2
1.3 <u>Closing Conditions</u>	3
ARTICLE II REPRESENTATIONS AND WARRANTIES	5
2.1 <u>Representations and Warranties of the Company</u>	5
2.2 Representations and Warranties of the Purchaser	12
ARTICLE III COVENANTS	16
3.1 Confidentiality	16
3.2 Listing	16
3.3 Efforts	17
3.4 Legend	17
3.5 Back Leverage	18
3.6 Corporate Actions	18
3.7 <u>Negative Covenants</u>	18
3.8 <u>Tax Matters</u>	19
3.9 <u>Stockholder Approval</u>	20
ARTICLE IV SURVIVAL	22
4.1 <u>Survival</u>	22
ARTICLE V SHAREHOLDER RIGHTS	22
5.1 Board Matters	22
5.2 Information Rights	25
5.3 <u>Transfer Restrictions</u>	25
5.4 <u>Standstill</u>	26
5.5 <u>Right of First Offer</u>	28
ARTICLE VI MISCELLANEOUS	29
6.1 Expenses	29
6.2 <u>Amendment; Waiver</u>	29
6.3 Counterparts; Electronic Transmission	30
6.4 <u>Governing Law</u>	30

6.5 WAIVER OF JURY TRIAL	30
6.6 Notices	30
6.7 Entire Agreement	31
6.8 <u>Assignment</u>	32
6.9 Interpretation; Other Definitions	32
6.10 <u>Captions</u>	37
6.11 Severability	37
6.12 No Third Party Beneficiaries	37
6.13 Public Announcements	37
6.14 Specific Performance	38
6.15 <u>Termination</u>	38
6.16 Effects of Termination	38
6.17 <u>Non-Recourse</u>	39
Schedule A: Legal Opinion	
Schedule B: Investor Requirements	
Schedule C: Competitors	
ANNEXES	
AnnexI – Form of Certificate of Designations	
Annex II – Form of Amended and Restated Registration Rights Agreement	
INDEX OF DEFINED TERMS	
Term Affiliate Agreement Beneficial Owner/Beneficially Own/Beneficial Ownership Board of Directors Board Representatives BSA/PATRIOT Act business day Bylaws Capitalization Date Certificate of Incorporation Class A Common Stock Class B Common Stock Closing	Location of Definition 6.9(g) Preamble 6.9(d) 5.1(a) 5.1(a) 2.2(c)(8) 6.9(e) 2.1(d) 2.1(b)(1) 2.1(d) Recitals 2.1(b)(1) 1.2(a)

ii

Term	Location of Definition
Closing Date	1.2(a)
Closing Notice	1.2(a) 1.2(a)
Common Stock	2.1(b)(1)
Company	Preamble
Company Material Adverse Effect	6.9(h)
Company Stock Awards	2.1(b)(1)
Company Stock Options	2.1(b)(1)
Confidentiality Agreement	3.1
Contract	6.9(i)
Derivative Instruments	6.9(j)
DGCL	2.2(f
Effect	6.9(k)
Encumbrance	6.9(l)
Equity Securities	6.9(m)
Exchange Act	2.1
GAAP	2.1(e)(2)
Governmental Entity	6.9(n)
HSR Act	6.9(o)
herein/hereof/hereunder	6.9(c)
including/includes/included/include	6.9(b)
Independent Board Representative	5.1(a)
Information	3.1
Knowledge of the Company	6.9(p)
Law	6.9(q)
Lien	6.9(r)
Non-Recourse Party	6.17
OFAC List	2.2(c)(8)
or	6.9(a)
person	6.9(f)
Preferred Stock	2.1(b)(1)
Purchase Price	1.1
Purchaser	Preamble
Purchaser Board Representative	5.1(a)
Reference Date	2.1
Registration Rights Agreement	6.9(t)
SEC	2.1(e)(1)
SEC Reports	2.1(e)(1) 2.1(e)(1)
Securities Act	2.1
Shares	1.1
Stock Plan	2.1(b)(1)
Subsidiary	6.9(u)
Target Acquisition	6.9(v)
Transfer	6.9(w)
Transfer Agent	
Transfer Agent	1.2(b)(1) 1.2(b)(1)
Unaffiliated Shareholders	1.2(b)(1)
	6.9(x)
Voting Debt Womente	2.1(b)(2) 2.1(b)(1)
Warrants	2.1(b)(1)

iii

INVESTMENT AGREEMENT, dated as of May 25, 2020 (this "<u>Agreement</u>"), by and between AdaptHealth Corp., a Delaware corporation (the "<u>Company</u>"), OEP AHCO Investment Holdings, LLC, a Delaware limited liability company (the "<u>Purchaser</u>"), and, solely for purposes of <u>Section 3.10</u>, One Equity Partners VII, L.P., a Delaware limited partnership (the "<u>Guarantor</u>").

RECITALS:

WHEREAS, the Company proposes to issue and sell to the Purchaser (including its permitted assignees pursuant to <u>Section 6.8</u>) (x) shares of Class A Common Stock, par value \$0.0001 per share, of the Company (the "<u>Class A Common Stock</u>") and (y) shares of Series A Convertible Preferred Stock, par value \$0.0001 per share, of the Company (the "<u>Series A Preferred Stock</u>"), having the designation, preferences, conversion or other rights, voting powers, restrictions, limitations as to dividends, qualifications and terms and conditions, as specified in the form of Certificate of Designation, Preferences and Rights attached hereto as <u>Annex I</u> (the "<u>Certificate of Designations</u>");

WHEREAS, contemporaneously with the execution hereof, the Company has entered into that certain Stock Purchase Agreement and Agreement and Plan of Merger, by and among the Company, Eleanor Merger Sub LLC, Solara Holdings, LLC and LCP Solara Blocker Seller, LLC, in its capacity as Blocker Seller and the Representative (in each case as defined therein) (the "Stock Purchase Agreement"), regarding the Target Acquisition; and

WHEREAS, capitalized terms used in this Agreement have the meanings set forth in Section 6.9 or such other section indicated in the preceding Index of Defined Terms.

NOW, THEREFORE, in consideration of the premises, and of the representations, warranties, covenants and agreements set forth herein, the parties agree as follows:

ARTICLEI

PURCHASE; CLOSING

1.1 Purchase.

(a) On the terms and subject to the conditions herein and subject to the satisfaction (or waiver) of the conditions set forth in Section 1.3 below, on the Closing Date (as defined below), the Company agrees to sell and issue to the Purchaser, and the Purchaser agrees to purchase from the Company, the Shares for the total price in cash of \$190,000,000 (the "<u>Initial Purchase Price</u>"); provided, that, in the event that, in one or more transactions, the Company sells additional shares of Class A Common Stock to third-party purchasers between the date hereof and the Closing Date (other than the shares of Common Stock issued in connection with the Target Acquisition) at a price per share greater than the Purchase Price Per Common Share (such sales, collectively, the "<u>Common Equity Offering</u>"), then the Company may elect in its sole discretion by providing written notice to the Purchaser at least three (3) business days prior to the Closing Date, which such notice shall contain the total purchase price of the shares sold in the Common Equity Offering (the "<u>Aggregate Common Equity Offering Amount</u>"), to reduce the Initial Purchase Price by an amount up to the Aggregate Common Equity Offering, not to exceed \$50,000,000 (the "<u>Purchase Price Cut-Back</u>"), and in such event the Stock Consideration Amount shall be reduced by an amount equal to (rounded down to the nearest whole number) (x) the amount of the Purchaser an amount equal to the product of (i) 0.03 and (ii) the amount of the Purchaser's agreement to permit the Purchase Price Cut-Back (the "<u>Commitment Fee</u>"), in cash. The Commitment Fee payable to the Purchaser shall be offset against such Purchaser's obligation to pay its Initial Purchase Price, and the Purchaser's Initial Purchase Price shall be reduced (without duplication) by an amount equal to the sum of (i) the amount of the Purchase Price, and the Purchaser's Initial Purchase Price shall be reduced (without duplication) by an amount equal to the sum of (i) the amount of the Purchase Price, and the Purch

(b) Purchased Preferred Shares and Purchased Common Shares to be issued and sold by the Company to the Purchaser pursuant to this Agreement are collectively referred to as the "<u>Shares</u>." The Company will use the proceeds of the Purchase Price for (a) payment of fees and expenses incurred in connection with the transactions contemplated by this Agreement, (b) other general corporate purposes, and (c) funding of the Target Acquisition.

1.2 <u>Closing</u>.

(a) The closing of the sale of Shares contemplated hereby (the "<u>Closing</u>") shall occur on the date, and immediately prior to the consummation, of the Target Acquisition (the consummation of the Target Acquisition, the "<u>Target Acquisition Closing</u>"). Upon (a) satisfaction or waiver of the conditions set forth in Section 1.3 of this Agreement (other than those conditions that by their nature are to be satisfied at the Closing, but subject to their satisfaction or waiver at or prior to the Closing), and (b) written notice from (or on behalf of) the Company to the Purchaser (the "<u>Closing Notice</u>") that the Company reasonably expects all conditions to the Target Acquisition Closing to be satisfied on a date that is not less than three (3) business days from the date of such Closing Notice, the Purchaser shall deliver to the Company, at least one (1) business day prior to the proposed date of the Closing specified in the Closing Notice, the Purchase Price, which shall be held in escrow by the Company until the Closing and immediately subsequent Target Acquisition Closing. In the event the Target Acquisition Closing does not occur on the date specified in such Closing Notice, the Company shall promptly (but not later than one (1) business day thereafter) return the Purchase Price to the Purchaser. The date that the Closing occurs shall be referred to as the "<u>Closing Date</u>".

(b) At the Closing:

(1) the Company will deliver to the Purchaser (i) a copy of the irrevocable instructions (the "<u>Transfer Instructions</u>") to Continental Stock Transfer & Trust Company (the "<u>Transfer Agent</u>") instructing the Transfer Agent to issue to such Purchaser, in book-entry form, the Purchased Common Shares and Purchased Preferred Shares on the Closing Date, (ii) an opinion of Willkie Farr & Gallagher LLP containing the opinions substantially in the form set forth in <u>Schedule A</u>, (iii) the executed Amended and Restated Registration Rights Agreement, in the form of <u>AnnexII</u> hereto; provided, that the Company and the holders of a majority of Registrable Securities (as defined in the form of Amended and Restated Registration Rights Agreement) shall be entitled to make amendments to the form of Amended and Restated Registration Rights Agreement to the extent that such amendments could be made following the effectiveness thereof after giving effect to the acquisition by Purchaser of the Shares contemplated hereby, and (iv) all other documents, instruments and writings required to be delivered by the Company to the Purchaser pursuant to this Agreement or otherwise required in connection herewith.

(2) the Purchaser will deliver or cause to be delivered (i) to a bank account previously designated by the Company in writing, the Purchase Price by wire transfer of immediately available funds (provided, however, that the delivery of the Purchase Price in escrow in accordance with Section 1.2 shall satisfy this obligation), (ii) the executed Amended and Restated Registration Rights Agreement and (iii) all other documents, instruments and writings required to be delivered by the Purchaser to the Company pursuant to this Agreement or otherwise required in connection herewith.

1.3 Closing Conditions.

(a) The obligations of the Purchaser, on the one hand, and the Company, on the other hand, to effect the Closing is subject to the satisfaction by the Purchaser and the Company at the Closing of the following conditions:

(1) the waiting period (and any extension thereof) applicable to the consummation of the transactions contemplated hereby under the HSR Act shall have expired or been terminated;

(2) no temporary restraining order, preliminary or permanent injunction or other judgement or order issued by any Governmental Entity shall have been issued, and no Law shall be in effect, restraining, enjoining, making illegal or otherwise prohibiting the consummation of the transactions contemplated by this Agreement; and

(3) all conditions precedent to the Company's obligation to consummate closing of the Target Acquisition shall have been satisfied or, waived (other than those conditions which, by their nature, are to be satisfied at the closing of the Target Acquisition) and the parties to the Target Acquisition are ready, willing and able to consummate the Target Acquisition immediately subsequent to the Closing.

(b) The obligation of the Purchaser to effect the Closing is subject to the satisfaction by the Company at the Closing of the following conditions:

(1) (A) the Fundamental Representations (other than Section 2.1(a)(1)(ii) and Section 2.1(a)(5)) (disregarding all qualifications as to materiality or "Company Material Adverse Effect" set forth therein) shall be true and correct in all material respects as of the Closing Date as if made on and as of the Closing Date (except to the extent any such Fundamental Representation speaks as of the date of this Agreement or any other specific date, in which case such representation or warranty shall be true and correct as of such date), (B) Section 2.1(a)(1)(ii) (for the avoidance of doubt, without disregarding any qualifications as to materiality or "Company Material Adverse Effect" set forth therein) shall be true and correct in all respects as of the Closing Date as if made on and as of the Closing Date, (C) Section 2.1(a)(5) shall be true and correct as of the Closing Date as if made on and as of the Closing Date, (C) Section 2.1(a)(5) shall be true and correct as of the Closing Date as if made on and as of the Closing Date as if made on and as of the Closing Date as if made on and as of the Closing Date as if made on and as of the Closing Date as if made on and as of the Closing Date (C) Section 2.1(a)(5) shall be true and correct as of the Closing Date as if made on and as of the Closing Date as if made on and as of the Closing Date as if made on and as of the Closing Date as if made on and as of the Closing Date as if made on and as of the Closing Date as though made on and as of the Closing Date (except to the extent any such representation or warranty speaks as of the date of this Agreement or warranty speaks as of the date of this Agreement or any other specific date, in which case such representation or warranty shall be true and correct as of such date), except, solely with respect to this clause (D), where the failure of such as though made on and warranties to be so true and correct would not, individually or in the aggregate, reasonably be expected to have a Company Material Adverse Effect;

(2) the Company shall have performed or complied with in all material respects all of the covenants and agreements required to be performed or complied with by the Company under this Agreement on or prior to the Closing Date;

(3) since the date of this Agreement, there shall not have occurred and be continuing any Company Material Adverse Effect;

(4) Purchaser shall have received a certificate, signed on behalf of the Company by an authorized officer thereof, certifying that the conditions set forth in Section 1.3(b)(1), 1.3(b)(2) and 1.3(b)(3) have been satisfied;

(5) the Company shall have delivered to Purchaser a copy of the Certificate of Designations that has been filed with the Secretary of State of the State of Delaware;

(6) (A) any Purchased Common Shares and (B) any shares of Common Stock issuable upon conversion of any Purchased Preferred Shares at the Conversion Rate specified in the Certificate of Designations as in effect on the date hereof shall have been approved for listing on the Nasdaq, subject to official notice of issuance;

(7) if the Company Stockholder Approval has not been obtained prior to the Closing, (A) Voting Parties shall have entered into Voting Agreements or (B) other Persons shall have entered into other voting agreements in substantially the form of the Voting Agreements, which Voting Agreements and other such voting agreements collectively represent at least 45% of the voting power of the Company as of the earlier of (x) the record date of the first Company Stockholders' Meeting held for the Company Stockholder Approval or (y) the Closing entitled to vote with respect to the Company Stockholder Approval and such Voting Agreements and other such voting agreements shall be in full force and effect; and

(8) the consents set forth on Schedule 2.1(a)(3) have been obtained and are in full force and effect.

(c) The obligation of the Company to effect the Closing is subject to the satisfaction by the Purchaser at the Closing of the following conditions:

(1) the representations and warranties of the Purchaser set forth in Section 2.2 (disregarding all qualifications as to materiality set forth therein) shall be true and correct as of the Closing Date as though made on the Closing Date (except to the extent any such representation or warranty speaks as of the date of this Agreement or any other specific date, in which case such representation or warranty shall be true and correct as of such date), except where the failure of such representations and warranties to be so true and correct would not, individually or in the aggregate, reasonably be expected to materially impair or delay the Purchaser's ability to perform or comply with its obligations under this Agreement or to consummate the transactions contemplated hereby or thereby;

(2) the Purchaser shall have performed or complied with in all material respects all covenants and agreements required to be performed or complied with by the Purchaser under this Agreement on or prior to the Closing Date; and

(3) the Company shall have received a certificate, signed on behalf of the Purchaser by an authorized officer thereof, certifying that the conditions set forth in Section 1.3(c)(1) and Section 1.3(c)(2) have been satisfied.

ARTICLEII

REPRESENTATIONS AND WARRANTIES

2.1 <u>Representations and Warranties of the Company</u>. Except as set forth (x) in publicly available SEC Reports filed by the Company with the SEC following December 31, 2019 (the "<u>Reference Date</u>"), excluding any disclosures set forth in risk factors or any "forward looking statements" within the meaning of the Securities Act of 1933 (the "<u>Securities Act</u>") or the Securities Exchange Act of 1934, as amended, (the "<u>Exchange Act</u>") or (y) in a correspondingly identified schedule attached hereto (provided that any such disclosure shall be deemed to be disclosed with respect to each other representation and warranty to which the relevance of such exception is reasonably apparent on the face of such disclosure), the Company represents and warrants to the Purchaser that:

(a) <u>Incorporation and Authority</u>.

(1) The Company is duly organized, validly existing and in good standing under the Laws of the State of Delaware. The Company has all requisite corporate or other applicable organizational power to (i) enter into, consummate the transactions contemplated by, and carry out its obligations under this Agreement, and (ii) own, lease and operate its properties and carry on its business as presently conducted, and the Company is duly qualified to do business and is in good standing in all jurisdictions where its ownership or leasing of property or the conduct of its business requires it to be so qualified, except for any failure under clause (ii) that would not, individually or in the aggregate, reasonably be expected to have a Company Material Adverse Effect. The execution and delivery by the Company of this Agreement and each other agreement, document, instrument, schedule or certificate contemplated by this Agreement to be executed by the Company in connection with the transactions contemplated hereunder (the "Ancillary Documents") and the consummation by the Company of the transactions contemplated by this Agreement and the Ancillary Documents have been duly authorized by all requisite corporate or other similar organizational action on the part of the Company. This Agreement has been and the Ancillary Documents will be duly executed and delivered by the Company. Assuming due authorization, execution and delivery by the other parties hereto, this Agreement constitutes and the Ancillary Documents will constitute the legal, valid and binding obligations of the Company, enforceable against it in accordance with their respective terms, subject in each case to the effect of any applicable bankruptcy, reorganization, insolvency, moratorium or similar Laws now or hereafter in effect relating to or affecting creditors' rights and remedies generally and subject, as to enforceability, to the effect of general equitable principles (regardless of whether enforcement is sought in a proceeding in equity or at law). True and complete copies of the Second Amended and Restated Certificate of Incorporation of the Company (as amended or modified from time to time prior to the date hereof, the "Certificate of Incorporation")) and the Amended and Restated Bylaws of the Company (as amended or modified from time to time prior to the date hereof, the "Bylaws")), each as in effect, have been made available to the Purchaser prior to the date hereof. The Board of Directors, at a meeting duly called and held or by written consent, adopted resolutions (i) directing that the Company submit to the holders of Common Stock a proposal to approve the removal of the Conversion Restriction (as such term is defined in the Certificate of Designations) at a meeting of the holders of Common Stock in accordance with the terms of this Agreement and (ii) recommending that the holders of the Common Stock approve the removal of the Conversion Restriction (as such term is defined in the Certificate of Designations) (such recommendation, the "Company Board Recommendation"), which resolutions have not been subsequently rescinded, modified or withdrawn. The affirmative vote (in person or by proxy) of the holders of a majority of the shares of Common Stock (excluding the Shares and any Common Stock issued in exchange for the Shares) voting at the stockholders meeting for the removal of the Conversion Restriction (as such term is defined in the Certificate of Designations) is the only vote or approval of the holders of any class or series of capital stock of the Company or any of its Subsidiaries that is required under the rules and regulations of the SEC, the General Corporation Law of the State of Delaware (the "DGCL") or Nasdaq to approve the transactions contemplated hereby and the consummation thereof, including the conversion of all the Shares under Nasdaq listing rule 5635 (the "Company Stockholder Approval").

(2) Each of the Company's Significant Subsidiaries (as defined in Rule 1-02 of Regulation S-X of the SEC) (i) is duly organized and validly existing under the Laws of its jurisdiction of organization, (ii) has all requisite corporate or other applicable entity power and authority to own its properties and conduct its business as presently conducted, and (iii) is duly qualified to do business and is in good standing in all jurisdictions where its ownership or leasing of property or the conduct of its business requires it to be so qualified, except, in the case of this clause (iii), where failure to be so qualified or in good standing, individually or in the aggregate, has not and would not reasonably be expected to have a Company Material Adverse Effect.

(3) Neither the execution and delivery by the Company of this Agreement or the Ancillary Documents, nor the consummation of the transactions contemplated hereby or thereby, nor compliance by the Company with any of the provisions hereof or thereof will (a) violate or conflict with the organizational documents of the Company, (b) conflict with or violate any Law applicable to the Company or by which any of its properties or assets is bound or subject or (c) result in any breach of, or constitute a default (or event which, with the giving of notice or lapse of time or both, would constitute a default) under, or give to any person any rights of termination, acceleration or cancellation of or result in the creation of any Lien on any of the assets or properties of the Company, any note, bond, mortgage, indenture, deed of trust, license, lease, agreement or other instrument or obligation to which the Company or any of its subsidiaries is a party or by which any of them or any of their respective properties or assets is bound or subject, except, in the case of clauses (b) and (c), for any such conflicts, violations, breaches, defaults, terminations, accelerations or creations as, individually or in the aggregate, would not reasonably be expected to have a Company Material Adverse Effect.

(4) Except for (a) the filing with the SEC of such current reports and other documents, if any, required to be filed with the SEC under the Exchange Act or Securities Act in connection with the transactions contemplated hereunder, including the filing with the SEC of a proxy statement relating to the Company Stockholders' Meeting (the "<u>Proxy Statement</u>"), (b) filings required under, and compliance with other applicable requirements of, the HSR Act, (c) compliance with any applicable state securities or blue sky laws, (d) any filings required by the Registration Rights Agreement or the Amended and Restated Registration Rights Agreement, as in effect at such time, (e) pursuant to the terms of any Contract by and between the Company or any of its Subsidiaries, on the one hand, and any Governmental Entity, on the other hand, and (f) the filing of the Certificate of Designations with the Secretary of State of the State of Delaware, no consent or approval of, or filing or registration with, any Governmental Entity is necessary for the execution, delivery and performance by the Company of this Agreement, other than such other consents, approvals, filings or registrations that, if not obtained, made or given, would not, individually or in the aggregate, reasonably be expected to have a Company Material Adverse Effect.

(5) As of the date hereof, the persons set forth on <u>Schedule 2.1(a)(5)</u> (the "<u>Voting Parties</u>") represent at least 45% of the voting power of the Company entitled to vote with respect to the Company Stockholder Approval.

(b) <u>Capitalization</u>.

(1) The total number of shares of all classes of capital stock which the Company is authorized to issue is 250,000,000 shares, which consists of (a) 245,000,000 shares of Common Stock, which Common Stock consists of (i) 210,000,000 shares of Class A Common Stock and (ii) 35,000,000 shares of Class B Common Stock ("<u>Class B Common Stock</u>") and (b) 5,000,000 shares of preferred stock, par value \$0.0001 per share ("<u>Preferred Stock</u>"). As of the close of business on March 31, 2020 (the "<u>Capitalization Date</u>"), there were 43,354,251 shares of Class A Common Stock outstanding, 30,563,799 shares of Class B Common Stock outstanding and no shares of Preferred Stock outstanding. As of the close of business on the Capitalization Date, (i) 3,037,761 shares of Class A Common Stock remained available for issuance pursuant to the AdaptHealth Corp. 2019 Stock Incentive Plan (the "<u>Stock Plan</u>"), (ii) options to purchase 3,416,666 shares of Class A Common Stock ("<u>Company Stock Options</u>") pursuant to the Stock Plan were outstanding, (iii) 1,486,956 unvested shares of Class A Common Stock remained available for issuance pursuant to the AdaptHealth 2019 Employee Stock Purchase Plan, and (v) public and private warrants to acquire 8,728,036 shares of Class A Common Stock were outstanding (the "<u>Warrants</u>"). All of the issued and outstanding shares of Common Stock have been duly authorized and validly issued and are fully paid, nonassessable and free of preemptive or similar rights. From the Capitalization Date through and as of the date of this Agreement, no other shares of Company Stock Awards in the ordinary course of business of Company Stock Awards in the ordinary course of business. The Company does not have outstanding stockholder purchase rights or "poison pill" or any similar arrangement in effect.

(2) No bonds, debentures, notes or other indebtedness having the right to vote (or convertible into or exchangeable for, securities having the right to vote) on any matters on which the stockholders of the Company may vote ("<u>Voting Debt</u>") are issued and outstanding. Except (i) pursuant to any cashless exercise provisions of any Company Stock Options or pursuant to the surrender of shares to the Company or the withholding of shares by the Company to cover tax withholding obligations under Company Stock Options or Company Stock Awards, (ii) for the Warrants and (iii) as set forth in Section 2.1(b)(1), the Company does not have and is not bound by any outstanding options, preemptive rights, rights of first offer, warrants, calls, commitments or other rights or agreements calling for the purchase, sale or issuance of, or securities or rights convertible into, or exchangeable for, any shares of Compon Stock or any other equity securities of the Company or Voting Debt or any securities representing the right to purchase or otherwise receive any shares of capital stock of the Company (including any rights plan or agreement).

(c) <u>Sale of Securities</u>. Based in part on the Purchaser's representations in Section 2.2, the offer and sale of the Shares is exempt from the registration and prospectus delivery requirements of the Securities Act and the rules and regulations promulgated thereunder. Without limiting the foregoing, neither the Company nor, to the Knowledge of the Company, any other person authorized by the Company to act on its behalf, has engaged in a general solicitation or general advertising (within the meaning of Regulation D of the Securities Act) of investors with respect to offer or sales of the Shares and neither the Company nor, to the Knowledge of the Company, any person acting on its behalf, has made any offers or sales of any security or solicited any offers to buy any security, under circumstances that would cause the offering or issuance of the Shares under this Agreement to be integrated with prior offerings by the Company for purposes of the Securities Act that would cause the offering or issuance of the Shares under this Agreement to be integrated with other offerings.

(d) Shares. The Shares to be delivered to the Purchaser hereunder (including the shares of Common Stock issuable upon conversion of the Purchased Preferred Shares) have been duly authorized and, when issued and paid for pursuant to this Agreement and, if applicable, the Certificate of Designations, shall be validly issued, fully paid and non-assessable and not subject to pre-emptive rights created by statute, the Company's Bylaws or Certificate of Incorporation or any contract to which the Company is a party or is otherwise bound. As of the Closing, the Company shall have the right, authority and power to sell, assign and transfer the Shares to the Purchaser. Upon delivery of such Shares to the Purchaser, the Purchaser shall acquire good, valid and marketable title to the Shares, free and clear of all Liens other than restrictions on transfer imposed by applicable securities Laws or in this Agreement, the Registration Rights Agreement or the Amended and Restated Registration Rights Agreement, as in effect at such time, or in the Certificate of Designations. The Shares to be issued hereunder and the shares of Class A Common Stock issuable upon conversion of the Purchased Preferred Shares have been duly authorized and reserved for such issuance.

(e) <u>SEC Reports; Financial Statements</u>.

(1) The Company has filed, on a timely basis, all forms, reports, prospectuses, proxy statements and documents (together with all amendments thereof and supplements thereto) required to be filed by it with the Securities and Exchange Commission (the "SEC") since February 15, 2018 (together with all exhibits and schedules thereto and all information incorporated therein by reference, the "SEC Reports"). The SEC Reports (as of the date filed with the SEC and, in the case of registration statements, prospectuses and proxy statements, on the dates of effectiveness and the dates of mailing, respectively, and, in the case of any SEC Reports amended or superseded by a filing prior to the date hereof, then on the date of such amending or superseding filing) (i) have complied in all material respects in accordance with either the applicable requirements of the Securities Act or the Exchange Act, as the case may be, and the applicable rules and regulations promulgated by the SEC thereunder and (ii) did not contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading.

(2) As of the date hereof, (A) none of the Company's Subsidiaries is required to file any documents with the SEC, (B) there are no outstanding or unresolved comments in comment letters from the SEC staff with respect to any of the SEC Reports, and (C) to the Company's knowledge, none of the SEC Reports is the subject of ongoing SEC review, outstanding SEC comment or outstanding SEC investigation. Each of the certifications and statements relating to the SEC Reports required by: (x) Rule 13a-14 or Rule 15d-14 under the Exchange Act; (y) 18 U.S.C. §1350 (Section 906 of the Sarbanes-Oxley Act); or (z) any other rule or regulation promulgated by the SEC or applicable to the SEC Reports is accurate and complete, and complete, as to form and content, in all material respects with all applicable Laws.

(3) The consolidated financial statements of the Company, and the related notes thereto, included or incorporated by reference in the SEC Reports, as of the date filed with the SEC (and, in the case of registration statements, prospectuses and proxy statements, on the dates of effectiveness and the dates of mailing, respectively, and, in the case of any SEC Report amended or superseded by a filing prior to the date hereof, then on the date of such amending or superseding filing), have complied as to form in all material respects with applicable accounting requirements and with the published rules and regulations of the SEC with respect thereto, were prepared in accordance with generally accepted accounting principles in the United States ("GAAP") applied on a consistent basis during the periods indicated (except as may be indicated in the notes thereto or, in the case of unaudited statements, as permitted by Form 10-Q of the SEC), and fairly presented, in all material respects (subject, in the case of the unaudited statements, to normal year-end adjustments and the absence of footnote disclosure, none of which, individually or in the aggregate, are material to the Company and its Subsidiaries taken as a whole), the consolidated financial position of the Company and its consolidated subsidiaries as of the date of such financial statements and the consolidated results of their operations and cash flows for each of the periods then ended.

(4) The Company (a) has implemented and maintains disclosure controls and procedures (as defined in Rule 13a-15(e) under the Exchange Act) that are reasonably designed to provide assurance that material information relating to the Company, including its consolidated Subsidiaries, is made known to the individuals responsible for the preparation of the Company's filings with the SEC, and (b) has disclosed, based on its most recent evaluation prior to the date of this Agreement, to the Company's outside auditors and the Board of Director's audit committee (i) any significant deficiencies and material weaknesses in the design or operation of internal controls over financial reporting (as defined in Rule 13a-15(f) under the Exchange Act) that are reasonably likely to adversely affect the Company's ability to record, process, summarize and report material financial information, and (ii) any fraud involving the Company, whether or not material, by management or other employees who have a significant role in the Company's internal controls over financial reporting. As of the date of this Agreement, to the Knowledge of the Company, the outside auditors and its chief executive officer and chief financial officer will be able to give the certifications and attestations required pursuant to the rules and regulations adopted pursuant to Section 404 of the Sarbanes-Oxley Act of 2002, without qualification, when next due.

(5) There is no transaction, arrangement or other relationship between the Company and/or any of its Subsidiaries and an unconsolidated or other offbalance sheet entity that is required to be disclosed by the Company in its SEC Reports and is not so disclosed.

(6) The Proxy Statement (including any amendment or supplement thereto) will comply as to form in all material respects with the requirements of the Exchange Act and will not, at the time it or any amendment or supplement thereto is filed with the SEC or at the time first published, sent or given to the stockholders of the Company, or at the time of the Company Stockholders' Meeting, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they are made, not misleading. Notwithstanding the foregoing, the Company makes no representation or warranty with respect to statements made or incorporated by reference therein based on information supplied by or on behalf of the Purchaser or any Affiliates thereof for inclusion or incorporation by reference in the Proxy Statement.

(f) Brokers and Finders. No broker, finder or investment banker is entitled to any brokerage, finder's or other fee or commission in connection with the consummation of the transactions contemplated by this Agreement based upon arrangements made by or on behalf of the Company or its Affiliates.

(g) Litigation. There are no actions or proceedings or, to the Knowledge of the Company, investigations pending or, to the Knowledge of the Company, any actions, proceedings, or investigations threatened in writing against the Company or any of its Affiliates or any of their respective assets, properties or businesses that (i) question the legality of the transactions contemplated by this Agreement or (ii) individually or in the aggregate would reasonably be expected to have a Company Material Adverse Effect. Except as has not had and would not reasonably be expected to have, individually or in the aggregate, a Company Material Adverse Effect, neither the Company nor any of its Subsidiaries is subject to any judgment, order or decree of any Governmental Entity.

(h) <u>Compliance with Laws</u>. Neither the Company nor any of its Subsidiaries is, or since February 15, 2018 has been, in violation of any applicable Law, except where such violation would not, individually or in the aggregate, reasonably be expected to have, or has not had, a Company Material Adverse Effect. To the Knowledge of the Company as of the date of this Agreement, neither the Company nor any of its Subsidiaries is being investigated with respect to any applicable Law, except for such of the foregoing as would not, individually or in the aggregate, reasonably be expected to have a Company Material Adverse Effect.

(i) <u>Absence of Changes</u>. Since December 31, 2019, there has not been any Company Material Adverse Effect or any event, change or occurrence that would, individually or in the aggregate, reasonably be expected to have a Company Material Adverse Effect.

(j) <u>Listing and Maintenance Requirements</u>. The shares of Class A Common Stock are registered pursuant to Section 12(b) of the Exchange Act and listed on the Nasdaq Global Market ("<u>Nasdaq</u>"), and the Company has taken no action designed to, or which to the Knowledge of the Company is reasonably likely to, have the effect of, terminating the registration of the shares of Class A Common Stock under the Exchange Act or delisting the Class A Common Stock from the Nasdaq nor has the Company received as of the date of this Agreement any notification that the SEC or the Nasdaq is contemplating terminating such registration or de-listing.

(k) <u>Indebtedness</u>. Neither the Company nor any of its Subsidiaries is, immediately prior to the execution and delivery of this Agreement, in default in the payment of any material indebtedness or in material default under any agreement relating to its material indebtedness.

(1) <u>Anti-Takeover Provisions</u>. Assuming the accuracy of the representations in Section 2.2(f), the Company and the Board of Directors have taken all necessary action, if any, in order to render inapplicable any "control share acquisition", "interested stockholder", "business combination", "fair price", "moratorium", or other similar anti-takeover provision under the Certificate of Incorporation, Bylaws or other organizational documents or the Laws of the State of Delaware which is applicable to the Purchaser as a result of the consummation of the transactions contemplated by this Agreement and the Ancillary Documents in the manner contemplated hereby and thereby, including, without limitation, the Company's issuance of the Shares and the Purchaser's ownership of the Shares and the shares of Class A Common Stock issuable upon conversion of the Purchased Preferred Shares.

(m) <u>Investment Company Status</u>. Neither the Company nor any Subsidiary is an "investment company," and, to the Company's knowledge, neither the Company nor any Subsidiary is a company controlled by an "investment company" or an "affiliated person" of, or "promoter" or "principal underwriter" for, an "investment company" as such terms are defined in the Investment Company Act of 1940, as amended.

(n) <u>No Disqualification Events</u>. With respect to the issuance of the Shares, none of the Company, any of its predecessors, any Affiliated issuer, any director, executive officer, other officer of the Company, any beneficial owner of 20% or more of the Company's outstanding voting equity securities, calculated on the basis of voting power, nor any promoter (as that term is defined in Rule 405 under the Securities Act) connected with the Company in any capacity at the time of sale is subject to any of the "Bad Actor" disqualifications described in Rule 506(d)(1)(i) to (viii) under the Securities Act except for items covered by Rule 506(d)(2) or (d)(3).

(o) <u>No Plan Assets</u>. Neither the Company nor any of its Subsidiaries holds "plan assets" within the meaning of the Department of Labor regulations located at 29 C.F.R. Section 2510.3-101, as modified by Section 3(42) of the Employee Retirement Income Security Act of 1974, as amended ("<u>Plan Assets</u>").

(p) <u>No Additional Representations</u>. Except for the representations and warranties made by the Company in this Section 2.1, none of the Company or any of its Affiliates or representatives makes any other representation or warranty of any kind or nature whatsoever, oral or written, express or implied, with respect to itself, its Affiliates, their respective businesses, this Agreement or the transactions contemplated by the Agreement.

2.2 <u>Representations and Warranties of the Purchaser</u>. The Purchaser hereby represents and warrants to the Company that:

(a) Incorporation and Authority. The Purchaser is duly organized, validly existing and in good standing under the Laws of its jurisdiction of organization. The Purchaser has all requisite corporate or other applicable organizational power to (i) enter into, consummate the transactions contemplated by, and carry out its obligations under this Agreement, and (ii) own, lease and operate its properties and carry on its business as it is now being conducted and is duly qualified to do business and is in good standing in all jurisdictions where its ownership or leasing of property or the conduct of its business requires it to be so qualified, except for any failure under clause (ii) that would not, individually or in the aggregate, reasonably be expected to adversely affect the Purchaser's ability to perform its obligations under this Agreement or the Ancillary Documents or consummate the transactions contemplated hereby or thereby on a timely basis. The execution and delivery by the Purchaser of this Agreement and the Ancillary Documents and the consummation by the Purchaser of the transactions contemplated by this Agreement and the Ancillary Documents will be duly authorized by all requisite corporate or other similar organization, execution and delivery by the other parties hereto, this Agreement constitutes and the Ancillary Documents will constitute the legal, valid and binding obligation of the Purchaser, enforceable against it in accordance with their respective terms, subject in each case to the effect of any applicable bankruptcy, reorganization, insolvency, moratorium or similar Laws now or hereafter in effect relating to or affecting creditors' rights and remedies generally and subject, as to enforceability, to the effect of general equitable principles (regardless of whether enforcement is sought in a proceeding in equity or at law).

(b) <u>Non-Contravention</u>.

(1) Neither the execution, delivery and performance by the Purchaser of this Agreement or the Ancillary Documents, nor the consummation of the transactions contemplated hereby or thereby, nor compliance by the Purchaser with any of the provisions hereof or thereof, will (A) violate, conflict with, or result in a breach of any provision of, or constitute a default (or an event which, with notice or lapse of time or both, would constitute a default) under, or result in the termination of, or accelerate the performance required by, or result in a right of termination or acceleration of, or result in the creation of any Lien upon any of the properties or assets of the Purchaser under any of the terms, conditions or provisions of (i) its governing instruments or (ii) any note, bond, mortgage, indenture, deed of trust, license, lease, agreement or other instrument or obligation to which the Purchaser is a party or by which it may be bound, or to which the Purchaser or any of the properties or assets of the Purchaser may be subject, or (B) subject to compliance with the statutes and regulations referred to in the next paragraph, violate any law, statute, ordinance, rule or regulation, permit, concession, grant, franchise or any judgment, ruling, order, writ, injunction or decree applicable to the Purchaser or any of their respective properties or assets except in the case of clauses (A)(ii) and (B) for such violations, conflicts and breaches as would not reasonably be expected to materially and adversely affect the Purchaser's ability to perform its respective obligations under this Agreement or consummate the transactions contemplated hereby on a timely basis.

(2) Other than (A) the securities or blue sky laws of the various states, (B) filings required under, and compliance with other applicable requirements of, the HSRAct, and (C) the filing by the Company of the Certificate of Designations with the Delaware Secretary of State, no notice to, registration, declaration or filing with, exemption or review by, or authorization, order, consent or approval of, any Governmental Entity, nor expiration or termination of any statutory waiting period, is necessary for the consummation by the Purchaser of the transactions contemplated by this Agreement or the Ancillary Documents.

(c) <u>Securities Law Matters</u>.

(1) The Purchaser acknowledges that the Shares have not been registered under the Securities Act or under any state securities laws. The Purchaser (A) acknowledges that it is acquiring the Shares pursuant to an exemption from registration under the Securities Act solely for investment with no present intention to distribute any of the Shares to any person in violation of applicable securities laws, (B) is acquiring the Shares only for its own account and not for the account of others, and not on behalf of any other account or person, (C) will not sell, transfer, pledge or otherwise dispose of any of the Shares, except in compliance with the registration requirements or exemption provisions of the Securities Act, any other applicable securities laws, (D) has such knowledge and experience in financial and business matters and in investments of this type that it is capable of evaluating the merits and risks of its investment in the Shares and of making an informed investment decision, (E) is an institutional "accredited investor" (as that term is defined by Rule 501 of the Securities Act) or is a "qualified institutional buyer" (as that term is defined in Rule 144A of the Securities Act), in each case of this clause (E), satisfying the requirements set forth on <u>Schedule B</u>, and (F) can bear the economic risk of (x) an investment in the Shares indefinitely and (y) a total loss in respect of such investment. The Purchaser has such knowledge and experience in business and financial matters so as to enable it to understand and evaluate the risks of and form an investment decision with respect to, its investment in the Shares and to protect its own interest in connection with such investment.

(2) In making its decision to purchase the Shares, the Purchaser has relied solely upon independent investigation made by the Purchaser. The Purchaser acknowledges and agrees that the Purchaser has had the opportunity to review the SEC Reports and the Purchaser has received such information as it deems necessary in order to make an investment decision with respect to the Shares. The Purchaser represents and agrees that the Purchaser's professional advisor(s), if any, have had the full opportunity to ask such questions, receive such answers and obtain such information as the Purchaser and the Purchaser's professional advisor(s), if any, have deemed necessary to make an investment decision with respect to the Shares.

(3) The Purchaser became aware of this offering of the Shares solely by means of direct contact between it and the Company or a representative of the Company, and the Shares were offered to the Purchaser solely by direct contact between it and the Company or a representative of the Company. The Purchaser did not become aware of this offering of the Shares, nor were the Shares offered to the Purchaser, by any other means. The Purchaser acknowledges that it was not induced to purchase the Shares through any form of general solicitation or general advertising.

(4) The Purchaser acknowledges that it is aware that there are risks incident to the purchase and ownership of the Shares. The Purchaser has such knowledge and experience in financial and business matters as to be capable of evaluating the merits and risks of an investment in the Shares, and the Purchaser has sought such accounting, legal and tax advice as the Purchaser has considered necessary to make an informed investment decision.

(5) The Purchaser is not (i) a person or entity named on the List of Specially Designated Nationals and Blocked Persons administered by the U.S. Treasury Department's Office of Foreign Assets Control or in any Executive Order issued by the President of the United States and administered by OFAC ("<u>OFAC List</u>"), or a person or entity prohibited by any OFAC sanctions program, (ii) a Designated National as defined in the Cuban Assets Control Regulations, 31 C.F.R. Part 515, or (iii) a non-U.S. shell bank or providing banking services indirectly to a non-U.S. shell bank. The Purchaser agrees to provide law enforcement agencies, if requested thereby, such records as required by applicable law, provided that the Purchaser is permitted to do so under applicable law. If the Purchaser is a financial institution subject to the Bank Secrecy Act (31 U.S.C. Section 5311 et seq.), as amended by the USA PATRIOT Act of 2001, and its implementing regulations (collectively, the " <u>BSA/PATRIOT Act</u>"), the Purchaser maintains policies and procedures reasonably designed to comply with applicable obligations under the BSA/PATRIOT Act. To the extent required, it maintains policies and procedures reasonably designed to ensure that the funds held by the Purchaser and used to purchase the Shares were legally derived.

(6) None of (i) the Purchaser, (ii) any of its directors, executive officers, other officers that may serve as a director or officer of any company in which it invests, general partners or managing members, nor (iii) any beneficial owner of the Company's voting equity securities (in accordance with Rule 506(d) of the Securities Act) held by the Purchaser is subject to any of the "Bad Actor" disqualifications described in Rule 506(d)(1)(i) to (viii) under the Securities Act (a "Disqualification Event"), except for Disqualification Events covered by Rule 506(d)(2)(ii) or (iii) or (d)(3) under the Securities Act and disclosed reasonably in advance of the Closing in writing in reasonable detail to the Company.

(7) The Purchaser does not hold Plan Assets.

(8) The Purchaser understands that the Shares are being offered and sold to it in reliance on specific exemptions from the registration requirements of the United States federal and state securities laws and that the Company is relying in part upon the truth and accuracy of, and Purchaser's compliance with, the representations, warranties, agreements, acknowledgments and understandings of such Purchaser set forth herein in order to determine the availability of such exemptions.

(d) <u>Financial Capability</u>. The Purchaser has and at the Closing will have available funds necessary to consummate the Closing on the terms and conditions contemplated by this Agreement. The Purchaser is not aware as of the date hereof of any reason why the funds sufficient to fulfill its obligations under <u>Article I</u> will not be available at the Closing. The Purchase Price to be paid by Purchaser, together with the purchase price for any shares of Class A Common Stock held by the Purchaser prior to the Closing, is less than the maximum amount that Purchaser or Guarantor is permitted to invest in any one portfolio investment pursuant to the terms of its organizational or governing documents or otherwise. Purchaser has uncalled capital commitments or otherwise has available funds in excess of the Purchase Price and all other unfunded contractually binding equity commitments of Purchaser that are currently outstanding.

(e) <u>Brokers and Finders</u>. Neither the Purchaser nor its Affiliates or any of their respective officers, directors, employees or agents has employed any broker or finder for which the Company will incur any liability for any financial advisory fees, brokerage fees, commissions or finder's fees.

(f) Ownership of Common Stock. As of the date of this Agreement, except as publicly disclosed by the Purchaser prior to the date hereof, neither the Purchaser nor any of its Affiliates owns (directly or indirectly, beneficially or of record) any shares of Class A Common Stock and neither the Purchaser nor any of its Affiliates of Class A Common Stock except pursuant to this Agreement. Neither the Purchaser nor any of its "affiliates" or "associates" is or has been, within three years of the date hereof, an "interested stockholder" of the Company, as those terms are defined in Section 203 of the DGCL, or has taken any action that would cause any anti-takeover statute under the DGCL to be applicable to this Agreement or any of the transactions contemplated hereby. As of the date hereof, there are no Contracts between the Purchaser, on the one hand, and any member of the Company's management or directors, on the other hand, that relate in any way to the Company or the transactions contemplated hereby.

(g) <u>No Additional Representations</u>. Except for the representations and warranties made by the Company in this Section 2.2, none of the Purchaser or any of its Affiliates or representatives makes any other representation or warranty of any kind or nature whatsoever, oral or written, express or implied, with respect to itself, its Affiliates, their respective businesses, this Agreement, any Ancillary Document or the transactions contemplated by the Agreement.

ARTICLE III

COVENANTS

3.1 <u>Confidentiality</u>. Subject to the terms and conditions set forth in the Confidentiality Agreement, which shall remain in full force and effect in accordance with its terms, each party to this Agreement will hold, and will cause its respective Affiliates and their respective directors, managers, officers, employees, agents, consultants and advisors to hold, in strict confidence, unless disclosure to a regulatory authority is reasonably necessary in connection with any reasonably necessary regulatory approval, examination or inspection or unless disclosure is required by judicial or administrative process or by other requirement of law or the applicable requirements of any regulatory agency or relevant stock exchange (in which case, other than in connection with a disclosure in connection with a routine audit or examination by, or document request from, a regulatory or self-regulatory authority, bank examiner or auditor, the party disclosing such information shall provide the other party with prior written notice of such permitted disclosure), all non-public records, books, contracts, instruments, computer data and other data and information (collectively, "Information") concerning the other party hereto furnished to it by or on behalf of such other party or its representatives pursuant to this Agreement (except to the extent that such information, can be shown to have been (1) previously known by such party from other sources, provided that such source was not known, after reasonable inquiry and investigation, by such party or (3) later lawfully acquired from other sources by the party to which it was furnished), and neither party hereto shall release or disclose such Information to any other person, except its auditors, attorneys, financial advisors, financial sources and other consultants and advisors.

3.2 Listing. At or prior to the Closing, the Company shall use its reasonable best efforts to cause the Purchased Common Shares and the shares of Class A Common Stock issuable upon the conversion of the Purchased Preferred Shares issued to Purchaser pursuant to this Agreement and pursuant to the Certificate of Designations to be approved for listing on Nasdaq, subject to official notice of issuance, following the Closing Date. From time to time following the Closing Date, in the event the number of shares of Class A Common Stock into which the Series A Preferred Stock is convertible increases under the Certificate of Designations, the Company shall apply to cause the number of shares of Class A Common Stock issuable upon conversion of the then outstanding shares of Series A Preferred Stock to be approved for listing on the Nasdaq.

Efforts. Subject to the other terms and conditions of this Agreement, each of the parties hereto shall use its respective reasonable best efforts to take, or cause 3.3 to be taken, all actions and to do, or cause to be done, all things reasonably necessary, proper or advisable under this Agreement and applicable Law to, as promptly as reasonably practicable following the date of this Agreement, consummate the Closing. In furtherance of the foregoing, to the extent required under the HSR Act, the Purchaser and the Company agree to each promptly, and in any event within ten (10) business days, following the date hereof make any filings required under the HSR Act and agree to supply as promptly as reasonably practicable any additional information and documentary material that may be requested pursuant to the HSR Act and to take all other actions necessary, proper or advisable to cause the expiration or termination of the applicable waiting periods under the HSR Act as soon as practicable, including, to the extent then available, by requesting early termination of the waiting period provided for under the HSR Act. Each party shall, in connection therewith, use its commercially reasonable efforts to: (i) cooperate in all respects with the other party or its affiliates in connection with any filing or submission and in connection with any investigation or other inquiry, including any proceeding initiated by a private person; (ii) keep the other party reasonably informed of any communication received by such party or its representatives from or given by such party or its representatives to, any governmental authority and of any communication received or given in connection with any proceeding by a private person, in each case regarding the purchase of the Shares; (iii) permit a representative of the other party and their respective outside counsel to review any communication given by it to, and consult with each other in advance of any meeting or conference with, any governmental authority or, in connection with any proceeding by a private person, with any other person, and to the extent permitted by such governmental authority or other person, give a representative or representatives of the other party the opportunity to attend and participate in such meetings and conferences; (iv) in the event a party's representative is prohibited from participating in or attending any meetings or conferences, the other party shall keep such party promptly and reasonably apprised with respect thereto; and (v) use commercially reasonable efforts to cooperate in the filing of any memoranda, white papers, filings, correspondence or other written communications explaining or defending the purchase of the Shares, articulating any regulatory or competitive argument, and/or responding to requests or objections made by any governmental authority.

3.4 Legend. The Purchaser agrees that all certificates or other instruments (including book-entry notations) representing the Shares or shares of Class A Common Stock issued upon conversion of the Series A Preferred Stock subject to this Agreement will bear a legend substantially to the following effect:

THE SECURITIES REPRESENTED HEREBY HAVE NOT BEEN REGISTERED UNDER THE SECURITIES ACT OF 1933, AS AMENDED (THE "SECURITIES ACT"), OR ANY STATE SECURITIES LAWS AND NEITHER THE SECURITIES NOR ANY INTEREST THEREIN MAY BE OFFERED, SOLD, TRANSFERRED, PLEDGED OR OTHERWISE DISPOSED OF EXCEPT PURSUANT TO AN EFFECTIVE REGISTRATION STATEMENT UNDER THE SECURITIES ACT OR SUCH LAWS OR AN EXEMPTION FROM REGISTRATION UNDER THE SECURITIES ACT AND SUCH LAWS WHICH, IN THE OPINION OF COUNSEL FOR THIS CORPORATION, IS AVAILABLE.

Upon request of the Purchaser, upon receipt by the Company of an opinion of counsel reasonably satisfactory to the Company to the effect that such legend is no longer required under the Securities Act, the Company shall promptly cause the legend to be removed from any Shares so requested.

3.5 Back Leverage.

(a) Subject to the additional terms and conditions set forth on <u>Schedule 3.5</u>, to the extent reasonably requested by the Purchaser, the Company will provide customary and reasonable cooperation in good faith as may be reasonably required in connection with the Purchaser obtaining any Back Leverage, including entering into a waiver of transfer restrictions in customary form to allow for Liens to be placed on the Shares.

(b) The Purchaser acknowledges and agrees that, notwithstanding the Company's obligations under this <u>Section 3.5</u>, neither the obtaining of the Back Leverage, nor the completion of any issuance of debt contemplated by the Back Leverage, is a condition to the Closing.

3.6 <u>Corporate Actions</u>.

(a) At any time that any Series A Preferred Stock is outstanding, the Company shall from time to time take all lawful action within its control to cause the authorized capital stock of the Company to include a sufficient number of authorized but unissued shares of Class A Common Stock to satisfy the conversion requirements of all shares of the Series A Preferred Stock then outstanding, based solely on the conversion rate then in effect.

(b) Prior to or upon the Closing, the Company shall file with the Secretary of State of the State of Delaware the Certificate of Designations in the form attached hereto as <u>AnnexI</u>.

(c) If any occurrence since the date of this Agreement until the Closing would have resulted in an adjustment to the Conversion Rate pursuant to the Certificate of Designations if the Series A Preferred Stock had been issued and outstanding since the date of this Agreement, the Company shall adjust the Conversion Rate, effective as of the Closing, in the same manner as would have been required by the Certificate of Designations if the Series A Preferred Stock had been issued and outstanding since the date of this Agreement.

(d) At any time that any Series A Preferred Stock is outstanding, the Company shall not adopt any stockholder rights agreement, "poison pill" or similar antitakeover agreement or plan that is applicable to Purchaser unless the Company has excluded Purchaser from the definition of "acquiring person" (or such similar term) as such term is defined in such anti-takeover agreement to the extent of Purchaser's beneficial ownership of Preferred Stock or Common Stock owned as of the date any such agreement or plan is adopted by the Company.

3.7 <u>Negative Covenants</u>. Except as required by applicable Law or to comply with any notice from a Governmental Entity, as expressly contemplated, required or permitted by this Agreement or as described in Section 3.7 of the Company Disclosure Schedule, during the period from the date of this Agreement until (x) in the case of clauses (b) through (e) below, the Closing Date (or such earlier date on which this Agreement may be terminated pursuant to Section 6.15) and (y) in the case of clause (a) below, the date on which the first Company Stockholders' Meeting is held for the Company Stockholder Approval (or such earlier date on which this Agreement may be terminated pursuant to Section 6.15), unless Purchaser otherwise consents in writing (such consent not to be unreasonably withheld, delayed or conditioned), the Company shall not:

(a) other than the authorization and issuance of the Series A Preferred Stock to Purchaser and the consummation of the other transactions contemplated hereunder and under the Acquisition Agreement, issue, sell or grant any shares of its capital stock or other equity or voting interests, or any securities or rights convertible into, exchangeable or exercisable for, or evidencing the right to subscribe for any shares of its capital stock or other equity or voting interests, or any rights, warrants or options to purchase any shares of its capital stock or other equity or voting Parties would beneficially own less than 45% of the outstanding equity interests eligible to vote for the Company Stockholder Approval;

(b) establish a record date for, declare, set aside for payment or pay any dividend on, or make any other distribution in respect of, any shares of its capital stock or other equity or voting interests;

(c) amend, alter, repeal or otherwise modify (whether by merger, consolidation or otherwise) any provision of the Certificate of Incorporation in a manner that would adversely affect the powers, preferences, rights or privileges of the Series A Preferred Stock or the Purchaser's investment therein

- (d) take any action prohibited as provided on <u>Schedule 3.7;</u> or
- (e) agree or commit to do any of the foregoing.

3.8 <u>Tax Matters</u>. The Purchaser and the Company agree (i) not to treat the Series A Preferred Stock (based on its terms as set forth in the Certificate of Designations) as "preferred stock" within the meaning of Section 305 of the Code and Treasury Regulation Section 1.305-5 and (ii) to not take a reporting position that the holders of Series A Preferred Stock are required to include as dividend income any amounts in respect of the Series A Preferred Stock unless and until dividends on the Series A Preferred Stock are paid in cash or other property, in each case, for United States federal income tax and withholding tax purposes.

3.9 Stockholder Approval.

As promptly as reasonably practicable after the execution of this Agreement, and in any event no later than the first to occur of (x) 30th day following the date (a) upon which the Company receives the Target Financial Statements and (y) the 15th day following the Closing, the Company shall prepare the Proxy Statement in preliminary form and file it with the SEC. The Board of Directors shall recommend to the Company's stockholders that the holders of the Common Stock approve the transactions contemplated hereunder and shall include such recommendation in the Proxy Statement. The Purchaser shall provide to the Company all information concerning the Purchaser and its respective Affiliates as may be reasonably requested by the Company in connection with the Proxy Statement and shall otherwise assist and cooperate with the Company in the preparation of the Proxy Statement and the resolution of any comments thereto received from the SEC. Each of the Company and the Purchaser shall promptly correct any information provided by it for use in the Proxy Statement if and to the extent such information shall have become false or misleading in any material respect. The Company shall notify the Purchaser promptly upon the receipt of any comments from the SEC and of any request by the SEC for amendments or supplements to the Proxy Statement and shall supply the Purchaser with copies of all written correspondence between the Company or any of its representatives, on the one hand, and the SEC, on the other hand, with respect to the Proxy Statement. The Company shall use its reasonable best efforts to respond as promptly as reasonably practicable to any comments received from the SEC concerning the Proxy Statement and to resolve such comments with the SEC, and shall use its reasonable best efforts to cause the Proxy Statement to be disseminated to its stockholders as promptly as reasonably practicable after the resolution of any such comments. Prior to the filing of the Proxy Statement (or any amendment or supplement thereto) or any dissemination thereof to the stockholders of the Company, or responding to any comments from the SEC with respect thereto, the Company shall provide the Purchaser with a reasonable opportunity to review and to propose comments on such document or response, which the Company shall consider in good faith. The Company shall inform the Purchaser promptly (and in any event within two Business Days) of the receipt of the Target Financial Statements (or the purported delivery of the Target Financial Statements by Solara Holdings, LLC).

(b) Subject to Section 3.9(a), the Company shall take all necessary actions in accordance with applicable Law, the organizational documents of the Company and the rules of Nasdaq to duly call, give notice of, convene and hold a meeting of its stockholders (including any adjournment, recess or postponement thereof, the "<u>Company Stockholders' Meeting</u>") for the purpose of obtaining the Company Stockholder Approval, as soon as reasonably practicable after the SEC confirms that it has no further comments on the Proxy Statement. The Company shall not submit any Additional Transaction for approval or adoption by the stockholders of the Company prior to (x) (but the Company may submit an Additional Transaction contemporaneously with) the first Company Stockholders' Meeting for the Company Stockholder Approval or (y) if earlier, the termination of this Agreement in accordance with its terms if termination occurs prior to the Closing. The Company shall use its reasonable best efforts to obtain the Company Stockholders' Meeting (i) after consultation with the Purchaser, to the extent necessary to ensure that any required supplement or amendment to the Proxy Statement is provided to the stockholders of the Company within a reasonable amount of time in advance of the Company Stockholder' Meeting, (ii) if as of the time for which the Company Stockholders' Meeting is originally scheduled (as set forth in the Proxy Statement) there are insufficient shares of Common Stock represented (either in person or by proxy) to constitute a quorum necessary to conduct the business of the Company Stockholders' Meeting or (iii) to solicit additional proxies if the Company reasonably believes it may be necessary to obtain the Company Stockholder Approval.

3.10 <u>Guarantee</u>.

(a) The Guarantee. The Guarantor hereby irrevocably and unconditionally guarantees to the Company the obligations of the Purchaser contained in this Agreement and the Ancillary Documents in accordance with the terms and conditions set forth herein and therein, including the due and punctual payment of all monetary obligations (including the payment of Purchase Price) of the Purchaser pursuant to this Agreement and the Ancillary Documents. In case of the failure of the Purchaser punctually to pay any such monetary obligation, the Guarantor hereby agrees to cause any such payment to be made punctually when and as the same shall become due and payable, and as if such payment were made by the Purchaser. The Guarantor shall cause the punctual performance and discharge by the Purchaser of, and the compliance by the Purchaser with, all covenants, agreements, obligations and undertakings of the Purchaser under this Agreement and the Ancillary Documents in accordance with the terms hereof and thereof.

(b) <u>Guarantee Unconditional</u>. The Guarantor hereby agrees that its obligations hereunder shall be as principal and not merely as surety, and shall be absolute, irrevocable and unconditional, irrespective of, and shall be unaffected by, any invalidity, irregularity or unenforceability of this Agreement or any Ancillary Document, any failure to enforce the provisions of this Agreement or any Ancillary Document, or any waiver, modification, consent or indulgence granted with respect thereto by the Company, the recovery of any judgment against the Purchaser or any action to enforce the same, or any other circumstances which may otherwise constitute a legal or equitable discharge of a surety or guarantor. The Guarantor hereby waives diligence, presentment, demand of payment, filing of claims with a court in the event of merger, insolvency or bankruptcy of the Purchaser, any right to require a proceeding first against the Purchaser, protest or notice with respect to any obligations guaranteed pursuant to Section 3.10(a) and all demands whatsoever, and covenants that the obligations under this Section 3.10 will not be discharged except by payment in full of any obligations guaranteed pursuant to Section 3.10(a).

(c) <u>Reinstatement</u>. The guarantees in this Section 3.10 shall continue to be effective or be reinstated, as the case may be, if at any time any obligations guaranteed pursuant to Section 3.10(a), in whole or in part, is rescinded or must otherwise be restored to the Purchaser or the Guarantor upon the bankruptcy, liquidation or reorganization of the Purchaser or otherwise.

(d) <u>Subrogation</u>. The Guarantor shall be subrogated to all rights of the Company against the Purchaser in respect of any amounts paid to the Company by the Guarantor pursuant to the provisions of this Section 3.10; provided, however, that the Guarantor shall not be entitled to enforce, or to receive any payments arising out of or based upon, such right of subrogation until the obligations guaranteed pursuant to Section 3.10(a), when and as the same shall become due and payable according to the terms of this Agreement or any Ancillary Documents shall have been paid in full.

(e) <u>Representations and Warranties of the Guarantor.</u>

(i) <u>Authority</u>. The Guarantor has all requisite power and authority to enter into this and to provide the undertakings contemplated by this Section 3.10. All acts and proceedings required to be taken to authorize the execution, delivery, and performance by the Guarantor of this Agreement and provide the undertakings contemplated by this Section 3.10 have been duly and properly taken. This Agreement has been duly executed, and delivered by the Guarantor and, assuming due authorization, execution and delivery by Seller, constitutes a valid and binding obligation of the Guarantor, enforceable against the Guarantor in accordance with its terms, except as such enforceability may be limited by bankruptcy, insolvency, moratorium and other similar laws affecting creditors' rights generally and by general principles of equity.

(ii) <u>Non Contravention</u>. The execution, delivery, and performance by the Guarantor of this Agreement does not, and the consummation by the Guarantor of its obligations under this Section 3.10 will not, (x) conflict with, or result in any violation of, any provision of the certificate of limited partnership or limited partnership agreement (or similar organizational documents) of the Guarantor, or (y) conflict with, result in any violation of, or constitute a default under, any Contract to which the Guarantor is a party or by which the Guarantor or the property or assets of the Guarantor is bound, or any judgment, order, writ, injunction, or decree to which the Guarantor has been specifically identified as subject, or result in any violation of any statute, law, ordinance, rule, or regulation applicable to the Guarantor or its property or assets (except where such conflict, violation, or default would not, reasonably be expected to, individually or in the aggregate, materially impair or delay the ability of the Guarantor to perform its obligations under this Section 3.10(a).

(iii) <u>Financial Capacity</u>. The Guarantor has, and will have, the financial capacity to pay and perform its obligations under this Section 3.10, including payment and satisfaction of obligations guaranteed pursuant to Section 3.10(a).

ARTICLE IV

SURVIVAL

4.1 <u>Survival</u>. The representations and warranties of the parties contained in this Agreement shall survive for twelve (12) months following the Closing. All of the covenants or other agreements of the parties contained in this Agreement to be performed prior to the Closing shall terminate at the Closing. All covenants or agreements of the parties contained in this Agreement to be performed following the Closing shall survive until fully performed or fulfilled, unless and to the extent that non-compliance with such covenants or agreements is waived in writing by the party entitled to such performance.

ARTICLE V

SHAREHOLDER RIGHTS

5.1 Board Matters.

Promptly following the Closing Date, and in any event within five (5) business days following the Closing Date, the Company shall increase the size of the (a) Board of Directors of the Company (the "Board of Directors") by two (2) members and, promptly following such increase, and in any event within five (5) business days following the Closing Date, shall cause Brad Coppens to be elected or appointed to the Board of Directors (the "Purchaser Board Representative") and one independent director nominated by the Company or Board of Directors to be elected or appointed to the Board of Directors (the "Independent Board Representative" and together with the Purchaser Board Representative, the "Board Representatives"). The election or appointment of the Board Representatives will be subject to satisfaction of all legal and governance requirements regarding service as a director of the Company and to the reasonable approval of the Nominating and Governance Committee of the Board of Directors (such approval not to be unreasonably withheld, conditioned or delayed); provided, that the parties hereto agree that election of Brad Coppens to serve as the Purchaser Board Representative shall be acceptable and no such approval of the Nominating and Governance Committee shall be required. For so long as the Purchaser or an Affiliate of Purchaser holds at least 25% of the Shares, on an as-converted basis, issued to the Purchaser hereunder, on an as-converted basis, the Purchaser shall continue to have the right to nominate the Purchaser Board Representative. Subject to this Section 5.1(a), so long as the Purchaser has the right to nominate the Purchaser Board Representative to the Board of Directors, the Company shall not increase the number of Directors on the Board of Directors without the prior consent of the Purchaser. The Company will reimburse the Purchaser Board Representative and any observer appointed pursuant to Section 5.1(b) for their respective reasonable and documented out-of-pocket expenses incurred in connection with travel to or from and attendance at each meeting of the Board of Directors. The Purchaser Board Representative will receive the same director compensation as each other non-executive director of the Board of Directors. The Purchaser agrees, upon the Company's request, to timely provide the Company with accurate and complete information relating to the Purchaser Board Representative as may be required to be disclosed by the Company under the Exchange Act and the rules and regulations promulgated thereunder. Notwithstanding any rights to be granted with respect to the Purchaser Board Representative hereunder, the Board of Directors may exclude the Purchaser Board Representative from access to any Board of Directors or committee materials or information or meeting or portion thereof or written consent if the Board of Directors determines, in good faith, including the Purchaser Board Representative in discussions relating to such determination (but not requiring the affirmative vote of such Purchaser Board Representative), that such access would reasonably be expected to result in a conflict of interest with the Company; provided, that such exclusion shall be limited to the portion of the Board of Directors or committee material or information and/or meeting or written consent that is the basis for such exclusion and shall not extend to any portion of the Board of Directors or committee material and/or meeting that does not involve or pertain to such exclusion.

Following the Closing Date and for so long as the Purchaser or an Affiliate of Purchaser holds at least 50% of the Shares, on an as-converted basis, issued to (b) the Purchaser hereunder, on an as-converted basis, the Purchaser shall have the right (in addition to the rights set forth in Section 5.1(a)) to designate a non-voting observer of the Board of Directors, which observer shall be an employee or partner of the Purchaser or one of its Affiliates. The observer shall be entitled to receive notice of and have the right to attend any and all meetings of the Board of Directors, and the Company shall provide the observer with copies of all notices, minutes, consents and other material in connection therewith at the same time as such materials are distributed to members of the Board of Directors; provided, that (A) the Purchaser shall cause the observer to agree to hold in confidence and trust and to act in a fiduciary manner with respect to all information provided to the observer pursuant hereto and (B) the Company and the Board of Directors shall have the right to withhold any information and to exclude the observer from any meeting or portion thereof (1) if doing so is, in the opinion of outside counsel to the Company, advisable or necessary to protect the attorney-client privilege between the Company and counsel or (2) if the Board of Directors determines in good faith, after consultation with outside counsel, that fiduciary requirements under applicable law would make attendance by the observer not advisable; provided, further, that, if the Company and/or the Board of Directors withhold any information or exclude the observer from any meeting pursuant to the foregoing clause (B), to the extent practicable they shall give Purchaser notice of such withholding or exclusion and the parties shall cooperate in seeking to allow disclosure of such information in a manner that is not reasonably likely (in the good faith belief of the Company and the Board of Directors (after consultation with outside counsel)) to contravene such applicable law or cause such privilege to be waived. The observer shall have no right to vote on any matters presented to the Board of Directors. All obligations of the Company pursuant to this Section 5.1(b) shall terminate, and, upon request by the Board of Directors, the Purchaser shall cause the observer to resign promptly from the Board of Directors, in each case upon the Purchaser ceasing to have the right to designate the observer pursuant to this Section 5.1.

22
23
20

(c) For so long as the Purchaser has the right to designate or nominate the Purchaser Board Representative pursuant to <u>Section 5.1(a)</u>, the Company and the Nominating and Governance Committee of the Board of Directors shall take such action as is required under applicable law, the rules and regulations in effect at such time of Nasdaq or such other market on which the Class A Common Stock is then listed or quoted or under the Bylaws or Certificate of Incorporation to include on the Board of Directors or in the slate of nominees recommended by the Board of Directors such person designated or nominated, as the case may be, by the Purchaser pursuant to <u>Section 5.1</u>. The Company shall use its reasonable best efforts to have the Purchaser Board Representative elected as a director of the Company and the Company shall solicit proxies for such person to the same extent as it does for any of its other nominees to the Board of Directors. For so long as the Purchaser has the right to designate or nominate the Purchaser Board Representative, in the event that a vacancy is created at any time by the death, disability, retirement, resignation or removal of the Purchaser Board Representative, the Purchaser may designate or nominate, as applicable, another individual to be elected to fill the vacancy created thereby, and the Company hereby agrees to take, at any time and from time, all actions necessary to accomplish the same. For so long as the Purchaser Board Representative pursuant to Section 5.1(a) and the Purchaser Board Representative is a member of the Board of Directors; <u>provided</u>, that if applicable SEC rules or Nasdaq listing rules do not permit the Purchaser Board Representative to be a member of such committee, then the Board of Directors; <u>provided</u>, that if applicable SEC rules or Nasdaq listing rules do not permit the Purchaser Board Representative is a anon-voting observer of such committee if designated by Purchaser.

(d) The Company shall maintain in effect at all times directors' and officers' indemnity insurance covering the Purchaser Board Representative to the same extent and on the same terms as any directors' and officers' indemnity insurance maintained by the Company with respect to the other non-executive members of the Board of Directors. Any directors' and officers' indemnity insurance shall be primary to any insurance coverage for the Purchaser Board Representative maintained by any other person. At the Closing, the Company and the Purchaser Board Representative shall enter into an indemnification agreement substantially similar to the form included as Exhibit 10.4 to the Company's Form 10-K for the fiscal year ended December 31, 2019 (the "Indemnification Agreement").

(e) All obligations of the Company pursuant to this Section 5.1 relating to the Purchaser Board Representative shall terminate, and, upon request by the Board of Directors, the Purchaser shall cause the Purchaser Board Representative to resign promptly from the Board of Directors, in each case upon the Purchaser ceasing to have the right to designate or nominate a director pursuant to <u>Section 5.1</u>. Any vacancy created by such resignation may be filled by the Board of Directors or the shareholders of the Company in accordance with the Bylaws or Certificate of Incorporation and applicable law.

5.2 Information Rights. From and after the date hereof and for so long as the Purchaser or an Affiliate of Purchaser holds at least 25% of the Shares, on an asconverted basis, issued to the Purchaser hereunder, on an as-converted basis:

(a) The Company will provide to the Purchaser the annual, quarterly and periodic reports of the Company in the time periods required for the filing of such reports with the SEC as soon as reasonably practicable after they become available; provided, however, that the filing of such reports with the SEC shall satisfy such delivery requirement.

(b) The Company will permit the Purchaser and its representatives, at the Purchaser's expense, to reasonable and customary rights to the information of the Company and access to management of the Company, all at such reasonable times and as often as may be reasonably requested.

5.3 <u>Transfer Restrictions</u>.

(a) Other than Permitted Transfers, the Purchaser shall not Transfer any Shares, or the shares of Class A Common Stock issuable upon conversion of the Purchased Preferred Shares (together, the "<u>Prohibited Shares</u>") during the period commencing on the Closing Date and continuing until the date that is the eighteen (18) month anniversary of the Closing Date (such date, the "<u>Restricted Period Termination Date</u>"). Notwithstanding the foregoing, in the event that following the six (6) month anniversary of the Closing Date (x) the volume weighted average price of the Class A Common Stock exceeds \$22.50 (as adjusted for share splits, share dividends, reorganizations, recapitalizations, reclassifications, combinations, exchanges of shares or other like changes) for twenty (20) consecutive trading days, 50% of the Prohibited Shares shall no longer be subject to the Transfer restrictions set forth in the first sentence of this Section 5.3(a) and (y) the volume weighted average price of the Class A Common Stock exceeds \$30.00 (as adjusted for share splits, share dividends, reorganizations, reclassifications, combinations, recapitalizations, reclassifications, combinations, exchanges of shares or other like changes) for twenty (20) consecutive trading days, 50% of the Prohibited Shares shall no longer be subject to the Transfer splits, share dividends, reorganizations, reclassifications, combinations, exchanges of shares or other like changes) for twenty (20) consecutive trading days, 100% of the Prohibited Shares shall no longer be subject to the Transfer restrictions set forth in the first sentence of this Section 5.3(a) (in either case, a "<u>Restricted Period Early Termination Event</u>").

(b) "<u>Permitted Transfer</u>" means, in each case so long as such Transfer is in accordance with applicable Law:

(1) a Transfer of Prohibited Shares to Affiliates of the Purchaser, so long as such transferee, to the extent it has not already done so, executes a customary joinder to this Section 5.3, in form and substance reasonably acceptable to the Company, in which such transferee agrees to be subject to the restrictions on Transfer in this Section 5.3;

(2) a Transfer of Prohibited Shares as a distribution in-kind to the Purchaser's investors, and to their subsequent investors, including limited partners, so long as all such transferees, to the extent it has not already done so, executes a customary joinder to this Section 5.3, in form and substance reasonably acceptable to the Company, in which such transferee agrees to be subject to the restrictions on Transfer in this Section 5.3;

(3) a Transfer of Prohibited Shares in connection with the exercise of piggyback registration rights set forth in the Registration Rights Agreement or the Amended and Restated Registration Rights Agreement, as in effect at such time;

(4) a Transfer of Prohibited Shares in connection with a sale of the Company approved by the Board of Directors or in connection with a tender offer into which a majority of the Unaffiliated Shareholders of the Company have tendered their respective shares of Common Stock;

- (5) a Transfer of Prohibited Shares to the Company; and
- (6) a Transfer of Prohibited Shares following a voluntary filing by the Company of a petition for relief under the United States Bankruptcy Code.

(c) Notwithstanding anything to the contrary contained herein, including the occurrence of the Restricted Period Termination Date, the Purchaser shall not Transfer any Prohibited Shares (i) other than in accordance with all applicable Laws and the other terms and conditions of this Agreement and (ii) other than any Transfers that are made pursuant to a registered offering or in accordance with an exemption from registration. Any Transfers of Prohibited Shares shall only be made to a person that, to Purchaser's knowledge, is not a competitor of the Company set forth on <u>Schedule C</u> attached hereto and that does not become a competitor of the Company after the date hereof, as set forth on a quarterly or annual report filed by the Company with the SEC.

(d) The Company may impose stop-transfer instructions and may stamp each certificate representing the Prohibited Shares with an appropriate legend to enforce the provisions of this Section 5.3. Any purported Transfer or other transaction in violation of this Section 5.3 shall be null and void.

5.4 <u>Standstill</u>.

(a) For so long as the Purchaser Beneficially Owns Shares representing at least twenty five percent (25%), on an as-converted basis, of the Shares issued to the Purchaser pursuant to this Agreement, on an as-converted basis, without the prior written consent of the Company, the Purchaser shall not, and shall cause its Affiliates not to, directly or indirectly:

(1) acquire, offer to acquire or agree or make a proposal to acquire Beneficial Ownership of Common Stock or any Derivative Instruments with respect to Common Stock, except pursuant to share splits, reverse share splits, share dividends or distributions, or combinations or any similar recapitalizations on or after the date hereof or offerings made available to holders of Common Stock generally on a pro rata basis including;

(2) effect or seek, offer or propose to effect, or announce any intention to effect or cause or participate in (A) any acquisition of any securities (or beneficial ownership thereof) or material assets of the Company or any of its Affiliates, including rights or options to acquire such ownership, in connection with any "change of control" of the Company; (B) initiate or propose any merger, tender offer, business combination, restructuring, recapitalization or other extraordinary transaction involving, or any change of control of, the Company or any of its Subsidiaries; (C) any shareholder proposal (including any precatory proposal) to be considered by the stockholders of the Company, or take any action to nominate any person for membership on the Board of Directors, or take any action to remove any director (other than the Purchaser Board Representative) from the Board of Directors of the Company or to change the composition of the board of directors of the Company or (D) make, or in any way participate in, directly or indirectly, any "solicitation" of "proxies" to vote, or seek to influence any person with respect to the voting of, Shares, or become a "participant" in a "solicitation" (as such terms are defined in Regulation 14A under the Exchange Act) with respect to Common Stock; provided that the foregoing shall not restrict the Purchaser's right to vote its Common Stock in its sole discretion;

(3) deposit any Common Stock into a voting trust or subject Common Stock to any proxy, arrangement or agreement with respect to the voting of such securities or other agreement having a similar effect (other than a proxy provided to the management of the Company in connection with an annual or special meeting of shareholders);

(4) initiate or propose a call for any special meeting of the Company's shareholders;

(5) otherwise act, alone or in concert with others, to seek or propose to influence, advise, change or control the management, Board of Directors, governing instruments or policies of the Company or any of its Subsidiaries;

(6) propose, or agree to, or enter into any discussions, negotiations or arrangements with, or provide any confidential information to, any third party with respect to any of the foregoing;

(7) form, join or in any way participate in any "group" (within the meaning of Section 13(d) of the Exchange Act and the rules and regulations promulgated thereunder) with respect to any securities of the Company or otherwise in connection with any of the foregoing;

(8) disclose any intention, plan, proposal or arrangement inconsistent with any of the foregoing;

(9) advise, assist, knowingly encourage or direct any other person to do any of the foregoing, or act as a financing source for or otherwise invest in any person in connection with such person doing any of the foregoing;

(10) take any action that could reasonably be expected to require the Company to make a public announcement regarding the possibility of a transaction with you or any of your affiliates or any of the other events described in this paragraph; or

(11) take any action challenging the validity or enforceability of this paragraph, or request the Company amend or waive any provision of this paragraph (including this clause (11)), provided, that the Purchaser may make confidential requests to the Company to amend or waive any of the limitations set forth in this Section 5.4(a), which the Company may accept or reject in its sole discretion, so long as any such request is made in a manner that does not require the public disclosure thereof by any person.

(b) The prohibition in Section 5.4(a)(1) shall not apply in connection with acquisitions made as a result of a share split, share dividend, reorganization, recapitalization, reclassification, combination, exchange of shares or other like change and shall not restrict (1) the ability of the Purchaser Board Representative to vote or from otherwise exercising his or her fiduciary duties, (2) the Purchaser's ability to vote, Transfer, convert or otherwise exercise rights under its Shares subject to the express obligations hereof.

5.5 Right of First Offer. Except for a Permitted Transfer, in the event of any proposed sale of shares of Series A Preferred Stock to a third party by Purchaser, the provisions of this Section 5.5 shall apply:

If and whenever Purchaser desires to sell shares of Series A Preferred Stock to a third party (other than a Permitted Transfer) (any such transaction, a (a) "Proposed Sale"), Purchaser shall first deliver to the Company written notice thereof (a "ROFO Notice"). The ROFO Notice shall disclose (i) Purchaser's bona fide intention to effectuate such Proposed Sale and (ii) the number of shares of Series A Preferred Stock to be sold. The Company may elect to offer to purchase all (but not less than all) of the shares of Series A Preferred Stock to be sold in the Proposed Sale by giving written notice of such offer (an "Offer Notice"), to Purchaser within ten (10) business days after the ROFO Notice has been delivered to the Company (the "Offer Period"). If the Purchaser timely accepts the Offer Notice, then Purchaser shall be obligated to sell pursuant to the Offer Notice, and the Company obligated to purchase pursuant to the Offer Notice, in each case, for the ROFO Price, and the Company and Purchaser shall use commercially reasonable efforts to effect the consummation of the Proposed Sale of such shares of Series A Preferred Stock contemplated by the Offer Notice as promptly as practicable, but, in any event, within fifteen (15) business days following delivery of the Offer Notice. If the Company notifies Purchaser that the Company does not wish to purchase the applicable shares of Series A Preferred Stock specified in the ROFO Notice, or if the Company fails, within the Offer Period, to notify Purchaser of its response to the ROFO Notice, then Purchaser will have a period of ninety (90) days after delivery of the ROFO Notice to consummate the Proposed Sale at a price per share of Series A Preferred Stock that is equal to or greater than the ROFO Price. If the aforesaid ninety (90) day period expires without Purchaser consummating the Proposed Sale, the foregoing rights of the Company under this Section 5.5 will apply with respect to such Proposed Sale and any other potential Proposed Sale. As used herein, the "ROFO Price" per share of Series A Preferred Stock means the greater of (x) the Conversion Price and (y) the amount obtained by multiplying (i) the Closing Price (as defined in the Certificate of Designations) of the Class A Common Stock on the trading day immediately prior to the closing of the sale of such shares of Series A Preferred Stock times (ii) the number of shares of Class A Common Stock into which a share of Series A Preferred Stock is then convertible.

(b) Purchaser will, in its sole discretion, decide whether or not to pursue, consummate, postpone or abandon any sale of shares of Series A Preferred Stock and the terms and conditions thereof. Neither Purchaser, nor any of its Affiliates will have any liability to any other holder of securities of the Company arising from, relating to or in connection with the pursuit, consummation, postponement, abandonment or terms and conditions of any sale of shares of Series A Preferred Stock except to the extent Purchaser will have failed to comply with the provisions of this Section 5.5 or any other binding agreement.

(c) The closing of a sale of shares of Series A Preferred Stock to which Section 5.5 applies will take place at such time and place as the Company will specify by notice to the Purchaser. At the closing of such sale of shares of Series A Preferred Stock, Purchaser will deliver the certificates evidencing the shares of Series A Preferred Stock, if any, to be sold by Purchaser, duly endorsed for transfer with signature guaranteed, free and clear of any liens or encumbrances, with any necessary transfer tax stamps affixed, against delivery of the applicable consideration. The purchase price specified in any Offer Notice shall be payable solely in immediately available funds at the closing of the transaction. For the avoidance of doubt, this Section 5.5 shall not apply to any sale of shares of the Class A Common Stock issuable upon conversion of the Series A Preferred Stock.

(d) For the avoidance of doubt, compliance with this Section 5.5 shall not eliminate or otherwise relieve the Purchaser from any restrictions on Transfer applicable to the Shares.

ARTICLE VI

MISCELLANEOUS

6.1 Expenses. Each of the parties will bear and pay all costs and expenses incurred by it or on its behalf in connection with the transactions contemplated pursuant to this Agreement, including the fees and expenses of attorneys, accountants, investment bankers and consultants; provided that the Company shall, upon the Closing of the transaction contemplated hereby reimburse the Purchaser for its reasonable and documented out-of-pocket third-party costs and expenses incurred in connection with due dilgence, the negotiation and preparation of this Agreement and undertaking of the transactions contemplated pursuant to this Agreement, including any documentary, stamp and similar issue or transfer tax due on the issuance of the Shares to Purchaser, and excluding any costs or expenses of obtaining any Back Leverage; provided that the maximum amount of such reimbursable costs and expenses shall not exceed \$500,000 in the aggregate. In addition, the Purchaser shall be responsible for any fees and expenses (including filing fees and fees and expenses of counsel) incurred in connection with making the filings and notifications required by the HSR Act, and the reimbursement set forth in the prior sentence shall not apply to any such fees and expenses.

6.2 <u>Amendment; Waiver</u>. No amendment or waiver of any provision of this Agreement will be effective with respect to any party unless made in writing and signed by an officer of a duly authorized representative of such party. No failure or delay by any party in exercising any right, power or privilege hereunder shall operate as a waiver thereof nor shall any single or partial exercise thereof preclude any other or further exercise thereof or the exercise of any other right, power or privilege. The conditions to each party's obligation to consummate the Closing are for the sole benefit of such party and may be waived by such party in whole or in part to the extent permitted by applicable law. No waiver of any party to this Agreement will be effective unless it is in a writing signed by a duly authorized officer of the waiving party that makes express reference to the provision or provisions subject to such waiver. The rights and remedies herein provided shall be cumulative and not exclusive of any rights or remedies provided by law.

6.3 <u>Counterparts; Electronic Transmission</u>. For the convenience of the parties hereto, this Agreement may be executed in any number of separate counterparts, each such counterpart being deemed to be an original instrument, and all such counterparts will together constitute the same agreement. Executed signature pages to this Agreement may be delivered by facsimile or other means of electronic transmission and such facsimiles or other means of electronic transmission will be deemed as sufficient as if actual signature pages had been delivered.

6.4 <u>Governing Law</u>. This Agreement will be governed by and construed in accordance with the laws of the State of Delaware. The parties hereby irrevocably and unconditionally consent to submit to the exclusive jurisdiction of the state and federal courts located in the State of Delaware for any actions, suits or proceedings arising out of or relating to this Agreement and the transactions contemplated hereby. The parties hereby irrevocably and unconditionally consent to the jurisdiction of such courts (and of the appropriate appellate courts therefrom) in any such action, suit or proceeding and irrevocably waive, to the fullest extent permitted by law, any objection that they may now or hereafter have to the laying of the venue of any such action, suit or proceeding in any such court or that any such action, suit or proceeding may be served on any party anywhere in the world, whether within or without the jurisdiction of any such court. Without limiting the foregoing, each party agrees that service of process on such party as provided in Section 6.6 shall be deemed effective service of process on such party.

6.5 WAIVER OF JURY TRIAL. EACH OF THE PARTIES HERETO HEREBY IRREVOCABLY WAIVES ANY AND ALL RIGHT TO TRIAL BY JURY IN ANY LEGAL PROCEEDING ARISING OUT OF OR RELATED TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY.

6.6 <u>Notices</u>. Any notice, request, instruction or other document to be given hereunder by any party to the other will be in writing and will be deemed to have been duly given (a) on the date of delivery if delivered personally or by telecopy, facsimile or electronic mail (so long as such transmission does not generate an error message or notice of non-delivery), (b) on the first business day following the date of dispatch if delivered by a recognized next-day courier service, or (c) on the third business day following the date of mailing if delivered by registered or certified mail, return receipt requested, postage prepaid. All notices hereunder shall be delivered as set forth below, or pursuant to such other instructions as may be designated in writing by the party to receive such notice.

If to Purchaser:

c/o One Equity Partners 510 Madison Avenue, 19th Floor New York, NY 10022 Attn: Gregory A. Belinfanti, Brad Coppens, Inna Etinberg, Jessica Marion and Dora Stojka Email: gregory.belinfanti@oneequity.com brad.coppens@oneequity.com jessica.marion@oneequity.com dora.stojka@oneequity.com

with a copy to:

Kirkland & Ellis LLP 300 North LaSalle Street Chicago, Illinois 60654 Attn: Jeremy S. Liss, P.C. Ross M. Leff, P.C. Jeffrey P. Swatzell Email: jeremy.liss@kirkland.com ross.leff@kirkland.com jeffrey.swatzell@kirkland.com

If to the Company:

AdaptHealth Corp. 220 West Germantown Pike Suite 250 Plymouth Meeting, PA 19462 Attention: General Counsel E-mail: cjoyce@adapthealth.com

with a copy to:

Willkie Farr & Gallagher LLP 787 Seventh Avenue New York, NY 10019 Attention: Steven J. Gartner Michael E. Brandt Danielle Scalzo E-mail: sgartner@willkie.com dscalzo@willkie.com Facsimile: 212-728-9962

6.7 Entire Agreement. This Agreement (including the Schedules hereto and the documents and instruments referred to in this Agreement), together with the Confidentiality Agreement, constitutes the entire agreement among the parties and supersedes all other prior agreements and understandings, both written and oral, between the parties with respect to the subject matter hereof and transactions contemplated hereby.

6.8 <u>Assignment</u>. Neither this Agreement, nor any of the rights, interests or obligations hereunder shall be assigned by any of the parties hereto (whether by operation of Law or otherwise) without the prior written consent of the other party; provided, however, that the Purchaser may, without the prior written consent of the other party; assign its rights, interests and obligations under this Agreement, in whole or in part, either pursuant to a Permitted Transfer or to an Affiliate, and, in each case, in the event of such assignment, the assignee shall agree in writing to be bound by the provisions of this Agreement, including the rights, interests and obligations so assigned; provided that, in each case, no such assignment will relieve the Purchaser of its obligations hereunder.

6.9 <u>Interpretation; Other Definitions</u>. Wherever required by the context of this Agreement, the singular shall include the plural and vice versa, and the masculine gender shall include the feminine and neuter genders and vice versa, and references to any agreement, document or instrument shall be deemed to refer to such agreement, document or instrument as amended, supplemented or modified from time to time. All article, section, paragraph or clause references not attributed to a particular document shall be references to such parts of this Agreement, and all exhibit, annex, letter and schedule references not attributed to a particular document shall be references to such exhibits, annexes, letters and schedules to this Agreement. In addition, the following terms are ascribed the following meanings:

(a) the word "<u>or</u>" is not exclusive;

(b) the words "includes," "included" and "include" are deemed to be followed by the words "without limitation";

(c) the terms "herein," "hereof" and "hereunder" and other words of similar import refer to this Agreement as a whole and not to any particular section, paragraph or subdivision;

(d) "Beneficial Owner," "Beneficially Own" or "Beneficial Ownership" has the meaning assigned to such term in Rule 13d-3 under the Exchange Act, and a person's Beneficial Ownership of securities shall be calculated in accordance with the provisions of such Rule (in each case, irrespective of whether or not such Rule is actually applicable in such circumstance).

(e) the term "business day" means any day except Saturday, Sunday and any day which shall be a legal holiday or a day on which banking institutions in the State of New York or State of Pennsylvania generally are authorized or required by law or other governmental action to close; and

(f) the term "person" has the meaning given to it in Section 3(a)(9) of the Exchange Act and as used in Sections 13(d)(3) and 14(d)(2) of the Exchange Act.

(g) the term "<u>made available</u>" means that the information referred to has been posted in the Merrill Queen "data room" established by the Company or its representatives or in publicly available SEC Reports filed by the Company with the SEC on or following the Reference Date at least two days prior to the date hereof, in each case.

(h) "Additional Transaction" means, in each case involving any person other than the Purchaser or one of its Affiliates, any (i) sale, lease, assignment, exchange or other transfer or disposition directly or indirectly by merger, consolidation, business combination, share exchange, joint venture or otherwise of assets of the Company or any Subsidiary; (ii) issuance, sale or other disposition, directly or indirectly (including, without limitation, by way of merger, consolidation, business combination, share exchange, joint venture or any similar transaction), of securities (or options, rights, or warrants to purchase, or securities convertible into or exchangeable for, such securities), including without limitation capital stock, partnership interests or other instruments directly or indirectly convertible into, exchangeable or exercisable for, or the value of which is determined with reference to, capital stock, partnership interests or membership interests ("Equity Securities") of the Company or any of its Subsidiaries; (iii) tender offer or exchange offer as defined pursuant to the Exchange Act that, if consummated, would result in any person beneficially owning any class or series (or the voting power of any class or series) of Equity Securities; or (iv) combination of the foregoing (in each case, other than the arrangements contemplated hereunder).

(i) "<u>Affiliate</u>" means, with respect to any specified person, any other person that, at the time of determination, directly or indirectly through one or more intermediaries, controls, is controlled by or is under common control with such specified person.

(j) "<u>as-converted basis</u>" means, for purposes of computing beneficial ownership, such number of shares of Common Stock calculated on a basis assuming all shares of Series A Preferred Stock had been converted by the holders thereof pursuant to Section 6 of the Certificate of Designations classifying the Series A Preferred Stock, but disregarding any restrictions or limitations upon the conversion of such Series Preferred Stock, including those limitations specified in Section 8(f) thereof.

(k) "<u>Amended and Restated Registration Rights Agreement</u>" means the Amended and Restated Registration Rights Agreement attached hereto as <u>Annex II</u>; provided, that the Company and the holders of a majority of Registrable Securities (as defined in the form of Amended and Restated Registration Rights Agreement) shall be entitled to make amendments to the form of Amended and Restated Registration Rights Agreement to the extent that such amendments could be made following the effectiveness thereof after giving effect to the acquisition by Purchaser of the Shares contemplated hereby.

(1) "Back Leverage" means the (1) incurrence of indebtedness by the Purchaser (or an Affiliate thereof) to (A) finance a portion of its purchase of the Shares, (B) finance a return of capital with respect to its investment in the Shares, or (C) refinance or replace indebtedness described in this clause (1), and (2) granting of Liens by the Purchaser to secure payment of such indebtedness, including on Shares held by the Purchaser (or each Affiliate of the Purchaser to which Shares are transferred pursuant to Section 5.3(b)(1)).

(m) "Common Stock" means the common stock, par value \$0.0001 per share, of the Company.

"Company Material Adverse Effect" means any change, effect, event, occurrence, condition, state of facts or development that, either alone or in combination, (n) has had, or would be reasonably expected to have, (a) a materially adverse effect on the business, operations, assets, liabilities or condition (financial or otherwise) or results of operations of the Company, taken as a whole; provided, however, that none of the following shall constitute or be deemed to contribute to a Company Material Adverse Effect, or shall otherwise be taken into account in determining whether a Company Material Adverse Effect has occurred or would be reasonably likely to occur: any adverse effect arising out of, resulting from or attributable to (i) (A) the economy generally or credit, currency, oil, financial, banking, securities, capital markets or financial markets generally (including any increased cost, or decreased availability, of capital or pricing or terms related to any financing for the transactions contemplated hereunder or any disruption thereof and any decline in the price of any security, commodity or market index), including changes in interest or exchange rates, and (B) changes or conditions generally affecting the industry or markets in which the Company participates, (ii) any changes or prospective changes in applicable Law, GAAP, or the enforcement or interpretation thereof after the date hereof or any action required to be taken under any Law by which the Company or its Subsidiaries (or any of their respective assets or properties) is bound, (iii) any international or national political, regulatory or social conditions, hostilities, cyber-attack, act of war, sabotage, terrorism, declaration of national emergency or military actions, or any escalation or worsening of any such hostilities, cyber-attack, act of war, sabotage, terrorism, declaration of national emergency or military actions, (iv) the failure of the Company to meet or achieve the results set forth in any internal budget, plan, projection or forecast; provided that this clause (iv) will not prevent a determination that any change, effect or other cause underlying such failure to meet budgets, plans, projections or forecasts has resulted in or contributed to a Company Material Adverse Effect, (v) actions of the Company expressly required by the terms of this Agreement or taken with the prior written consent of the Purchaser, (vi) the negotiation or execution of this Agreement or any other Ancillary Document or announcement, pendency or consummation of this Agreement or the transactions contemplated hereby or thereby or the identity, nature or ownership of the Purchaser, including the impact thereof on the relationships, contractual or otherwise, of the Company or any of its Subsidiaries with any of its or their business relations or employees, (vii) epidemics, pandemics or disease or virus outbreaks (including the COVID-19 virus) or (viii) hurricanes, earthquakes, tsunamis, tornados, mudslides, floods or other natural disasters, weather conditions, explosions or fires or other force majeure events or acts of God, whether or not caused by any person, or any national or international calamity or crisis; provided that the matters described in clauses (i), (iii) and (viii) shall be included and taken into account in the term "Company Material Adverse Effect" to the extent any such matter has a disproportionate adverse impact on the business, operations, assets, liabilities or condition (financial or otherwise) or results of operations of the Company, taken as a whole, relative to the other participants in the industries in which they operate; and (b) a material impairment on or material delay in the ability of the Company to perform its material obligations under this Agreement or to consummate the transactions contemplated by this Agreement.

(o) "<u>Confidentiality Agreement</u>" means that certain Non-Disclosure Agreement, dated as of February 27, 2020, by and between the Company and the Purchaser.

(p) "<u>Contract</u>" means any contract, agreement, instrument, undertaking, indenture, commitment, loan, license, settlement, consent, note or other legally binding obligation (whether or not in writing).

- (q) "<u>Conversion Price</u>" has the meaning set forth in the Certificate of Designations.
- (r) "<u>Conversion Rate</u>" has the meaning set forth in the Certificate of Designations.
 - 34

(s) "Derivative Instruments" means any and all derivative securities (as defined under Rule 16a-1 under the Exchange Act) that increase in value as the value of any Equity Securities of the Company increases, including a long convertible security, a long call option and a short put option position, in each case, regardless of whether (i) such derivative security conveys any voting rights in any Equity Security, (ii) such derivative security is required to be, or is capable of being, settled through delivery of any Equity Security or (iii) other transactions that hedge the value of such derivative security.

(t) "Effect" means any change, event, effect, development or circumstance.

(u) "Encumbrance" means any mortgage, commitment, transfer restriction, deed of trust, pledge, option, power of sale, retention of title, right of pre-emption, right of first refusal, executorial attachment, hypothecation, security interest, encumbrance, claim, lien or charge of any kind, or an agreement, arrangement or obligation to create any of the foregoing.

(v) "Equity Securities" means any and all (i) shares, interests, participations or other equivalents (however designated) of capital stock or other voting securities of a corporation, any and all equivalent or analogous ownership (or profit) or voting interests in a person (other than a corporation), (ii) securities convertible into or exchangeable for shares, interests, participations or other equivalents (however designated) of capital stock or voting securities of a corporation, and securities convertible into or exchangeable for any equivalent or analogous ownership (or profit) or voting interests in a person (other than a corporation), and securities convertible into or exchangeable for any equivalent or analogous ownership (or profit) or voting interests in a person (other than a corporation), and (iii) any and all warrants, rights or options to purchase any of the foregoing, whether voting or nonvoting, and, in each case, whether or not such shares, interests, participations, equivalents, securities, warrants, options, rights or other interests are authorized or otherwise existing on any date of determination; provided that Equity Securities shall not include preferred stock that are not convertible or exchangeable for common stock in a corporation.

(w) "Fundamental Representations" means the representations and warranties set forth in Sections 2.1(a)(1), (b)(1), (c), (d) and (f).

(x) "<u>Governmental Entity</u>" means any court, administrative or regulatory agency or commission or other governmental or arbitral body or authority or instrumentality, including any state-controlled or owned corporation or enterprise, in each case whether federal, state, local or foreign, and any applicable industry self-regulatory organization.

(y) "<u>HSR Act</u>" means the United States Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, and the rules and regulations promulgated thereunder.

(z) "Knowledge of the Company" means the actual knowledge after reasonable inquiry of one or more of Luke McGee, Joshua Parnes, Christopher Joyce, Wendy Russalesi, Gregg Holst, John Gentile, Shaw Rietkerk or Andy Palan.

(aa) "<u>Law</u>" means any federal, state, local or foreign law, statute or ordinance, or any rule, code, treaty, constitution, regulation, judgment, order, writ, injunction, ruling, decree, administrative interpretation or agency requirement of any Governmental Entity.

(bb) "<u>Lien</u>" means any mortgage, deed of trust, pledge, hypothecation, security interest, encumbrance, claim, escheat, encroachment, lien, charge of any kind, option, easement, purchase right, right of first refusal, right of pre-emption, conditional sale agreement, covenant, condition or other similar restriction (including restrictions on transfer) or any agreement to create any of the foregoing.

(cc) "<u>Outside Date</u>" means the date that is 150 days after the date hereof.

(dd) "<u>Purchased Common Shares</u>" means a number of shares of Class A Common Stock equal to the lesser of (i) the Stock Consideration Amount and (ii) the maximum number of shares of Common Stock as are issuable by the Company without obtaining prior shareholder approval pursuant to Nasdaq listing rule 5635 (assuming that the shares of Common Stock issuable in connection with the Target Acquisition are issued at the Closing without any reduction therefrom as a result of Nasdaq listing rule 5635) (which, under this clause (ii) shall not be less than 9,911,932 shares of Class A Common Stock).

(ee) "<u>Purchased Preferred Shares</u>" means a number of shares of Series A Preferred Stock equal to the quotient obtained by dividing (i) (x) the Stock Consideration Amount, less (y) the number of Purchased Common Shares issued hereunder, divided by (ii) the Conversion Rate.

- (ff) "<u>Purchase Price Per Common Share</u>" means \$13.75.
- (gg) "Purchase Price Per Preferred Share" means \$1,000.

(hh) "Registration Rights Agreement" means that certain Registration Rights Agreement, dated as of November 8, 2019, by and among AdaptHealth Holdings LLC and certain other parties thereto.

(ii) "Stock Consideration Amount" means 13,818,182 (subject to reduction as set forth in Section 1.1(a)).

(j) "Subsidiary" means, with respect to any person, (a) any corporation of which a majority of the total voting power of shares of capital entitled (without regard to the occurrence of any contingency) to vote in the election of directors, managers or trustees thereof is at the time owned or controlled, directly or indirectly, by such person or one or more of the other Subsidiaries of such person or a combination thereof, or (b) any partnership, limited liability company, association or other business entity of which a majority of the partnership, limited liability company or other similar ownership interest is at the time owned or controlled, directly or indirectly, by such person or one or more Subsidiaries of such person or a combination thereof. For purposes of this definition, a person is deemed to have a majority ownership interest in a partnership, limited liability company, association or other business entity of other business entity of the gains or losses of such partnership, limited liability company, association or other business entity or is or controls the managing member or general partner or similar position of such partnership, limited liability company, association or other business entity.

(kk) "Target Acquisition" means the acquisition by the Company or one of its Subsidiaries of Solara Holdings, LLC.

(II) "<u>Target Financial Statements</u>" means for purposes of Section 3.9, the 2019 Audited Financials (as defined in the Stock Purchase Agreement as in effect on the date hereof).

(mm) "<u>Transfer</u>" means (i) any direct or indirect sale, lease, assignment, Encumbrance, disposition or other transfer (by operation of law or otherwise), either voluntary or involuntary, or entry into any Contract, option or other arrangement or understanding with respect to any sale, lease, assignment, Encumbrance, disposition or other transfer (by operation of law or otherwise), of any Equity Security or (ii) to enter into any Derivative Instrument, swap or any other Contract, agreement, transaction or series of transactions that hedges or transfers, in whole or in part, directly or indirectly, the economic consequence of ownership of any Equity Security, whether any such Derivative Instrument, swap, Contract, agreement, transaction or series of transactions is to be settled by delivery of securities, in cash or otherwise; provided, however, that, notwithstanding anything to the contrary in this Agreement, a Transfer shall not include the conversion of one or more shares of Series A Preferred Stock into shares of Class A Common Stock pursuant to the Certificate of Designations.

(nn) "<u>Unaffiliated Shareholders</u>" means the shareholders of the Company, other than (1) the Purchaser, (2) any Affiliates or representatives of the Purchaser or any person acting for or on behalf of the Purchaser or (3) any shareholder that is a member of a "group" (within the meaning of Section 13(d) of the Exchange Act and the rules and regulations promulgated thereunder) with the Purchaser.

(00) "Voting Agreements" means those certain Voting Agreements, dated as of the date hereof, by and among the Purchaser and each of the Voting Parties.

6.10 <u>Captions</u>. The article, section, paragraph and clause captions herein are for convenience of reference only, do not constitute part of this Agreement and will not be deemed to limit or otherwise affect any of the provisions hereof.

6.11 Severability. If any provision of this Agreement or the application thereof to any person (including the officers and directors of the parties hereto) or circumstance is determined by a court of competent jurisdiction to be invalid, void or unenforceable, the remaining provisions hereof, or the application of such provision to persons or circumstances other than those as to which it has been held invalid or unenforceable, will remain in full force and effect and shall in no way be affected, impaired or invalidated thereby, so long as the economic or legal substance of the transactions contemplated hereby is not affected in any manner materially adverse to any party. Upon such determination, the parties shall negotiate in good faith in an effort to agree upon a suitable and equitable substitute provision to effect the original intent of the parties.

6.12 <u>No Third Party Beneficiaries</u>. Nothing contained in this Agreement, expressed or implied, is intended to confer upon any person other than the parties hereto (and their permitted assigns), any benefit, right or remedies.

6.13 <u>Public Announcements</u>. Subject to each party's disclosure obligations imposed by law or regulation or the rules of any stock exchange upon which its securities are listed, each of the parties hereto will cooperate with each other in the development and distribution of all news releases and other public information disclosures with respect to this Agreement and any of the transactions contemplated by this Agreement, and neither the Company nor the Purchaser will make any such news release or public disclosure without first consulting with the other, and, in each case, also receiving the other's consent (which shall not be unreasonably withheld, conditioned or delayed) and each party shall coordinate with the party whose consent is required with respect to any such news release or public disclosure.

6.14 <u>Specific Performance</u>. The parties agree that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. It is accordingly agreed that, without the necessity of posting bond or other undertaking, the parties shall be entitled to specific performance of the terms hereof, this being in addition to any other remedies to which they are entitled at law or equity, and in the event that any action or suit is brought in equity to enforce the provisions of this Agreement, and no party will allege, and each party hereby waives, the defense or counterclaim that there is an adequate remedy at law.

- 6.15 <u>Termination</u>. Prior to the Closing, this Agreement may only be terminated:
- (a) by mutual written agreement of the Company and the Purchaser;

(b) by the Company or the Purchaser, upon written notice to the other party, if the Closing has not occurred by the Outside Date; provided, however that the right to terminate this Agreement pursuant to this Section 6.15(b) shall not be available to any party whose failure to materially comply with any of its obligations under this Agreement shall have been the cause of, or shall have resulted in, the failure of the Closing to occur on or prior to such date;

(c) by the Company if (i) the Purchaser shall have breached any representation, warranty, covenant or agreement of the Purchaser set forth in this Agreement, (ii) such breach or misrepresentation is not cured or capable of being cured by the Outside Date, and (iii) such breach or misrepresentation would cause any of the conditions set forth in Section 1.3(c)(1) or 1.3(c)(2) not to be satisfied; *provided* that the Company is not then in breach of this Agreement so as to cause the conditions to the Closing set forth in either Section 1.3(a) or Section 1.3(b) to not (in the absence of a waiver) be satisfied as of the Closing Date; or

(d) by the Purchaser if (i) the Company shall have breached any representation, warranty, covenant or agreement of the Company set forth in this Agreement, (ii) such breach or misrepresentation is not cured or capable of being cured by the Outside Date, and (iii) such breach or misrepresentation would cause any of the conditions set forth in Section 1.3(b)(1) or 1.3(b)(2) not to be satisfied; *provided* that the Purchaser is not then in breach of this Agreement so as to cause the conditions to the Closing set forth in either Section 1.3(a) or Section 1.3(c) to not (in the absence of a waiver) be satisfied as of the Closing Date.

6.16 Effects of Termination. In the event of any termination of this Agreement in accordance with Section 6.15, neither party (or any of its Affiliates) shall have any liability or obligation to the other (or any of its Affiliates) under or in respect of this Agreement, except to the extent of (A) any liability arising from any willful and material breach by such party of its obligations of this Agreement arising prior to such termination and (B) any fraud or intentional or willful breach of this Agreement. In the event of any such termination, this Agreement shall become void and have no effect, and the transactions contemplated hereby shall be abandoned without further action by the parties hereto, in each case, except (x) as set forth in the preceding sentence and (y) that the provisions of Sections 3.1 (Confidentiality), 6.2 to 6.14 (Amendment; Waiver; Counterparts; Electronic Transmission; Governing Law; Waiver of Jury Trial; Notices; Entire Agreement; Assignment; Interpretation; Other Definitions; Captions; Severability; No Third Party Beneficiaries; Public Announcements; and Specific Performance), this Section 6.16 (Effects of Termination) and Section 6.17 (Non-Recourse) shall survive the termination of this Agreement.

6.17 Non-Recourse. This Agreement may only be enforced against, and any claims or causes of action that may be based upon, arise out of or relate to this Agreement, or the negotiation, execution or performance of this Agreement may only be made against the entities that are expressly identified as parties hereto, including entities that become parties hereto after the date hereof, including permitted assignees and successors, or that agree in writing for the benefit of the Company to be bound by the terms of this Agreement applicable to the Purchaser, and no former, current or future equityholders, controlling persons, directors, officers, employees, agents or Affiliates of any party hereto or any former, current or future equityholder, controlling person, director, officer, employee, general or limited partner, member, manager, advisor, agent or Affiliate of any of the foregoing (each, a "Non-Recourse Party") shall have any liability for any obligations or liabilities of the parties to this Agreement or for any claim (whether in tort, contract or otherwise) based on, in respect of, or by reason of, the transactions contemplated hereby or in respect of any representations made or alleged to be made in connection herewith. Without limiting the rights of any party against the other parties hereto, in no event shall any party or any of its Affiliates seek to enforce this Agreement against, make any claims for breach of this Agreement against, or seek to recover monetary damages from, any Non-Recourse Party.

IN WITNESS WHEREOF, this Agreement has been duly executed and delivered by the duly authorized officers of the parties hereto as of the date first herein above written.

ADAPTHEALTH CORP.

By: /s/ Luke McGee

Name: Luke McGee Title: Authorized Signatory

[Signature Page to Investment Agreement]

IN WITNESS WHEREOF, this Agreement has been duly executed and delivered by the duly authorized officers of the parties hereto as of the date first herein above written.

ADAPTHEALTH CORP.

By:

Name: Title:

OEP AHCO INVESTMENT HOLDINGS, LLC

By: /s/ Brad Coppens Name: Brad Coppens Title: Authorized Officer

[Signature Page to Investment Agreement]

ONE EQUITY PARTNERS VII, L.P.

By:	OEP VII General Partner, L.P.
Its:	General Partner

- OEP VII GP, L.L.C. General Partner By: Its:

- By:/s/ Gregory BelinfantiName:Gregory BelinfantiIts:Authorized Signatory

[Signature Page to Investment Agreement]

<u>Annex I</u>

[FORM OF]

CERTIFICATE OF DESIGNATION, PREFERENCES AND RIGHTS

OF

SERIES A CONVERTIBLE PREFERRED STOCK

PAR VALUE \$0.0001

OF

ADAPTHEALTH CORP.

On [•], 2020, the Board of Directors of AdaptHealth Corp., a Delaware corporation (the "<u>Company</u>"), adopted the following resolution designating and creating, out of the authorized and unissued shares of preferred stock of the Company, [•] authorized shares of a series of preferred stock of the Company titled the "Series A Convertible Preferred Stock":

RESOLVED that, pursuant to the Certificate of Incorporation, the By-Laws and applicable law, a series of preferred stock of the Company titled the "Series A Convertible Preferred Stock," and having a par value of \$0.0001 per share and an initial number of authorized shares equal to [•], is hereby designated and created out of the authorized and unissued shares of preferred stock of the Company, which series has the rights, designations, preferences, voting powers and other provisions set forth below.

SECTION 1. <u>Classification and Number of Shares</u>. The shares of such series of Preferred Stock shall be classified as "Series A Convertible Preferred Stock" (the "<u>Series A Preferred Stock</u>"). The number of authorized shares constituting the Series A Preferred Stock shall be [•]. That number from time to time may be increased or decreased (but not below the number of shares of Series A Preferred Stock then outstanding) by (a) further resolution duly adopted by the Board and (b) the filing of a certificate of increase or decrease with the Secretary of State of the State of Delaware. The Company shall not have the authority to issue fractional shares of Series A Preferred Stock.

SECTION 2. <u>Ranking</u>. The Series A Preferred Stock will rank, with respect to rights on the distribution of assets on any voluntary or involuntary liquidation, dissolution or winding up of the affairs of the Company:

(a) on a parity basis with each other class or series of Capital Stock of the Company now existing or hereafter authorized, classified or reclassified, the terms of which expressly provide that such class or series ranks on a parity basis with the Series A Preferred Stock as to rights on the distribution of assets on any voluntary or involuntary liquidation, dissolution or winding up of the affairs of the Company (such Capital Stock, "<u>Parity Stock</u>");

(b) junior to each other class or series of Capital Stock of the Company now existing or hereafter authorized, classified or reclassified, the terms of which expressly provide that such class or series ranks senior to the Series A Preferred Stock as to rights on the distribution of assets on any voluntary or involuntary liquidation, dissolution or winding up of the affairs of the Company (such Capital Stock, "Senior Stock"); and

(c) senior to the Common Stock and each other class or series of Capital Stock of the Company now existing or hereafter authorized, classified or reclassified, other than Parity Stock and Senior Stock (such Capital Stock, "Junior Stock").

SECTION 3. <u>Definitions</u>. As used herein with respect to Series A Preferred Stock:

question.

"Acquisition Agreement" means that certain Stock Purchase Agreement and Agreement and Plan of Merger, dated as of May 25, 2020, by and among the Company, AdaptHealth LLC, Eleanor Merger Sub LLC, Solara Holdings, LLC and LCP Solara Blocker Seller, LLC, in its capacity as Blocker Seller and the Representative (in each case as defined therein), as may be amended from time to time.

"Affiliate" means, as to any Person, any other Person that, directly or indirectly, controls, or is controlled by, or is under common control with, such Person; provided, however, (a) that the Company and its Subsidiaries shall not be deemed to be Affiliates of the Investor or any of its Affiliates and (b) portfolio companies in which any Person or any of its Affiliates has an investment shall not be deemed an Affiliate of such Person. For this purpose, "control" (including, with its correlative meanings, "controlled by" and "under common control with") shall mean the possession, directly or indirectly, of the power to direct or cause the direction of management or policies of a Person, whether through the ownership of securities or partnership or other ownership interests, by contract or otherwise.

Any Person shall be deemed to "beneficially own," to have "beneficial ownership" of, or to be "beneficially owning" any securities (which securities shall also be deemed "beneficially owned" by such Person) that such Person together with such Person's Affiliates is deemed to "beneficially own" within the meaning of Rules 13d-3 and 13d-5 under the Exchange Act; provided, however, that any Person shall be deemed to beneficially own any securities that such Person has the right to acquire, whether or not such right is exercisable within sixty (60) days or thereafter (including assuming conversion of all Series A Preferred Stock, if any, owned by such Person into Class A Common Stock).

"Board" means the Board of Directors of the Company or any committee thereof duly authorized to act on behalf of such Board of Directors for the purposes in

"Business Day" means any weekday that is not a day on which banking institutions in New York, New York or the State of Pennsylvania are authorized or required by law, regulation or executive order to be closed.

-2-

"By-Laws" means the Amended and Restated By Laws of the Company, as may be amended from time to time.

"Capital Stock" means, with respect to any Person, any and all shares of, interests in, rights to purchase, warrants to purchase, options for, participations in or other equivalents of or interests in (however designated) stock issued by such Person.

"Cash Payment" has the meaning set forth in Section 8(f).

"Cash Payment Deadline" has the meaning set forth in Section 8(f).

"Certificate of Designations" means this Certificate of Designation, Preferences and Rights, as may be amended from time to time.

"Certificate of Incorporation" means the Second Amended and Restated Certificate of Incorporation of the Company, as modified by the Certificate of Correction to the Second Amended and Restated Certificate of Incorporation of the Company, and as may be amended from time to time.

"Change of Control" means the occurrence, directly or indirectly, of one of the following, whether in a single transaction or a series of transactions:

(a) any "person" or "group" (as such terms are used in Sections 13(d) and 14(d) of the Exchange Act), is or becomes the "beneficial owner" (as defined in Rules 13d-3 and 13d-5 under the Exchange Act), directly or indirectly, of a majority of the total voting power of the Voting Stock of the Company, other than as a result of any such transaction in which the holders of securities that represented 100% of the Voting Stock of the Company immediately prior to such transaction are substantially the same as the holders of securities that represent a majority of the total voting power of all classes of Voting Stock of the surviving Person or any parent entity that wholly owns such surviving Person immediately after such transaction; or

(b) the merger or consolidation of the Company with or into another Person or the merger of another Person with or into the Company, or the sale, lease or transfer of all or substantially all of the assets of the Company (determined on a consolidated basis) to another Person, or any recapitalization, reclassification or other transaction in which all or substantially all of the Class A Common Stock is exchanged for or converted into cash, securities or other property, other than (i) a transaction following which holders of securities that represented 100% of the Voting Stock of the Company immediately prior to such transaction own, directly or indirectly (in substantially the same proportion to each other as immediately prior to such transaction, other than changes in proportionality as a result of any cash/stock election provided under the terms of the definitive agreement regarding such transaction), at least a majority of the voting power of the Voting Stock of the surviving Person in such merger or consolidation transaction immediately after such transaction or (ii) a sale, lease or transfer to a Subsidiary or a Person that becomes a Subsidiary of the Company.

"Class A Common Stock" means the Common Stock of the Company designated as Class A common stock, \$0.0001 par value per share.

-3-

"Class B Common Stock" means the Common Stock of the Company designated as Class B common stock, \$0.0001 par value per share.

"close of business" means 5:00 p.m. (New York City time).

"<u>Closing Price</u>" of the Class A Common Stock on any date of determination means the closing sale price or, if no closing sale price is reported, the last reported sale price of the shares of the Class A Common Stock on the NASDAQ on such date. If the Class A Common Stock is not traded on the NASDAQ on any date of determination, the Closing Price of the Class A Common Stock on such date of determination means the closing sale price as reported in the composite transactions for the principal United States securities exchange or automated quotation system on which the Class A Common Stock is so listed or quoted, or, if no closing sale price is reported, the last reported sale price on the principal United States securities exchange or automated quotation system on which the Class A Common Stock is so listed or quoted, or, if no closing sale price is reported, the last reported sale price on the principal United States securities exchange or automated quotation system on which the Class A Common Stock is so listed or quoted, or if the Class A Common Stock is not so listed or quoted on a United States securities exchange or automated quotation system, the last quoted bid price for the Class A Common Stock in the over-the-counter market as reported by OTC Markets Group Inc. or any similar organization, or, if that bid price is not available, the market price of the Class A Common Stock on that date as mutually agreed between the Company and the Holders of a majority of the Series A Preferred Stock or, in the absence of such agreement, as determined by an Independent Financial Advisor retained by the Company for such purpose.

"Common Stock" means (i) the common stock, \$0.0001 par value per share, of the Company, consisting of Class A Common Stock and Class B Common Stock and (ii) any capital stock into which such common stock shall have been changed or any share capital resulting from a reclassification of such common stock.

"Company" has the meaning set forth in the recitals above.

"Constituent Person" has the meaning set forth in Section 11(a)(iii).

"Conversion Date" has the meaning set forth in Section 8.

"Conversion Notice" has the meaning set forth in Section 8(a)(i).

"Conversion Price" means, for each share of Series A Preferred Stock, a dollar amount equal to \$1,000 divided by the Conversion Rate.

"Conversion Rate" means, for each share of Series A Preferred Stock, 72.727273 shares of Class A Common Stock, subject to adjustment as set forth herein.

"<u>Current Market Price</u>" per share of Class A Common Stock, as of any date of determination, means the arithmetic average of the VWAP per share of Class A Common Stock for each of the ten (10) consecutive full Trading Days ending on, and including, the Trading Day immediately preceding such day, appropriately adjusted to take into account the occurrence during such period of any event described in Section 10.

"Default Cash Dividends" has the meaning set forth in Section 8(f).

-4-

"Distributed Property" has the meaning set forth in Section 10(a)(iii).

"Distribution Transaction" means any dividend or other distribution of equity securities of a Subsidiary of the Company to holders of Class A Common Stock in which such Person ceases to be a Subsidiary of the Company by reason of such dividend or distribution of equity securities, whether by means of a spin-off, split-off, redemption, reclassification, exchange, stock dividend, share distribution, rights offering or similar transaction.

"Exchange Act" means the Securities Exchange Act of 1934, as amended.

"Exchange Property" has the meaning set forth in Section 11(a).

"Exchange Property Unit" has the meaning set forth in Section 11(a).

"<u>Fair Market Value</u>" means, with respect to any security or other property, the fair market value of such security or other property as reasonably determined by a majority of the Board acting in good faith, or as mutually agreed between the Company and the Holders of a majority of the Series A Preferred Stock or, in the absence of such agreement, (a) after consultation with an Independent Financial Advisor, as to any security or other property with a Fair Market Value of less than \$25,000,000, or (b) otherwise using an Independent Financial Advisor to provide a valuation opinion.

"Guarantor" means One Equity Partners VII, L.P.

"Holder" means a Person in whose name any Series A Preferred Stock is registered in the Register.

"Independent Financial Advisor" means an accounting, appraisal, investment banking firm or consultant of nationally recognized standing; provided, however, that such firm or consultant shall not be an Affiliate of the Company and shall be reasonably acceptable to the Holders of at least a majority of the shares of Series A Preferred Stock outstanding at such time.

"Investment Agreement" means that certain Investment Agreement between the Company, the Investor and the Guarantor dated as of May 25, 2020, as it may be amended, supplemented or otherwise modified from time to time, with respect to certain terms and conditions concerning, among other things, the rights of and restrictions on the Holders.

"Investor" means OEP AHCO Investment Holdings, LLC.

"Issuance Date" means, with respect to any share of Series A Preferred Stock, the date of issuance of such share.

"Junior Stock" has the meaning set forth in Section 2(c).

"Liquidation Preference" means, with respect to any share of Series A Preferred Stock, as of any date, \$0.0001 per share.

-5-

"Mandatory Conversion" has the meaning set forth in Section 7.

"Mandatory Conversion Date" has the meaning set forth in Section 7.

"<u>Market Disruption Event</u>" means, with respect to any date, the occurrence or existence, during the one-half hour period ending at the scheduled close of trading on such date on the principal U.S. national or regional securities exchange or other market on which the Class A Common Stock is listed for trading or trades (or for purposes of determining the VWAP per share of Class A Common Stock, any period or periods aggregating one half-hour or longer during the regular trading session on the relevant day), of any material suspension or limitation imposed on trading (by reason of movements in price exceeding limits permitted by the relevant exchange or otherwise) in the Class A Common Stock or in any options contracts or futures contracts relating to the Class A Common Stock.

"<u>NASDAQ</u>" means the NASDAQ Stock Market (or its successor).

"Notice of Mandatory Conversion" has the meaning set forth in Section 7(b).

"Optional Conversion" has the meaning set forth in Section 6(a).

"Original Issuance Date" means the date of closing pursuant to the Investment Agreement.

"Original Issue Price" means, with respect to any share of Series A Preferred Stock, as of any date, \$1,000.00 per share.

"Parity Stock" has the meaning set forth in Section 2(a).

"Participating Dividends" has the meaning set forth in Section 4(b).

"Participating Dividend Record Date" has the meaning set forth in Section 4(d).

"Person" means any individual, corporation, estate, partnership, joint venture, association, joint-stock company, limited liability company, trust, unincorporated organization or any other entity.

"Preferred Stock" means the preferred stock, \$0.0001 par value per share, of the Company.

"Record Date" means, with respect to any dividend, distribution or other transaction or event in which the holders of the Class A Common Stock have the right to receive any cash, securities or other property or in which the Class A Common Stock is exchanged for or converted into any combination of cash, securities or other property, the date fixed for determination of holders of the Class A Common Stock entitled to receive such cash, securities or other property (whether such date is fixed by the Board or by statute, contract or otherwise).

"Register" means the securities register maintained in respect of the Series A Preferred Stock by the Company, or, to the extent the Company has engaged a transfer agent, such transfer agent.

-6-

"Reorganization Event" has the meaning set forth in Section 11(a)(iii).

"Requisite Stockholder Approval" means the stockholder approval contemplated by NASDAQ listing rule 5635 (or its successor) with respect to the issuance of shares of Common Stock upon conversion of the Series A Preferred Stock in excess of the limitations imposed by such rule.

"Senior Stock" has the meaning set forth in Section 2(b).

"Series A Preferred Stock" has the meaning set forth in Section 1.

"Share Cap" means zero (0) shares of Common Stock.

A "<u>Subsidiary</u>" of any Person means any corporation, limited liability company, partnership, association, trust or other entity of which securities or other ownership interests representing more than 50% of the ordinary voting power (or, in the case of a partnership, more than 50% of the general partnership interests) are owned by such Person or one or more Subsidiaries of such Person.

"Trading Day" means a day on which the NASDAQ is open for the transaction of business and on which there has not occurred a Market Disruption Event.

"Trigger Event" has the meaning set forth in Section 10(a)(v).

"Voting Stock" means (a) with respect to the Company, the Common Stock and any other Capital Stock of the Company having the right to vote generally in any election of directors of the Board and (b) with respect to any other Person, all Capital Stock of such Person having the right to vote generally in any election of directors of such Person or other similar governing body.

"<u>VWAP</u>" per share of Class A Common Stock on any Trading Day means the per share volume-weighted average price as displayed under the heading Bloomberg VWAP on Bloomberg (or, if Bloomberg ceases to publish such price, any successor service reasonably chosen by the Company) page "AHCO <equity>AQR" (or its equivalent successor if such page is not available) in respect of the period from the open of trading on the relevant Trading Day until the close of trading on such Trading Day (or if such volume-weighted average price is unavailable, the market price of one (1) share of Class A Common Stock on such Trading Day determined, using a volume-weighted average method, as mutually agreed between the Company and the Holders of a majority of the Series A Preferred Stock or, in the absence of such agreement, as determined by an Independent Financial Advisor retained by the Company for such purpose).

SECTION 4. Dividends. (a) Holders shall be entitled to receive dividends of the type and in the amount determined as set forth in this Section 4.

(b) <u>Participating Dividends</u>. Holders shall be entitled to participate equally and ratably with the holders of shares of Class A Common Stock in all cash dividends paid on the shares of Class A Common Stock as if immediately prior to each Participating Dividend Record Date, all shares of Series A Preferred Stock then outstanding were converted into shares of Class A Common Stock in accordance with <u>Section 6</u>. Dividends payable pursuant to this Section 4(b) (the "<u>Participating Dividends</u>") shall be payable on the same date that such dividends are payable to holders of shares of Class A Common Stock, unless the full dividends contemplated by this Section 4(b) are paid substantially at the same time to Holders.

-7-

(c) [Reserved]

(d) <u>Record Date for Participating Dividends</u>. Each Participating Dividend shall be paid pro rata to the Holders of shares of Preferred Stock entitled thereto. Each Participating Dividend shall be payable to the Holders of Preferred Stock as they appear on the Register at the close of business on the record date designated by the Board for such dividends (each such date, a "<u>Participating Dividend Record Date</u>") which shall be the same day as the record date for the payment of dividends to the holders of shares of Class A Common Stock.

(e) <u>Conversion Following a Participating Dividend Record Date</u>. If the Conversion Date for any shares of Series A Preferred Stock is prior to the close of business on a Participating Dividend Record Date, the Holder of such shares will not be entitled to any dividend in respect of such Participating Dividend Record Date. If the Conversion Date for any shares of Series A Preferred Stock is after the close of business on a Participating Dividend Record Date but prior to the corresponding payment date for such dividend, the Holder of such shares as of such Participating Dividend Record Date shall be entitled to receive such dividend, notwithstanding the conversion of such shares prior to the applicable dividend payment date.

SECTION 5. <u>Liquidation Rights</u>. (a) <u>Liquidation</u>. In the event of any voluntary or involuntary liquidation, dissolution or winding up of the affairs of the Company, the Holders shall be entitled, out of assets legally available therefor, before any distribution or payment out of the assets of the Company may be made to or set aside for the holders of any Junior Stock and subject to the rights of the holders of any Senior Stock or Parity Stock and the rights of the Company's existing and future creditors, to receive in full a liquidating distribution in cash and in the amount per share of Series A Preferred Stock equal to the greater of (i) the Liquidation Preference and (ii) the amount such Holders would have received had such Holders, immediately prior to such voluntary or involuntary liquidation, dissolution or winding up of the affairs of the Company, converted such shares of Series A Preferred Stock into Class A Common Stock pursuant to Section 6 (notwithstanding the Conversion Restriction). Holders shall not be entitled to any further payments in the event of any such voluntary or involuntary liquidation or winding up of the affairs of the Company's remaining assets.

(b) <u>Partial Payment</u>. If in connection with any distribution described in Section 5(a) above, the assets of the Company or proceeds therefrom are not sufficient to pay in full the aggregate liquidating distributions required to be paid pursuant to Section 5(a) above to all Holders and the liquidating distributions payable to all holders of any Parity Stock, the amounts distributed to the Holders and to the holders of all such Parity Stock shall be paid pro rata in accordance with the respective aggregate liquidating distributions to which they would otherwise be entitled if all amounts payable thereon were paid in full.

-8-

(c) <u>Merger. Consolidation and Sale of Assets Not Liquidation</u>. For purposes of this <u>Section 5</u>, the sale, conveyance, exchange or transfer (for cash, shares of stock, securities or other consideration) of all or substantially all of the property and assets of the Company shall not be deemed a voluntary or involuntary liquidation, dissolution or winding up of the affairs of the Company, nor shall the merger, consolidation, statutory exchange or any other business combination transaction of the Company into or with any other Person or the merger, consolidation, statutory exchange or any other business combination for or with the Company be deemed to be a voluntary or involuntary liquidation, dissolution or winding up of the affairs of the Company.

SECTION 6. Right of the Holders to Convert.

(a) Subject to the terms of Section 8(f), each Holder shall have the right, at such Holder's option, subject to the conversion procedures set forth in Section 8, to convert (an "Optional Conversion") each share of such Holder's Series A Preferred Stock at any time into (i) the number of shares of Class A Common Stock equal to the quotient of (A) the Original Issue Price divided by (B) the Conversion Price as of the applicable Conversion Date <u>plus</u> (ii) cash in lieu of fractional shares as set out in <u>Section 8(e)</u>. The right of Optional Conversion may be exercised as to all or any portion of such Holder's Series A Preferred Stock from time to time; provided, however, that, in each case, no right of Optional Conversion may be exercised by a Holder in respect of fewer than 10,000 shares of Series A Preferred Stock (unless such conversion relates to all shares of Series A Preferred Stock held by such Holder or all shares of Series A Preferred Stock that may be converted in compliance with <u>Section 8(f)</u>.

(b) The Company shall at all times reserve and keep available out of its authorized and unissued Common Stock and Class A Common Stock, solely for issuance upon the conversion of the Series A Preferred Stock, such number of shares of Class A Common Stock as shall from time to time be issuable upon the conversion of all the shares of Series A Preferred Stock pursuant to <u>Section 6</u>. Any shares of Class A Common Stock issued upon conversion of Series A Preferred Stock shall be duly authorized, validly issued, fully paid and nonassessable and will not be subject to preemptive rights or subscription rights of any other stockholder of the Company.

SECTION 7. <u>Mandatory Conversion by the Company</u>. (a) Following the day upon which the Requisite Stockholder Approval is obtained, the Company may elect to convert up to 100% of the outstanding shares of Series A Preferred Stock (to the extent not previously redeemed by the Company) into shares of Class A Common Stock (the election to convert shares of Series A Preferred Stock pursuant to this <u>Section 7</u>, a "<u>Mandatory Conversion</u>", and the date selected by the Company for any Mandatory Conversion pursuant to this Section 7, the "<u>Mandatory Conversion Date</u>"). In the case of a Mandatory Conversion, each share of Series A Preferred Stock then outstanding that is to be converted pursuant to such Mandatory Conversion shall be converted into (A) the number of shares of Class A Common Stock equal to the quotient of (1) the Original Issue Price with respect to such share of Series A Preferred Stock as of the Mandatory Conversion Date <u>divided by</u> (2) the Conversion Price of such share in effect as of the Mandatory Conversion Date <u>plus</u> (B) cash in lieu of fractional shares as set out in <u>Section 8(e)</u>.

-9-

- (b) <u>Notice of Mandatory Conversion</u>. If the Company elects to effect a Mandatory Conversion, the Company shall provide notice of a Mandatory Conversion to each Holder (such notice, a "<u>Notice of Mandatory Conversion</u>"). The Mandatory Conversion Date selected by the Company shall be no less than five (5) Business Days and no more than twenty (20) Business Days after the date on which the Company provides the Notice of Mandatory Conversion to the Holders. The Notice of Mandatory Conversion shall state, as appropriate:
 - (i) the number of shares of Series A Preferred Stock held by such Holder that are subject to the Mandatory Conversion;
 - (ii) the Mandatory Conversion Date selected by the Company; and

(iii) the Conversion Rate as in effect on the Mandatory Conversion Date, the number of shares of Class A Common Stock to be issued to such Holder upon conversion of each share of Series A Preferred Stock held by such Holder and, if applicable, the cash in lieu of fractional shares to be paid thereon.

SECTION 8. <u>Conversion Procedures and Effect of Conversion</u>. (a) <u>Conversion Procedure</u>. A Holder must do each of the following in order to receive shares of Class A Common Stock upon conversion of shares of Series A Preferred Stock pursuant to this <u>Section</u> 8:

(i) in the case of an Optional Conversion, complete and manually sign the conversion notice in the form attached hereto as Exhibit I (the "Conversion Notice"), and deliver such notice to the Company; provided, however, that a Conversion Notice may be conditional on the completion of a Change of Control or other corporate transaction as such Holder may specify;

- (ii) deliver to the Company the certificate or certificates (if any) representing the shares of Series A Preferred Stock to be converted;
- (iii) if required, furnish appropriate endorsements and transfer documents; and
- (iv) if required, pay any stock transfer, documentary, stamp or similar taxes not payable by the Company pursuant to Section 17.

The foregoing clauses (ii), (iii) and (iv) shall be conditions to the issuance of shares of Class A Common Stock to the Holders in the event of a Mandatory Conversion pursuant to <u>Section</u> 7 (but, for the avoidance of doubt, not the Mandatory Conversion of the shares of Series A Preferred Stock on the Mandatory Conversion Date).

The "<u>Conversion Date</u>" means (A) with respect to an Optional Conversion pursuant to Section 6(a), the date on which such Holder complies with the procedures in this <u>Section 8</u> (including the satisfaction of any conditions to conversion set forth in the Conversion Notice) and (B) with respect to Mandatory Conversion pursuant to Section 7, the Mandatory Conversion Date.

(b) Effect of Conversion. Effective immediately prior to the close of business on the Conversion Date applicable to any shares of Series A Preferred Stock, Participating Dividends shall no longer accrue or be declared on any such shares of Series A Preferred Stock, and on conversion, such shares of Series A Preferred Stock shall cease to be outstanding.

(c) Record Holder of Underlying Securities as of Conversion Date. The Person or Persons entitled to receive the Class A Common Stock and, to the extent applicable, cash on a Conversion Date shall be treated for all purposes as the record holder(s) of such shares of Class A Common Stock and, to the extent applicable, cash as of the close of business on such Conversion Date. As promptly as practicable on or after the Conversion Date and compliance by the applicable Holder with the relevant procedures contained in Section 8 (and in any event no later than three (3) Trading Days thereafter), the Company shall issue the number of whole shares of Class A Common Stock shall be made, at the option of the Company, in certificated form or by book-entry. Any such certificates shall be delivered by the Company to the appropriate Holder on a book-entry basis or by mailing certificates evidencing the shares to the Holders at their respective addresses as set forth in the Conversion Notice (in the case of an Optional Conversion) or in the records of the Company (in the case of a Mandatory Conversion). In the event that a Holder shall not by written notice designate the name in which shares of Class A Common Stock (and payments of cash in lieu of fractional shares) to be delivered upon conversion of shares of Series A Preferred Stock should be registered or paid, or the manner in which shares and cash should be delivered, the Company shall be entitled to register and deliver such shares, and make such payment, in the name of the Holder and in the manner shown on the records of the Company.

(d) <u>No Adjustment</u>. No adjustment to shares of Series A Preferred Stock being converted on a Conversion Date or to the shares of Class A Common Stock deliverable to the Holders upon the conversion thereof shall be made in respect of dividends or other distributions payable to holders of the Class A Common Stock as of any date prior to the close of business on such Conversion Date (it being understood that the foregoing shall not limit any Holder's right to receive Participating Dividends payable prior to such time or the operation of Section 10(a) in respect of events occurring prior to such time). Until the Conversion Date with respect to any share of Series A Preferred Stock has occurred, such share of Series A Preferred Stock will remain outstanding and will be entitled to all of the powers, designations, preferences and other rights provided herein.

(e) <u>Fractional Shares</u>. No fractional shares of Class A Common Stock will be delivered to the Holders upon conversion. In lieu of fractional shares otherwise issuable, the Holders will be entitled to receive, at the Company's sole discretion, either (i) an amount in cash equal to the fraction of a share of Class A Common Stock <u>multiplied</u> by the Closing Price of the Class A Common Stock on the Trading Day immediately preceding the applicable Conversion Date or (ii) one (1) additional whole share of Class A Common Stock. To determine whether the number of shares of Class A Common Stock to be delivered to a Holder upon the conversion of such Holder's shares of Series A Preferred Stock will include a fractional share, such determination shall be based on the aggregate number of shares of Series A Preferred Stock of such Holder that are being converted on any single Conversion Date.

-11-

(f) <u>Restriction on Conversions</u>.

(i) Limitation on Conversion. Notwithstanding anything in this Certificate of Designations to the contrary, unless and until the Requisite Stockholder Approval (to the extent and only to the extent required under the listing rules of NASDAQ) is obtained, the Holders shall not have the right to acquire shares of Common Stock issuable upon conversion of the Series A Preferred Stock, and the Company shall not be required to issue shares of Common Stock issuable upon conversion of the Series A Preferred Stock, in excess of the Share Cap (the "Conversion Restriction"). Any purported delivery of shares of Common Stock upon conversion of any Series A Preferred Stock will be void and have no effect to the extent, and only to the extent, that such delivery would result in issuance of shares of Common Stock in excess of the Share Cap in violation of the listing rules of Nasdaq. Notwithstanding the foregoing, for any conversion following the six-month anniversary of the Original Issuance Date, in the case of a conversion pursuant to Section 6, Section 7 or this Section 8, the Holder or the Company, as applicable, may request a conversion of a number of shares of Series A Preferred Stock that would result in the issuance of shares that, but for the Conversation Restriction, would exceed the Share Cap; provided, that, in lieu of any shares of Class A Common Stock otherwise deliverable upon conversion that would, but for the Conversion Restriction, exceed the Share Cap, the Company shall instead deliver to the requisite Holder an amount of cash per share of such Class A Common Stock equal to the VWAP per share of Class A Common Stock on the Trading Day immediately preceding the Conversion Date (such payment, a "Cash Payment"), payable within not more than thirty (30) days following the applicable Conversion Date (such date, the "Cash Payment Deadline"); provided, however, that the Holder may not request the conversion of shares in excess of the Share Cap prior to the earlier to occur of (A) the Restricted Period Termination Date (as such term is defined in the Investment Agreement) and (B) if any, a Restricted Period Early Termination Event (as such term is defined in the Investment Agreement) following which the transfer restrictions in Section 5.3(a) of the Investment Agreement are not applicable to 100% of the Prohibited Shares (as such term is defined in the Investment Agreement) (following which the Holder may request conversion of shares that would, but for the Conversion Restriction, exceed the Share Cap which shall be treated in accordance with the forgoing proviso); provided, further, that, for any conversion that would, but for the Conversation Restriction, exceed the Share Cap that is in connection with a Change of Control, in lieu of the purchase price per share payable in accordance with the foregoing, the Company shall instead deliver to the requisite Holder, at the same time as such consideration is delivered to the holders of Class A Common Stock, the consideration received by the holders of Class A Common Stock in respect thereof, and if such payment is in violation of the Nasdaq listing rules applicable to the Company, the Company shall instead deliver to the Holder converting shares in excess of the Share Cap that would be in violation of the Nasdaq listing rules cash in an amount equal to the Fair Market Value of such consideration.

(ii) <u>Covenant to Seek the Requisite Stockholder Approval</u>. The Company will use its reasonable best efforts to obtain the Requisite Stockholder Approval, including by seeking such approval, if not previously obtained, at each regular annual meeting of its stockholders occurring after the first Company Stockholders' Meeting (as such term is defined in the Investment Agreement) and endorsing its approval in the related proxy materials. The Company will promptly notify the Holders if the Requisite Stockholder Approval is obtained.

-12-

(iii) <u>Cash Dividends Upon Default of Cash Payment Obligations</u>. If the Company fails to make any required Cash Payment by the required Cash Payment Deadline on any share of Series A Preferred Stock, then the Holder thereof will be entitled to receive cumulative cash dividends on each such share at a rate per annum of 4.00% on the Original Issuance Price (such dividends, "<u>Default Cash Dividends</u>"). Default Cash Dividends, if any, shall accumulate on a daily basis from, and including, the Cash Payment Deadline to, but excluding, the date upon which the required Cash Payment is made (whether or not there shall be earnings or funds of the Company legally available for the payment of Default Cash Dividends or the Company declares the payment of Default Cash Dividends). Default Cash Dividends, if any, shall be payable quarterly in arrears on March 31, June 30, September 31 and December 31 of each year to the applicable Holder as it appears on the Company's Register at the close of business on the March 15, June 15, September 15 and December 15 preceding the applicable payment date. Default Cash Dividends, if any, payable for any period less than a full quarterly dividend period (based upon the number of days elapsed during the period) shall be computed on the basis of a 360-day year consisting of twelve 30-day months.

SECTION 9. Change of Control.

(a) <u>Change of Control Notice.</u> On or before the twentieth (20th) Business Day prior to the date on which the Company anticipates consummating a Change of Control (or, if later, promptly after the Company discovers that a Change of Control may occur), a written notice shall be sent by or on behalf of the Company to the Holders as they appear in the records of the Company, which notice shall set forth a description of the anticipated Change of Control and contain the date on which the Change of Control is anticipated to be effected (or, if applicable, the date on which a Schedule TO or other schedule, form or report disclosing a Change of Control was filed).

(b) <u>Conversion Right</u>. Prior to the consummation of any Change of Control, each Holder shall be entitled, subject to Section 8(f), to exercise an Optional Conversion in respect of any and all of its Series A Preferred Stock prior to or conditioned upon such Change of Control.

SECTION 10. <u>Anti-Dilution Adjustments</u>. (a) <u>Adjustments</u>. The Conversion Rate will be subject to adjustment, without duplication, upon the occurrence of the following events, except that the Company shall not make any adjustment to the Conversion Rate if Holders of the Series A Preferred Stock participate, at the same time and upon the same terms as holders of Class A Common Stock and solely as a result of holding shares of Series A Preferred Stock, in any transaction described in this Section 10, without having to convert their Series A Preferred Stock, as if they held a number of shares of Class A Common Stock equal to the Conversion Rate <u>multiplied by</u> the number of shares of Series A Preferred Stock held by such Holders:

(i) The issuance of Class A Common Stock as a dividend or distribution to all or substantially all holders of Class A Common Stock, or a subdivision or combination (including, without limitation, a stock split or a reverse stock split) of Class A Common Stock or a reclassification of Class A Common Stock into a greater or lesser number of shares of Class A Common Stock, in which event the Conversion Rate shall be adjusted based on the following formula:

 $CR_1 = CR_0 x (OS_1 / OS_0)$

where,

-13-

 CR_0 = the Conversion Rate in effect immediately prior to (i) the close of business on (i) the Record Date for such dividend or distribution or (ii) the effective date of such subdivision, combination or reclassification;

 CR_1 = the new Conversion Rate in effect immediately after (i) the close of business on (i) the Record Date for such dividend or distribution or (ii) the effective date of such subdivision, combination or reclassification;

 OS_0 = the number of shares of Class A Common Stock outstanding immediately prior to the close of business on (i) the Record Date for such dividend or distribution or (ii) the effective date of such subdivision, combination or reclassification; and

 OS_1 = the number of shares of Class A Common Stock that would be outstanding immediately after, and solely as a result of, the completion of such dividend, distribution, subdivision, combination or reclassification.

Any adjustment made pursuant to this clause (i) shall be effective immediately after the close of business on (i) the Record Date for such dividend or distribution or (ii) the effective date of such subdivision, combination or reclassification. If any such dividend, distribution, subdivision, combination or reclassification is announced or declared but does not occur, the Conversion Rate shall be readjusted, effective as of the date the Board announces that such dividend, distribution, subdivision, combination or reclassification is effect if such dividend, distribution, subdivision, combination or reclassification had not been declared.

(ii) The dividend, distribution or other issuance to all or substantially all holders of Class A Common Stock of rights (other than rights, options or warrants distributed in connection with a stockholder rights plan (in which event the provisions of Section 10(a)(v) shall apply)), options or warrants entitling them to subscribe for or purchase shares of Class A Common Stock for a period expiring forty-five (45) days or less from the date of issuance thereof, at a price per share that is less than the Current Market Price as of the Record Date for such issuance, in which event the Conversion Rate will be increased based on the following formula:

 $CR_1 = CR_0 x [(OS_0 + X)] / (OS_0 + Y)$

where,

CR0 = the Conversion Rate in effect immediately prior to the close of business on the Record Date for such dividend, distribution or other issuance;

CR1 = the new Conversion Rate in effect immediately after the close of business on the Record Date for such dividend, distribution or other issuance;

 OS_0 = the number of shares of Class A Common Stock outstanding immediately prior to the close of business on the Record Date for such dividend, distribution or other issuance;

-14-

X = the total number of shares of Class A Common Stock issuable pursuant to such rights, options or warrants; and

Y = the number of shares of Class A Common Stock equal to the aggregate price payable to exercise such rights, options or warrants divided by the Current Market Price as of the Record Date for such dividend, distribution or other issuance.

For purposes of this clause (ii), in determining whether any rights, options or warrants entitle the holders to purchase the Class A Common Stock at a price per share that is less than the Current Market Price as of the Record Date for such dividend, distribution or issuance, there shall be taken into account any consideration the Company receives for such rights, options or warrants, and any amount payable on exercise thereof, with the value of such consideration, if other than cash, to be the Fair Market Value thereof.

Any adjustment made pursuant to this clause (ii) shall become effective immediately following the close of business on the Record Date for such dividend, distribution or other issuance. In the event that such rights, options or warrants are not so issued, the Conversion Rate shall be readjusted, effective as of the date the Board publicly announced its decision not to issue such rights, options or warrants to the Conversion Rate that would then be in effect if such dividend, distribution or issuance had not been declared. To the extent that such rights, options or warrants are not exercised prior to their expiration or shares of Class A Common Stock are otherwise not delivered pursuant to such rights, options or warrants upon the exercise of such rights, options or warrants, the Conversion Rate shall be readjusted to the Conversion Rate that would then be in effect had the adjustments made upon the dividend, distribution or other issuance of such rights, options or warrants been made on the basis of the delivery of only the number of shares of Class A Common Stock actually delivered.

(iii) The Company shall, by dividend or otherwise, distribute to all or substantially all holders of its Class A Common Stock (other than for cash in lieu of fractional shares), shares of any class of its Capital Stock, evidences of its indebtedness, assets, other property or securities, but excluding (A) dividends or distributions referred to in Section 10(a)(i) or Section 10(a)(i) hereof; (B) Distribution Transactions as to which Section 10(a)(iv) shall apply; (C) dividends or distributions paid exclusively in cash; and (D) rights, options or warrants distributed in connection with a stockholder rights plan as to which Section 10(a)(v) shall apply (any of such shares of its Capital Stock, indebtedness, assets or property that are not so excluded are hereinafter called the "Distributed Property"), then, in each such case the Conversion Rate shall be adjusted based on the following formula:

 $CR_1 = CR_0 x [SP_0 / (SP_0 - FMV)]$

where,

CR0 = the Conversion Rate in effect immediately prior to the close of business on the Record Date for such dividend or distribution;

 CR_1 = the new Conversion Rate in effect immediately after the close of business on the Record Date for such dividend or distribution;

SP0 = the Current Market Price as of the Record Date for such dividend or distribution; and

FMV= the Fair Market Value of the portion of Distributed Property distributed with respect to each outstanding share of Class A Common Stock on the Record Date for such dividend or distribution; provided, however, that, if FMV is equal or greater than SP₀, then in lieu of the foregoing adjustment, the Company shall distribute to each holder of Series A Preferred Stock on the date the applicable Distributed Property is distributed to holders of Class A Common Stock, but without requiring such holder to convert its shares of Series A Preferred Stock, in respect of each share of Series A Preferred Stock held by such holder, the amount of Distributed Property such holder would have received had such holder owned a number of shares of Class A Common Stock equal to the Conversion Rate on the Record Date for such dividend or distribution

Any adjustment made pursuant to this clause (iii) shall be effective immediately after the close of business on the Record Date for such dividend or distribution. If any such dividend or distribution is declared but does not occur, the Conversion Rate shall be readjusted, effective as of the date the Board announces that such dividend or distribution shall not occur to the Conversion Rate that would then be in effect if such dividend or distribution had not been declared.

(iv) The Company effects a Distribution Transaction, in which case the Conversion Rate in effect immediately prior to the effective date of the Distribution Transaction shall be increased based on the following formula:

 $CR_1 = CR_0 x [(FMV + MP_0) / MP_0]$

where,

CR0 = the Conversion Rate in effect immediately prior to the close of business on the effective date of the Distribution Transaction;

CR1 = the new Conversion Rate in effect immediately after the close of business on the effective date of the Distribution Transaction;

FMV= the arithmetic average of the volume-weighted average prices for a share of the capital stock or other interest distributed to holders of Class A Common Stock on the principal United States securities exchange or automated quotation system on which such capital stock or other interest trades, as reported by Bloomberg (or, if Bloomberg ceases to publish such price, any successor service chosen by the Company) in respect of the period from the open of trading on the relevant Trading Day until the close of trading on such Trading Day (or if such volume-weighted average price is unavailable, the market price of one (1) share of such capital stock or other interest on such Trading Day determined, using a volume-weighted average method, as mutually agreed between the Company and the Holders of a majority of the Series A Preferred Stock or, in the absence of such agreement, as determined by an Independent Financial Advisor retained for such purpose by the Company), for each of the ten (10) consecutive full Trading Days commencing with, and including, the effective date of the Distribution Transaction; and

-16-

 MP_0 = the arithmetic average of the VWAP per share of Class A Common Stock for each of the ten (10) consecutive full Trading Days commencing on, and including, the effective date of the Distribution Transaction

Such adjustment shall become effective immediately following the close of business on the effective date of the Distribution Transaction. If an adjustment to the Conversion Rate is required under this Section 10(a)(iv), delivery of any additional shares of Class A Common Stock that may be deliverable upon conversion as a result of an adjustment required under this Section 10(a)(iv) shall be delayed only to the extent necessary in order to complete the calculations provided for in this Section 10(a)(iv).

(v) If the Company has a stockholder rights plan in effect with respect to the Class A Common Stock on any Conversion Date, upon conversion of any shares of the Series A Preferred Stock, Holders of such shares will receive, in addition to the applicable number of shares of Class A Common Stock, the rights under such rights plan relating to such Class A Common Stock, unless, prior to such Conversion Date, the rights have (i) become exercisable or (ii) separated from the shares of Class A Common Stock (the first of such events to occur, a "Trigger Event"), in which case, the Conversion Rate will be adjusted, effective automatically at the time of such Trigger Event, as if the Company had made a distribution of such rights to all holders of the Class A Common Stock as described in Section 10(a)(ii) (without giving effect to the forty-five (45)-day limit on the exercisability of rights, options or warrants ordinarily subject to such Section 10(a)(ii)), subject to appropriate readjustment in the event of the expiration, termination or redemption of such rights prior to the exercise, deemed exercise or exchange thereof. Notwithstanding the foregoing, to the extent any such stockholder rights had not been issued, but the Company had instead issued such shares of Class A Common Stock or other property or securities, as a dividend or distribution of shares of Class A Common Stock or other property or securities as a dividend or distribution of shares of Class A Common Stock or other property or securities as a dividend or distribution of shares of Class A Common Stock or other property or securities as a dividend or distribution of shares of Class A Common Stock or other property or securities as a dividend or distribution of shares of Class A Common Stock or other property or securities as a dividend or distribution of shares of Class A Common Stock pursuant to Section 10(a)(i) or Section 10(a)(iii), as applicable.

To the extent that such rights are not exercised prior to their expiration, termination or redemption, the Conversion Rate shall be readjusted to the Conversion Rate that would then be in effect had the adjustments made upon the occurrence of the Trigger Event been made on the basis of the issuance of, and the receipt of the exercise price with respect to, only the number of shares of Class A Common Stock actually issued pursuant to such rights.

Notwithstanding anything to the contrary in this Section 10(a)(v), no adjustment shall be required to be made to the Conversion Rate with respect to any Holder which is, or is an "affiliate" or "associate" of, an "acquiring person" under such stockholder rights plan or with respect to any direct or indirect transferee of such Holder who receives Series A Preferred Stock in such transfer after the time such Holder becomes, or its affiliate or associate becomes, such an "acquiring person."

-17-

(b) <u>Calculation of Adjustments</u>. All adjustments to the Conversion Rate shall be calculated by the Company to the nearest 1/10,000th of one (1) share of Class A Common Stock (or if there is not a nearest 1/10,000th of a share, to the next lower 1/10,000th of a share). No adjustment to the Conversion Rate will be required, unless such adjustment would require an increase or decrease of at least one percent (1%) of the Conversion Rate; <u>provided</u>, <u>however</u>, that any such adjustment that is not required to be made will be carried forward and taken into account in any subsequent adjustment; <u>provided</u>, <u>further</u>, that any such adjustment of less than one percent (1%) that has not been made will be made upon any Conversion Date.

(c) When No Adjustment Required. (i) Except as otherwise provided in this Section 10, the Conversion Rate will not be adjusted for the issuance of Class A Common Stock or any securities convertible into or exchangeable for Class A Common Stock or carrying the right to purchase any of the foregoing, or for the repurchase of Class A Common Stock.

(ii) Except as otherwise provided in this Section 10, the Conversion Rate will not be adjusted as a result of the issuance of, the distribution of separate certificates representing, the exercise or redemption of, or the termination or invalidation of, rights pursuant to any stockholder rights plans.

(iii) No adjustment to the Conversion Rate will be made:

(A) upon the issuance of any shares of Class A Common Stock pursuant to any present or future plan providing for the reinvestment of dividends or interest payable on securities of the Company and the investment of additional optional amounts in Class A Common Stock under any plan in which purchases are made at market prices on the date or dates of purchase, without discount, and whether or not the Company bears the ordinary costs of administration and operation of the plan, including brokerage commissions;

(B) upon the issuance of any shares of Class A Common Stock or options or rights to purchase such shares pursuant to any present or future employee, director or consultant benefit plan or program of or assumed by the Company or any of its Subsidiaries or of any employee agreements or arrangements or programs;

(C) upon the issuance of any shares of Class A Common Stock pursuant to any option, warrant, right, or exercisable, exchangeable or convertible

security;

(D) for dividends or distributions declared or paid to holders of Class A Common Stock in which Holders participate pursuant to Section 4(b); or

(E) for a change solely in the par value of the Class A Common Stock.

(d) <u>Successive Adjustments</u>. After an adjustment to the Conversion Rate under this Section 10, any subsequent event requiring an adjustment under this Section 10 shall cause an adjustment to each such Conversion Rate as so adjusted.

-18-

(e) <u>Multiple Adjustments</u>. For the avoidance of doubt, if an event occurs that would trigger an adjustment to the Conversion Rate pursuant to this Section 10 under more than one subsection hereof, such event, to the extent fully taken into account in a single adjustment, shall not result in multiple adjustments hereunder; provided, however, that if more than one subsection of this Section 10 is applicable to a single event, the subsection shall be applied that produces the largest adjustment.

(f) <u>Tax Adjustments</u>. The Company may, but shall not be required to, make such increases in the Conversion Rate, in addition to those required by this Section 10, as the Board considers to be advisable in order to avoid or diminish any income tax to any holders of shares of Company stock resulting from any dividend or distribution of stock or issuance of rights or warrants to purchase or subscribe for stock or from any event treated as such for income tax purposes or for any other reason.

(g) <u>Notice of Adjustments</u>. Whenever the Conversion Rate is adjusted as provided under this Section 10, the Company shall as soon as reasonably practicable following the occurrence of an event that requires such adjustment (or if the Company is not aware of such occurrence, as soon as reasonably practicable after becoming so aware) or the date the Company makes an adjustment pursuant to Section 10(f):

(i) compute the adjusted applicable Conversion Rate in accordance with this Section 10; and

(ii) (A) in the event that the Company shall give notice or make a public announcement to the holders of Class A Common Stock of any action of the type described in Section 10 (but only if the action of the type described in Section 10 would result in an adjustment to the Conversion Price or a change in the type of securities or property to be delivered upon conversion of the Series A Preferred Stock), the Company shall, at the time of such notice or announcement, and in the case of any action that would require the fixing of a record date, at least ten (10) days prior to such record date, give notice to each Holder by mail, first-class postage prepaid, at the address appearing in the Register, which notice shall specify the record date, if any, with respect to any such action, the approximate date on which such action is to take place and the facts with respect to such action as shall be reasonably necessary to indicate the effect on the Conversion Price and the number, kind or class of shares or other securities or property, which shall be deliverable upon conversion or redemption of the Series A Preferred Stock or (B) in the event that the Company does not give notice or make a public announcement as set forth in subclause (A) of this clause (ii), the Company is not aware of such occurrence, as soon as practicable after becoming so aware), provide, or cause to be provided, a written notice to the Holders of the occurrence of such event, in the same manner and with the same detail as the notice set forth in subclause (A) of this clause (ii); and

(iii) whenever the Conversion Price shall be adjusted pursuant to one or more provisions of Section 10, the Company shall, as soon as practicable following the determination of the revised Conversion Price, (A) file at the principal office of the Company, a statement showing in reasonable detail the facts requiring such adjustment, the Conversion Price that shall be in effect after such adjustment and the method by which the adjustment to the Conversion Price was determined and (B) cause a copy of such statement to be sent in the manner set forth in subclause (A) of clause (ii) to each Holder.

-19-

SECTION 11. Adjustment for Reorganization Events.

(a) <u>Reorganization Events</u>. In the event of:

(i) any reclassification, statutory exchange, merger, consolidation or other similar business combination of the Company with or into another Person, in each case, pursuant to which at least a majority of the Class A Common Stock is changed or converted into, or exchanged for, cash, securities or other property of the Company or another Person;

(ii) any sale, transfer, lease or conveyance to another Person of all or substantially all the property and assets of the Company, in each case pursuant to which the Class A Common Stock is converted into cash, securities or other property; or

(iii) any statutory exchange of securities of the Company with another Person (other than in connection with a merger or acquisition) or reclassification, recapitalization or reorganization of the Class A Common Stock into other securities;

(each of which is referred to as a "Reorganization Event" and the cash, securities or other property into which the Class A Common Stock is changed, converted or exchanged, the "Exchange Property" and the amount and kind of Exchange Property that a holder of one (1) share of Class A Common Stock would be entitled to receive on account of such Reorganization Event (without giving effect to any arrangement not to issue or deliver a fractional portion of any security or other property), an "Exchange Property Unit"), then, notwithstanding anything to the contrary in this Certificate of Designations, from and after the effective time of such Reorganization Event, without the consent of the Holders, each share of Series A Preferred Stock will remain outstanding (unless converted in accordance with Section 11(d)) and (I) the consideration due upon conversion of any Series A Preferred Stock will be determined in the same manner as if each reference to any number of shares of Class A Common Stock in Section 10 or in this Section 11, or in any related definitions, were instead a reference to the same number of Exchange Property Units; (II) for purposes of Sections 6 and 7, each reference to any number of shares of Class A Common Stock in such Sections (or in any related definitions) will instead be deemed to be a reference to the same number of Exchange Property Units (and the terms of any conversion shall be based upon the Original Issue Price at the time of such subsequent conversion); and (III) other references to "Class A Common Stock" shall refer to the Exchange Property with appropriate adjustment to preserve, to the greatest extent possible (so long as there is no detrimental effect to the Company), the economic and other rights in respect of the Series A Preferred Stock granted by this Certificate of Designations and the Investment Agreement; provided, however, that the foregoing shall not apply if such Holder is a Person with which the Company consolidated or into which the Company merged or which merged into the Company or to which such sale or transfer was made, as the case may be (any such Person, a "Constituent Person"), or an Affiliate of a Constituent Person, to the extent such Reorganization Event provides for different treatment of Class A Common Stock held by such Persons. If the kind or amount of securities, cash and other property receivable upon such Reorganization Event is not the same for each share of Class A Common Stock held immediately prior to such Reorganization Event by a Person (other than a Constituent Person or an Affiliate thereof), then for the purpose of this Section 11(a), the kind and amount of securities, cash and other property receivable upon conversion following such Reorganization Event will be deemed to be the weighted average of the types and amounts of consideration received by the holders of Class A Common Stock.

-20-

(b) <u>Successive Reorganization Events</u>. The above provisions of this Section 11 shall similarly apply to successive Reorganization Events.

(c) <u>Reorganization Event Notice</u>. The Company (or any successor) shall, no less than twenty (20) Business Days prior to the anticipated effective date of any Reorganization Event, provide written notice to the Holders of such occurrence of such event and of the kind and amount of the cash, securities or other property that constitutes the Exchange Property. Failure to deliver such notice shall not affect the operation of this <u>Section 11</u>.

(d) <u>Reorganization Event Agreements</u>. The Company shall not enter into any agreement for a transaction constituting a Reorganization Event, unless (i) such agreement provides for or does not interfere with or prevent (as applicable) conversion of the Series A Preferred Stock into the Exchange Property in a manner that is consistent with and gives effect to this <u>Section 11</u>, and (ii) to the extent that the Company is not the surviving corporation in such Reorganization Event or will be dissolved in connection with such Reorganization Event, proper provision shall be made in the agreements governing such Reorganization Event for the conversion of the Series A Preferred Stock into stock of the Person surviving such Reorganization Event or such other continuing entity in such Reorganization Event.

(e) <u>Change of Control</u>. For the sake of clarity, if a Reorganization Event constitutes a Change of Control, then <u>Section 9</u> shall take precedence over this <u>Section 11</u> to the extent there is any inconsistency between such sections.

SECTION 12. Adverse Changes; Voting Rights.

(a) So long as any shares of Series A Preferred Stock are outstanding, in addition to any other vote required by applicable law, the Company may not take any of the following actions (including by means of merger, consolidation, reorganization, recapitalization or otherwise) without the prior affirmative vote or written consent from the Holders of at least a majority of the then-issued and outstanding shares of Series A Preferred Stock, voting as a separate class: amend, alter, repeal or otherwise modify (whether by merger, consolidation or otherwise) any provision of the Certificate of Incorporation (including this Certificate of Designations) in a manner that would adversely affect the powers, preferences, rights or privileges of the Series A Preferred Stock.

(b) Each Holder of Series A Preferred Stock will have one (1) vote per share on any matter on which Holders of Series A Preferred Stock are entitled to vote separately as a class, whether at a meeting or by written consent.

(c) For the avoidance of doubt and notwithstanding anything to the contrary in the Certificate of Incorporation or By-Laws, the Holders shall have the exclusive consent and voting rights set forth in <u>Section 12(a)</u> and may take action or consent to any action with respect to such rights without a meeting by delivering a consent in writing or by electronic transmission of the Holders of the Series A Preferred Stock entitled to cast not less than the minimum number of votes that would be necessary to authorize, take or consent to such action at a meeting of stockholders.

-21-

(d) Except as otherwise provided herein or as otherwise required by Delaware General Corporation Law, the Series A Preferred Stock shall have no voting rights.

SECTION 13. <u>Status of Shares</u>. Shares of Series A Preferred Stock that have been issued and reacquired in any manner, whether by redemption, repurchase or otherwise or upon any conversion of shares of Series A Preferred Stock to Class A Common Stock, shall thereupon be retired and shall have the status of authorized and unissued shares of preferred stock of the Company undesignated as to series, and may be redesignated as any series of preferred stock of the Company and reissued.

SECTION 14. Term. Except as expressly provided in this Certificate of Designations, the shares of Series A Preferred Stock shall not be redeemable or otherwise mature and the term of the Series A Preferred Stock shall be perpetual.

SECTION 15. Creation of Capital Stock. The Board, without the vote of the Holders, may authorize and issue additional shares of Capital Stock of the Company.

SECTION 16. No Sinking Fund. Shares of Series A Preferred Stock shall not be subject to or entitled to the operation of a retirement or sinking fund.

SECTION 17. Taxes. (a) Transfer Taxes. The Company shall pay any and all stock transfer, documentary, stamp and similar taxes that may be payable in respect of any issuance or delivery of shares of Series A Preferred Stock or shares of Class A Common Stock or other securities issued on account of Series A Preferred Stock pursuant hereto or certificates representing such shares or securities. The Company shall not, however, be required to pay any such tax that may be payable in respect of any transfer involved in the issuance or delivery of shares of Series A Preferred Stock, shares of Class A Common Stock or other securities in a name other than the name in which the shares of Series A Preferred Stock with respect to which such shares or other securities are issued or delivered were registered, or in respect of any payment to any Person other than a payment to the registered holder thereof, and shall not be required to make any such issuance, delivery or payment, unless and until the Person otherwise entitled to such issuance, delivery or payment has paid to the Company the amount of any such tax or has established, to the satisfaction of the Company, that such tax has been paid or is not payable.

(b) <u>Withholding</u>. All payments and distributions (or deemed distributions) on the shares of Series A Preferred Stock (and on the shares of Class A Common Stock received upon their conversion) shall be subject to withholding and backup withholding of taxes to the extent required by law, subject to applicable exemptions, and amounts withheld, if any, shall be treated as received by the Holders. The Company shall use commercially reasonably efforts to notify the Holders of any amounts expected to be deducted and withheld pursuant to the preceding sentence reasonably prior to the relevant payment date and the basis for such deduction and withholding and shall reasonably cooperate with the applicable Holders to reduce or eliminate any such deductions and withholdings to the extent permitted under applicable law.

-22-

(c) <u>Tax Treatment</u>. The Series A Preferred Stock is intended to be treated as common stock that does not constitute "preferred stock" within the meaning of Section 305 of the Internal Revenue Code of 1986, as amended, and the Company shall apply the provisions of this Certificate of Designations consistent with such intention.

SECTION 18. <u>Notices</u>. All notices referred to herein shall be in writing and, unless otherwise specified herein, all notices hereunder shall be deemed to have been given upon the earlier of receipt thereof or three (3) Business Days after the mailing thereof if sent by registered or certified mail (unless first-class mail shall be specifically permitted for such notice under the terms of this Certificate of Designations) with postage prepaid, addressed, (i) if to the Company, to its office at [•] (Attention: [•]), or to any transfer or other agent of the Company designated to receive such notice as permitted by this Certificate of Designations; (ii) if to any Holder, to such Holder at the address of such Holder as listed in the Register; or (iii) to such other address as the Company or any such Holder, as the case may be, shall have designated by notice similarly given.

SECTION 19. <u>Facts Ascertainable</u>. When the terms of this Certificate of Designations refer to a specific agreement or other document to determine the meaning or operation of a provision hereof, the Secretary of the Company shall maintain a copy of such agreement or document at the principal executive offices of the Company and a copy thereof shall be provided free of charge to any Holder who makes a request therefor. The Secretary of the Company shall also maintain a written record of the Issuance Date, the number of shares of Series A Preferred Stock issued to a Holder and the date of each such issuance, and shall furnish such written record free of charge to any Holder who makes a request therefor.

SECTION 20. <u>Waiver</u>. Notwithstanding any provision in this Certificate of Designations to the contrary, any provision contained herein and any right of the Holders of Series A Preferred Stock granted hereunder may be waived as to all shares of Series A Preferred Stock (and the Holders thereof) upon the written consent of the Holders of a majority of the shares of Series A Preferred Stock then outstanding.

SECTION 21. Severability. If any term of the Series A Preferred Stock set forth herein is invalid, unlawful or incapable of being enforced by reason of any rule of law or public policy, all other terms set forth herein, which can be given effect without the invalid, unlawful or unenforceable term will, nevertheless, remain in full force and effect, and no term herein set forth will be deemed dependent upon any other such term, unless so expressed herein.

SECTION 22. No Other Rights. Except as expressly provided in any agreement between a Holder and the Company, the Series A Convertible Preferred Stock will have no rights, preferences or voting powers, except as provided in this Certificate of Designations or the Certificate of Incorporation or as provided by applicable law.

[Signature Page Follows]

-23-

This Certificate of Designations has been approved by the Board in the manner and by the vote required by law.

The undersigned acknowledges this Certificate of Designations to be the corporate act of the Company and, as to all matters or facts required to be verified under oath, the undersigned acknowledges that, to the best of his or her knowledge, information and belief, these matters and facts are true in all material respects and that this statement is made under the penalties for perjury.

IN WITNESS WHEREOF, the Company has caused this Certificate of Designations to be signed in its name and on its behalf by its ______ and attested to by its ______ on this _____ day of $[\bullet]$, 2020.

Name:

Title:

ADAPTHEALTH CORP.

By:

Name: Title:

-24-

Exhibit I ADAPTHEALTH CORP.CONVERSION NOTICE

Reference is made to the Certificate of Designation, Preferences and Rights of the Series A Convertible Preferred Stock of AdaptHealth Corp. (the "Certificate of Designations"). In accordance with and pursuant to the Certificate of Designations, the undersigned hereby elects to convert the number of shares of Series A Convertible Preferred Stock, par value 0.0001 per share (the "Series A Preferred Stock"), of AdaptHealth Corp., a Delaware corporation (the "Company"), indicated below into shares of Class A Common Stock, par value 0.0001 per share, of the Company (the "Class A Common Stock"), [as of the date specified below][[upon/immediately prior to], and subject to the occurrence of, [•]].

Date of Conversion (if applicable):	
Number of shares of Series A Preferred Stock to be converted:	
Share certificate no(s). of Series A Preferred Stock to be converted:	
Tax ID Number (if applicable):	

Please confirm the following information:

Conversion Price:	
Number of shares of	Class A Common Stock to be issued:

Please issue the shares of Class A Common Stock into which the shares of Series A Preferred Stock are being converted in the following name and to the following address:

Issue to:	_
Address:	
Telephone Number:	
Email:	
Authorization:	
By:	_
Title:	_
Dated:	_
Account Number (if electronic book entry transfer):	
Transaction Code Number (if electronic book entry transfer): _	

Payment Instructions for cash payment in lieu of fractional shares:

<u>Annex II</u>

FORM OF AMENDED AND RESTATED REGISTRATION RIGHTS AGREEMENT

(See attached)

AMENDED AND RESTATED REGISTRATION RIGHTS AGREEMENT

This AMENDED AND RESTATED REGISTRATION RIGHTS AGREEMENT (this "<u>Agreement</u>") is dated as of $[\bullet]$, 2020, by and among (i) AdaptHealth Holdings LLC, a Delaware limited liability company (the "<u>Company</u>"), (ii) AdaptHealth Corp., a Delaware corporation ("<u>Pubco</u>"), (iii) each of the Persons listed on the Schedule of Investors attached hereto as of the date hereof, and (iv) each of the other Persons set forth from time to time on the Schedule of Investors who, at any time, own securities of the Company or Pubco and enter into a Joinder to this Agreement agreeing to be bound by the terms hereof (each Person identified in the foregoing (iii) and (iv), an "<u>Investor</u>" and, collectively, the "<u>Investors</u>"). This Agreement shall become effective as of the Closing pursuant to the Investment Agreement (as defined below). Unless otherwise provided in this Agreement, capitalized terms used herein shall have the meanings set forth in <u>Section 12</u> hereof.

WHEREAS, Pubco and certain of the Investors are parties to that certain Registration Rights Agreement, dated as of November 8, 2019 (the "Prior Agreement");

WHEREAS, Pubco and [One Equity Partners VII, L.P.] ("<u>OEP</u>") entered into an Investment Agreement, dated as of [•], 2020 (the "<u>Investment Agreement</u>"), pursuant to which, OEP agreed to purchase and Pubco agreed to sell [•] shares of Series A Convertible Preferred Stock, par value \$0.0001 per share (the "<u>Series A Preferred</u> <u>Stock</u>"), of Pubco;

WHEREAS, Section 13(d) of the Prior Agreement provides that an amendment may occur with the prior written consent of Pubco and the holders of a majority of the Registrable Securities (as such term is used therein) then outstanding (the "Requisite Holders"); and

WHEREAS, Pubco and the Requisite Holders desire to terminate the terms and conditions of the Prior Agreement and to provide for the terms and conditions included herein and to include the recipients of the other Registrable Securities identified herein.

NOW, THEREFORE, in consideration of the mutual covenants contained herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties to this Agreement hereby agree as follows:

1. <u>Resale Shelf Registration Rights</u>.

(a) Registration Statement Covering Resale of Registrable Securities. Pubco filed with the Commission a Registration Statement on Form S-1 ("Form S-1") for an offering to be made on a delayed or continuous basis pursuant to Rule 415 of the Securities Act registering the resale from time to time by the Investors of all of the Registrable Securities held by the Investors (other than OEP) (the "Existing Resale Shelf Registration Statement"). Pubco shall use reasonable best efforts to keep the Existing Resale Shelf Registration Statement continuously effective and to be supplemented and amended to the extent necessary to ensure that such Existing Resale Shelf Registration Statement is available or, if not available, to ensure that another Registration Statement is available (which replacement Registration Statement shall be deemed an Existing Resale Shelf Registration Statement), under the Securities Act at all times until such date as all Registrable Securities covered by the Existing Resale Shelf Registration Statement, including the Registrable Securities held by the OEP Parties to be added to the Existing Resale Shelf Registration Statement in accordance with Section 1(b), have been disposed of in accordance with the intended method(s) of distribution set forth in such Registration Statement or such securities have been withdrawn (the "Effectiveness Period"); provided, that Pubco may amend the Existing Resale Shelf Registration Statement to Rule 415 under the Securities Act (or any successor or similar provision adopted by the Commission then in effect) at any time beginning on the effective date for such Registration Statement (subject to lock-up restrictions provided in this Agreement and in the Lock-Up Agreements), and shall provide that such Registrable Securities may be sold pursuant to any method or combination of methods legally available to, and requested by, the Investor; provided, that Pubco may amend the Existing Resale Shelf Registration Statement in accordance with <u>Section 1(b)</u> and <u>Section 1(c)</u>

Pubco shall use its reasonable best efforts to prepare and file or cause to be prepared and filed with the SEC no later than the date that is sixty (60) (b) days prior to the Restricted Period Termination Date (as defined in the Investment Agreement), either, at the sole discretion of Pubco, (x) a Registration Statement for an offering to be made on a continuous basis pursuant to Rule 415 of the Securities Act (provided, that such Registration Statement replaces and becomes the Existing Resale Shelf Registration Statement) or (y) an amendment or prospectus supplement to the Existing Resale Shelf Registration Statement registering the resale from time to time by the OEP Parties of all of the Registrable Securities held by the OEP Parties (such Registration Statement, as amended, the "OEP Resale Shelf Registration Statement"). The OEP Resale Shelf Registration Statement shall be on Form S-3 or, if Form S-3 is not then available to Pubco, on Form S-1 or such other appropriate form permitting Registration of the Registrable Securities for resale by the OEP Parties. The OEP Resale Shelf Registration Statement shall contain a Prospectus in such form as to permit the OEP Parties to sell the Registrable Securities pursuant to Rule 415 under the Securities Act (or any successor or similar provision adopted by the SEC then in effect) at any time beginning on the effective date for the OEP Resale Shelf Registration Statement (subject to lock-up or transfer restrictions pursuant to Section [5.3] of the Investment Agreement and in this Agreement), and shall provide that the Registrable Securities held by the OEP Parties may be sold pursuant to any method or combination of methods legally available to, and requested by, the OEP Parties. Pubco shall use its reasonable best efforts to cause the OEP Resale Shelf Registration Statement to be declared effective as soon as possible after filing, but in no event later than the Restricted Period Termination Date. Once effective, Pubco shall use its reasonable best efforts to keep the OEP Resale Shelf Registration Statement continuously effective and in compliance with the Securities Act and useable for the resale of the Registrable Securities covered by the OEP Resale Shelf Registration Statement, including by filing successive replacement or renewal OEP Resale Shelf Registration Statements upon the expiration of the OEP Resale Shelf Registration Statement, until such time as the OEP Parties no longer hold Registrable Securities. The OEP Parties shall be entitled, at any time and from time to time when the OEP Resale Shelf Registration Statement is effective (subject to lock-up or transfer restrictions provided the Investment Agreement and in this Agreement), to sell any or all of the Registrable Securities covered by the OEP Resale Shelf Registration Statement. Notwithstanding the first sentence of this Section 1(b), in the event of the occurrence of a Restricted Period Early Termination Event (as defined in the Investment Agreement) prior to the date that is sixty (60) days prior to the Restricted Period Termination Date, Pubco shall use reasonable best efforts to file and cause to be declared effective the OEP Resale Shelf Registration Statement for the Registrable Securities with respect to which the lockup period has terminated within thirty (30) days following such termination. Pubco shall pay all Registration Expenses in connection with the filing of the OEP Resale Shelf Registration Statement.

- 2 -

(c) If any OEP Party becomes a holder of Registrable Securities after the OEP Resale Shelf Registration Statement is declared effective in accordance with Section 1(b), and such OEP Party has executed a Joinder entitling it to the benefits of this Agreement, then Pubco shall, as promptly as is reasonably practicable following delivery of written notice to Pubco of such OEP Party becoming a holder of Registrable Securities and requesting for its name to be included as a selling securityholder in the Prospectus related to the OEP Resale Shelf Registration Statement:

(i) if required and permitted by applicable law, file with the Commission a supplement to the related Prospectus or a post-effective amendment to the OEP Resale Shelf Registration Statement so that such OEP Party is named as a selling securityholder in the OEP Resale Shelf Registration Statement and the related Prospectus in such a manner as to permit such OEP Party to deliver a prospectus to purchasers of the Registrable Securities in accordance with applicable law; provided, however, that the Company shall not be required to file more than one post-effective amendment or a supplement to the related prospectus for such purpose in any 45-day period;

(ii) if, pursuant to Section 1(c)(i), Pubco shall have filed a post-effective amendment to the OEP Resale Shelf Registration Statement that is not automatically effective, use its reasonable best efforts to cause such post-effective amendment to become effective under the Securities Act as promptly as is reasonably practicable; and

Section 1(c)(i).

(iii) notify such OEP Party as promptly as is reasonably practicable after the effectiveness of any post-effective amendment filed pursuant to

(d) Registrations effected pursuant to this <u>Section 1</u> shall not be counted as Demand Registrations effected pursuant to <u>Section 2</u>.

2. <u>Demand Registrations</u>.

(a) <u>Requests for Registration</u>. Subject to the terms and conditions of this Agreement and of the Lock-Up Agreements, at any time or from time to time, the holders of Registrable Securities may request registration under the Securities Act of all or any portion of their Registrable Securities on Form S-1 or any similar long-form registration statement ("<u>Long-Form Registrations</u>") or, if available, on Form S-3 (including a shelf registration pursuant to Rule 415 under the Securities Act) or any similar short-form registration statement, including an automatic shelf registration statement (as defined in Rule 405) (an "<u>Automatic Shelf Registration Statement</u>"), if available to Pubco ("<u>Short-Form Registrations</u>") in accordance with <u>Section 2(b)</u> and <u>Section 2(c)</u> below (such holders being referred to herein as the "<u>Initiating Investors</u>" and all registrations requested by the Initiating Investors being referred to herein as "<u>Demand Registrations</u>"). Each request for a Demand Registration shall specify the approximate number of Registrable Securities requested registration to all other holders of Registrable Securities and, subject to the terms and conditions set forth herein, shall include in such registration under state blue sky laws or in compliance with other registration requirements and in any related underwriting) all such Registrable Securities with respect to which Pubco has received written requests for inclusion therein within five (5) Business Days after the receipt of Pubco's notice. Each holder of Registrable Securities agrees that such holder shall treat as confidential the receipt of the notice of Demand Registration and shall not disclose or use the information contained in such notice of Demand Registration without the prior written consent of Pubco until such time as the information contained therein is or becomes available to the public generally, other than as a result of disclosure by the holder in breach of the terms of this Agreement.

- 3 -

(b) Long-Form Registrations. The OEP Parties may request an aggregate of three (3), and the Investors (other than any OEP Party) holding a majority of the Registrable Securities (other than those held by the OEP Parties) may request one (1) Long-Form Registration in which Pubco shall pay all Registration Expenses whether or not any such Long-Form Registration has become effective; provided that, Pubco shall not be obligated to effect, or to take any action to effect, any Long-Form Registration unless the aggregate market price of the Registrable Securities requested to be registered in such Long-Form Registration exceeds \$20,000,000 at the time of request; provided, further, that Pubco shall only be obligated to effect, or take any action to effect, three (3) Long-Form Registrations in the case of any request therefor by any of the Investors other than the OEP Parties. A registration shall not count as a permitted Long-Form Registration until it has become effective and unless the holders of Registrable Securities are able to register and sell at least 90% of the Registrable Securities requested to be included in such registration; provided that in any event Pubco shall pay all Registration Expenses in connection with any registration initiated as a Long-Form Registration whether or not it has become effective and whether or not such registration has counted as one of the permitted Long-Form Registrations hereunder.

Short-Form Registrations. In addition to the Long-Form Registration provided pursuant to Section 2(b), each of (i) the Investors holding a majority of (c) the Common Units not held by Pubco, (ii) the Investors holding a majority of the Founder Shares, (iii) the Investors holding a majority of the PIPE Shares and (iv) the OEP Parties, in each case, shall be entitled to request an unlimited number of Short-Form Registrations in which Pubco shall pay all Registration Expenses whether or not any such Short-Form Registration has become effective; provided, however, that Pubco shall not be obligated to effect any such Short-Form Registration: (i) if the holders of Registrable Securities, together with the holders of any other securities of Pubco entitled to inclusion in such Short-Form Registration, propose to sell Registrable Securities with an aggregate market price at the time of request of less than \$5,000,000, or (ii) if Pubco has, within the twelve (12) month period preceding the date of such request, already effected three (3) Short-Form Registrations for the holders of Registrable Securities requesting a Short-Form Registration pursuant to this Section 2(c). Demand Registrations shall be Short-Form Registrations whenever Pubco is permitted to use any applicable short form registration and if the managing underwriters (if any) agree to the use of a Short-Form Registration. For so long as Pubco is subject to the reporting requirements of the Exchange Act, Pubco shall use its reasonable best efforts to make Short-Form Registrations available for the offer and sale of Registrable Securities. If Pubco is qualified to and, pursuant to the request of the holders of a majority of the Registrable Securities, has filed with the Commission a registration statement under the Securities Act on Form S-3 pursuant to Rule 415 (a "Shelf Registration"), then Pubco shall use its reasonable best efforts to cause the Shelf Registration to be declared effective under the Securities Act as soon as practicable after filing, and, if Pubco is a WKSI at the time of any such request, to cause such Shelf Registration to be an Automatic Shelf Registration Statement, and once effective, Pubco shall cause such Shelf Registration to remain effective (including by filing a new Shelf Registration, if necessary) for a period ending on the earlier of (i) the date on which all Registrable Securities included in such registration have been sold or distributed pursuant to the Shelf Registration or (ii) the date as of which all of the Registrable Securities included in such registration are able to be sold within a 90-day period in compliance with Rule 144 under the Securities Act. If for any reason Pubco ceases to be a WKSI or becomes ineligible to utilize Form S-3, Pubco shall prepare and file with the Commission a registration statement or registration statements on such form that is available for the sale of Registrable Securities.

- 4

Shelf Takedowns. At any time when the Existing Resale Shelf Registration Statement, OEP Resale Shelf Registration Statement or a Shelf Registration for (d) the sale or distribution by holders of Registrable Securities on a delayed or continuous basis pursuant to Rule 415, including by way of an underwritten offering, block sale or other distribution plan (each, a "Resale Shelf Registration") is effective and its use has not been otherwise suspended by Pubco in accordance with the terms of Section 2(f) below, upon a written demand (a "Takedown Demand") by any Investor that is, in either case, a Shelf Participant holding Registrable Securities at such time (the "Initiating Holder"), Pubco will facilitate in the manner described in this Agreement a "takedown" of Registrable Securities off of such Resale Shelf Registration (a "takedown offering") and Pubco shall pay all Registration Expenses in connection therewith; provided that Pubco will provide (x) in connection with any non-marketed underwritten takedown offering (other than a Block Trade), at least two (2) Business Days' notice of such Takedown Demand to each holder of Registrable Securities (other than the Initiating Holder) that is a Shelf Participant, (y) in connection with any Block Trade initiated prior to November 8, 2022, notice of such Takedown Demand to each holder of Registrable Securities (other than the Initiating Holder) that is a Shelf Participant no later than noon Eastern time on the Business Day prior to the requested Takedown Demand and (z) in connection with any marketed underwritten takedown offering, at least five (5) Business Days' notice of such Takedown Demand to each holder of Registrable Securities (other than the Initiating Holder) that is a Shelf Participant. In connection with (x) any non-marketed underwritten takedown offering initiated prior to November 8, 2022 and (y) any marketed underwritten takedown offering, if any Shelf Participants entitled to receive a notice pursuant to the preceding sentence request inclusion of their Registrable Securities (by notice to Pubco, which notice must be received by Pubco no later than (A) in the case of a non-marketed underwritten takedown offering (other than a Block Trade), the Business Day following the date notice is given to such participant, (B) in the case of a Block Trade, by 10:00 p.m. Eastern time on the date notice is given to such participant and (C) in the case of a marketed underwritten takedown offering, three (3) Business Days following the date notice is given to such participant), the Initiating Holder and the other Shelf Participants that request inclusion of their Registrable Securities shall be entitled to sell their Registrable Securities in such offering. Each holder of Registrable Securities that is a Shelf Participant agrees that such holder shall treat as confidential the receipt of the notice of a Takedown Demand and shall not disclose or use the information contained in such notice without the prior written consent of Pubco until such time as the information contained therein is or becomes available to the public generally, other than as a result of disclosure by the holder in breach of the terms of this Agreement.

- 5 -

(e) <u>Priority on Demand Registrations and Takedown Offerings</u>. Pubco shall not include in any Demand Registration that is an underwritten offering any securities that are not Registrable Securities without the prior written consent of the managing underwriters and the holders of a majority of the Registrable Securities then outstanding. If a Demand Registration or a takedown offering is an underwritten offering and the managing underwriters advise Pubco in writing that in their opinion the number of Registrable Securities and, if permitted hereunder, other securities requested to be included in such offering exceeds the number of Registrable Securities and other securities, if any, which can be sold in an orderly manner in such offering within a price range acceptable to the holders of a majority of the Registrable Securities held by Initiating Investors, Pubco shall include in such offering prior to the inclusion of any securities which are not Registrable Securities the Registrable Securities requested to be included in such registrable Securities requested to be included in such offering securities the Registrable Securities requested to be included in such offering prior to the inclusion of any securities which are not Registrable Securities the Registrable Securities requested to be included in such registrable Securities of such Registrable Securities on the basis of the number of Registrable Securities owned by each such holder).

(f) <u>Restrictions on Demand Registrations and Takedown Offerings</u>. Any demand for the filing of a registration statement or for a registered offering (including a takedown offering) hereunder will be subject to the constraints of any applicable lock-up arrangements to which any demanding Investor is party, and any such demand must be deferred until such lock-up arrangements no longer apply.

(i) Pubco shall not be obligated to effect any Demand Registration within 30 days prior to Pubco's good faith estimate of the date of filing of an underwritten Public Offering of Pubco's securities and for such a period of time after such a filing as the managing underwriters request, provided that such period shall not exceed 90 days from the effective date of any such underwritten Public Offering. Pubco may postpone, for up to 60 days from the date of the request (the "Suspension Period"), the filing or the effectiveness of a registration statement for a Demand Registration or suspend the use of a prospectus that is part of any Resale Shelf Registration (and therefore suspend sales of the Registrable Securities included therein) by providing written notice to the holders of Registrable Securities if the board of directors of Pubco reasonably determines in good faith that the offer or sale of Registrable Securities would be expected to have a material adverse effect on any proposal or plan by Pubco or any subsidiary thereof to engage in any material acquisition or disposition of assets or stock (other than in the ordinary course of business) or any material merger, consolidation, tender offer, recapitalization, reorganization or similar transaction or would require Pubco to disclose any material nonpublic information which would reasonably be likely to be detrimental to Pubco and its subsidiaries; provided that in such event, the holders of Registrable Securities initially requesting such Demand Registration or Takedown Demand shall be entitled to withdraw such request. Pubco may delay or suspend the effectiveness of a Demand Registration or takedown offering pursuant to this <u>Section 2(f)(i)</u> only once in any consecutive twelve-month period; provided that, for the avoidance of doubt, Pubco may in any event delay or suspend the effectiveness of Demand Registration or takedown offering in the case of an event described under <u>Section 5(g)</u> to enable it to comply with its obligations set forth in <u>Section 5(f)</u>.

- 6 -

(ii) In the case of an event that causes Pubco to suspend the use of any Resale Shelf Registration as set forth in <u>Section 2(f)(i)</u> or pursuant to <u>Section 5(g)</u> (a "<u>Suspension Event</u>"), Pubco shall give a notice to the holders of Registrable Securities registered pursuant to such Shelf Registration (a "<u>Suspension Notice</u>") to suspend sales of the Registrable Securities and such notice shall state generally the basis for the notice and that such suspension shall continue only for so long as the Suspension Event or its effect is continuing. A holder of Registrable Securities shall not effect any sales of the Registrable Securities pursuant to such Resale Shelf Registration (or such filings) at any time after it has received a Suspension Notice from Pubco and prior to receipt of an End of Suspension Notice (as defined below). Each holder of Registrable Securities agrees that such holder shall treat as confidential the receipt of the Suspension Notice and shall not disclose or use the information contained in such Suspension Notice without the prior written consent of Pubco until such time as the information contained therein is or becomes available to the public generally, other than as a result of disclosure by such holder in breach of the terms of this Agreement. The holders of Registrable Securities may recommence effecting sales of the Registrable Securities pursuant to the Resale Shelf Registration (or such filings) following further written notice to such effect (an "<u>End of Suspension Notice</u>") from Pubco, which End of Suspension Notice shall be given by Pubco to the holders of Registrable Securities and to such holders' counsel, if any, promptly following the conclusion of any Suspension Event.

(iii) Notwithstanding any provision herein to the contrary, if Pubco shall give a Suspension Notice with respect to any Resale Shelf Registration pursuant to this Section 2(f). Pubco agrees that it shall extend the period of time during which such Resale Shelf Registration shall be maintained effective pursuant to this Agreement by the number of days during the period from the date of receipt by the holders of the Suspension Notice to and including the date of receipt by the holders of the End of Suspension Notice and provide copies of the supplemented or amended prospectus necessary to resume sales, with respect to each Suspension Event; provided that such period of time shall not be extended beyond the date that Common Stock covered by such Resale Shelf Registration are no longer Registrable Securities.

(g) <u>Selection of Underwriters</u>. In connection with any Demand Registration, the Applicable Approving Party shall have the right to select the investment banker(s) and manager(s) to administer the offering; provided that such selection shall be subject to the written consent of Pubco, which consent will not be unreasonably withheld, conditioned or delayed. If any takedown offering is an underwritten offering, the Applicable Approving Party shall have the right to select the investment banker(s) and manager(s) to administer such takedown offering. In each case, the Applicable Approving Party shall have the right to approve the underwriting arrangements with such investment banker(s) and manager(s) on behalf of all holders of Registrable Securities participating in such offering. All Investors proposing to distribute their securities through underwriting shall (together with Pubco and the Company) enter into an underwriting agreement in customary form with the underwriter or underwriters selected for such underwriting.

- 7 -

(h) <u>Other Registration Rights</u>. Pubco represents and warrants to each holder of Registrable Securities that the registration rights granted in this Agreement do not conflict with any other registration rights granted by Pubco. Except as provided in this Agreement, Pubco shall not grant to any Persons the right to request Pubco to register any equity securities of Pubco, or any securities, options or rights convertible or exchangeable into or exercisable for such securities, without the prior written consent of the holders of a majority of the Registrable Securities then outstanding.

(i) <u>Revocation of Demand Notice or Takedown Notice</u>. At any time prior to the effective date of the registration statement relating to a Demand Registration or the "pricing" of any offering relating to a Takedown Demand, the holders of Registrable Securities that requested such Demand Registration or takedown offering may revoke such request for a Demand Registration or takedown offering on behalf of all holders of Registrable Securities participating in such Demand Registration or takedown offering without liability to such holders of Registrable Securities, in each case by providing written notice to Pubco.

3. <u>Piggyback Registrations</u>.

(a) <u>Right to Piggyback</u>. Whenever Pubco proposes to register an offering of any of its securities under the Securities Act (other than (i) pursuant to the Existing Resale Shelf Registration Statement or OEP Resale Shelf Registration Statement, (ii) pursuant to a Demand Registration, (iii) pursuant to a Takedown Demand, (iv) in connection with registrations on Form S-4 or S-8 promulgated by the Commission or any successor forms, (v) a registration relating solely to employment benefit plans, (vi) in connection with a registration the primary purpose of which is to register debt securities, or (vii) a registration on any form that does not include substantially the same information as would be required to be included in a registration statement covering the sale of Registrable Securities) and the registration form to be used may be used for the registration of Registrable Securities of its intention to effect such a Piggyback Registration and, subject to the terms of <u>Sections 3(c)</u> and 3(d) hereof, shall include in such Piggyback Registration (and in all related registrations or qualifications under blue sky laws or in compliance with other registration requirements and in any related underwriting) all Registrable Securities with respect to which Pubco has received written requests for inclusion at any time prior to executing the underwriting agreement or, if none, prior to the applicable registration statement becoming effective.

(b) <u>Piggyback Expenses</u>. The Registration Expenses of the holders of Registrable Securities shall be paid by Pubco in all Piggyback Registrations, whether or not any such registration became effective.

- 8 -

(c) <u>Priority on Primary Registrations</u>. If a Piggyback Registration is an underwritten primary registration on behalf of Pubco, and the managing underwriters advise Pubco in writing that in their opinion the number of securities requested to be included in such registration exceeds the number of securities which can be sold in such offering without adversely affecting the marketability, proposed offering price, timing or method of distribution of the offering, Pubco shall include in such registration (i) first, the securities Pubco proposes to sell, (ii) second, the Registrable Securities requested to be included in such registration by the Investors which, in the opinion of such underwriters, can be sold, without any such adverse effect (pro rata among the holders of such Registrable Securities on the basis of the number of Registrable Securities owned by each such holder), and (iii) third, other securities requested to be included in such registration of such underwriters, can be sold, without any such adverse effect.

(d) <u>Priority on Secondary Registrations</u>. If a Piggyback Registration is an underwritten secondary registration on behalf of holders of Pubco's securities other than holders of Registrable Securities, and the managing underwriters advise Pubco in writing that in their opinion the number of securities requested to be included in such registration exceeds the number of securities which can be sold in such offering without adversely affecting the marketability, proposed offering price, timing or method of distribution of the offering, Pubco shall include in such registration (i) first, the securities requested to be included therein by the holders initially requesting such registration, (ii) second, the Registrable Securities requested to be included in such registration by the Investors which, in the opinion of such underwriters, can be sold, without any such adverse effect (pro rata among the holders of such Registrable Securities on the basis of the number of Registrable Securities owned by each such holder), and (iii) third, other securities requested to be included in such registration which, in the opinion of such underwriters effect.

(e) <u>Other Registrations</u>. If Pubco has previously filed a Registration Statement with respect to Registrable Securities pursuant to <u>Section 2</u> or pursuant to this <u>Section 3</u>, and if such previous registration has not been withdrawn or abandoned, then Pubco shall not be required to file or cause to be effected any other registration of any of its equity securities or securities convertible or exchangeable into or exercisable for its equity securities under the Securities Act (except on Form S-8 or any successor form) at the request of any holder or holders of such securities until a period of at least 90 days has elapsed from the effective date of such previous registration; provided, however, that Pubco shall at all times remain obligated to file or amend, as applicable, any OEP Resale Shelf Registration Statement in accordance with <u>Section 1(b)</u> and/or <u>Section 1(c)</u> in the time periods specified therein.

(f) <u>Right to Terminate Registration</u>. Pubco shall have the right to terminate or withdraw any registration initiated by it under this <u>Section 3</u> whether or not any holder of Registrable Securities has elected to include securities in such registration. The Registration Expenses of such withdrawn registration shall be borne by Pubco in accordance with <u>Section 7</u>.

4. <u>Agreements of Holders</u>.

(a) If required by the managing underwriter(s), in connection with any underwritten Public Offering on or after the date hereof, each holder of Registrable Securities that beneficially owns 1% or more of the outstanding Common Stock shall enter into lock-up agreements with the managing underwriter(s) of such underwritten Public Offering in such form as agreed to by such managing underwriter(s); provided, however, that:

-9-

(i) Deerfield shall not be required to enter into lock-up agreements pursuant to this <u>Section 4(a)</u> on more than two (2) occasions.

(ii) The lock-up agreements to which Deerfield enters into pursuant to this Section 4(a) shall be for a period of not more than sixty (60) days.

(iii) The obligation of Deerfield to enter into lock-up agreements pursuant to this Section 4(a) shall terminate on November 8, 2021.

(iv) Deerfield shall not be required to enter into a lock-up agreement pursuant to this <u>Section 4(a)</u> within six (6) months following the expiration of a previous lock-up agreement pursuant to this <u>Section 4(a)</u>.

(b) The holders of Registrable Securities shall use reasonable best efforts to provide such information as may reasonably be requested by Pubco, or the managing underwriter, if any, in connection with the preparation of any Registration Statement, including amendments and supplements thereto, in order to effect the Registration of any Registrable Securities under the Securities Act pursuant to Section 3 and in connection with Pubco's obligation to comply with federal and applicable state securities laws.

5 . <u>Registration Procedures</u>. In connection with the Registration to be effected pursuant to the Existing Resale Shelf Registration Statement or OEP Resale Shelf Registration Statement, and whenever the holders of Registrable Securities have requested that any Registrable Securities be registered pursuant to this Agreement or have initiated a takedown offering, Pubco shall use its reasonable best efforts to effect the registration and the sale of such Registrable Securities in accordance with the intended method of disposition thereof, and pursuant thereto Pubco shall as expeditiously as reasonably possible:

(a) prepare in accordance with the Securities Act and all applicable rules and regulations promulgated thereunder and file with the Commission a registration statement, and all amendments and supplements thereto and related prospectuses as may be necessary to comply with applicable securities laws, with respect to such Registrable Securities and use its reasonable best efforts to cause such registration statement to become effective (provided that at least five (5) Business Days before filing a registration statement or prospectus or any amendments or supplements thereto, Pubco shall furnish to counsel selected by the Applicable Approving Party copies of all such documents proposed to be filed, which documents shall be subject to the review and comment of such counsel);

(b) notify each holder of Registrable Securities of (A) the issuance by the Commission of any stop order suspending the effectiveness of any registration statement or the initiation of any proceedings for that purpose, (B) the receipt by Pubco or its counsel of any notification with respect to the suspension of the qualification of the Registrable Securities for sale in any jurisdiction or the initiation or threatening of any proceeding for such purpose, and (C) the effectiveness of each registration statement filed hereunder;

- 10 -

(c) prepare and file with the Commission such amendments and supplements to such registration statement and the prospectus used in connection therewith as may be necessary to keep such registration statement effective for a period ending when all of the securities covered by such registration statement have been disposed of in accordance with the intended methods of distribution by the sellers thereof set forth in such registration statement (but not in any event before the expiration of any longer period required under the Securities Act or, if such registration statement relates to an underwritten Public Offering, such longer period as in the opinion of counsel for the underwriters a prospectus is required by law to be delivered in connection with sale of Registrable Securities by an underwriter or dealer) and comply with the provisions of the Securities Act with respect to the disposition of all securities covered by such registration statement during such period in accordance with the intended methods of disposition by the sellers thereof set forth in such registration statement;

(d) furnish to each seller of Registrable Securities thereunder such number of copies of such registration statement, each amendment and supplement thereto, the prospectus included in such registration statement (including each preliminary prospectus), each Free-Writing Prospectus and such other documents as such seller may reasonably request in order to facilitate the disposition of the Registrable Securities owned by such seller;

(e) during any period in which a prospectus is required to be delivered under the Securities Act, promptly file all documents required to be filed with the Commission, including pursuant to Sections 13(a), 13(c), 14 or 15(d) of the Securities Act;

(f) use its reasonable best efforts to register or qualify such Registrable Securities under such other securities or blue sky laws of such jurisdictions as the lead underwriter or the Applicable Approving Party reasonably requests and do any and all other acts and things which may be reasonably necessary or advisable to enable such seller to consummate the disposition in such jurisdictions of the Registrable Securities owned by such seller (provided that Pubco shall not be required to (i) qualify generally to do business in any jurisdiction where it would not otherwise be required to qualify but for this Section 5(f). (ii) consent to general service of process in any such jurisdiction or (iii) subject itself to taxation in any such jurisdiction);

(g) promptly notify in writing each seller of such Registrable Securities (i) after it receives notice thereof, of the date and time when such registration statement and each post-effective amendment thereto has become effective or a prospectus or supplement to any prospectus relating to a registration statement has been filed and when any registration or qualification has become effective under a state securities or blue sky law or any exemption thereunder has been obtained, (ii) after receipt thereof, of any request by the Commission for the amendment or supplementing of such registration statement or prospectus or for additional information, and (iii) at any time when a prospectus relating thereto is required to be delivered under the Securities Act, of the happening of any event as a result of which the prospectus included in such registration statement contains an untrue statement of a material fact or omits any fact necessary to make the statements therein not misleading, and, at the request of any such seller, Pubco promptly shall prepare, file with the Commission and furnish to each such seller a reasonable number of copies of a supplement or amendment to such prospectus so that, as thereafter delivered to the purchasers of such Registrable Securities, such prospectus shall not contain an untrue statement of a material fact or omit to state any fact necessary to make the statements therein not misleading;

- 11 -

(h) cause all such Registrable Securities to be listed on each securities exchange on which similar securities issued by Pubco are then listed and, if not so listed, to be listed on a securities exchange and, without limiting the generality of the foregoing, to arrange for at least two market makers to register as such with respect to such Registrable Securities with FINRA;

(i) provide a transfer agent and registrar for all such Registrable Securities not later than the effective date of such registration statement;

(j) enter into and perform such customary agreements (including underwriting agreements in customary form) and take all such other actions as the Applicable Approving Party or the underwriters, if any, reasonably request in order to expedite or facilitate the disposition of such Registrable Securities (including, without limitation, effecting a stock split or a combination of shares and preparing for and participating in such number of "road shows", investor presentations and marketing events as the underwriters managing such offering may reasonably request);

(k) make available for inspection by any seller of Registrable Securities, any underwriter participating in any disposition pursuant to such registration statement and any attorney, accountant or other agent retained by any such seller or underwriter, all financial and other records, pertinent corporate and business documents and properties of Pubco as shall be necessary to enable them to exercise their due diligence responsibility, and cause Pubco's officers, managers, directors, employees, agents, representatives and independent accountants to supply all information reasonably requested by any such seller, underwriter, attorney, accountant or agent in connection with such registration statement;

(1) take all reasonable actions to ensure that any Free-Writing Prospectus utilized in connection with any Demand Registration (including any Shelf Registration), takedown offering or Piggyback Registration hereunder complies in all material respects with the Securities Act, is filed in accordance with the Securities Act to the extent required thereby, is retained in accordance with the Securities Act to the extent required thereby and, when taken together with the related prospectus, shall not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements therein, in light of the circumstances under which they were made, not misleading;

(m) otherwise use its reasonable best efforts to comply with all applicable rules and regulations of the Commission;

(n) permit any holder of Registrable Securities who, in its good faith judgment (based on the advice of counsel), could reasonably be expected to be deemed to be an underwriter or a controlling Person of Pubco to participate in the preparation of such registration or comparable statement and to require the insertion therein of material furnished to Pubco in writing, which in the reasonable judgment of such holder and its counsel should be included;

- 12 -

(o) in the event of the issuance of any stop order suspending the effectiveness of a registration statement, or of any order suspending or preventing the use of any related prospectus or suspending the qualification of any Common Stock included in such registration statement for sale in any jurisdiction, use its reasonable best efforts promptly to obtain the withdrawal of such order;

(p) use its reasonable best efforts to cause such Registrable Securities covered by such registration statement to be registered with or approved by such other governmental agencies or authorities as may be necessary to enable the sellers thereof to consummate the disposition of such Registrable Securities;

(q) cooperate with the holders of Registrable Securities covered by the registration statement and the managing underwriter or agent, if any, to facilitate the timely preparation and delivery of certificates (not bearing any restrictive legends) representing securities to be sold under the registration statement and enable such securities to be in such denominations and registered in such names as the managing underwriter, or agent, if any, or such holders may request;

(r) cooperate with each holder of Registrable Securities covered by the registration statement and each underwriter or agent participating in the disposition of such Registrable Securities and their respective counsel in connection with any filings required to be made with FINRA;

(s) if such registration includes an underwritten Public Offering, use its reasonable best efforts to obtain a cold comfort letter from Pubco's independent public accountants and addressed to the underwriters, in customary form and covering such matters of the type customarily covered by cold comfort letters as the underwriters in such registration reasonably request;

(t) provide a legal opinion of Pubco's outside counsel, dated the effective date of such registration statement (and, if such registration includes an underwritten Public Offering, dated the date of the closing under the underwriting agreement), with respect to the registration statement, each amendment and supplement thereto, the prospectus included therein (including the preliminary prospectus) and such other documents relating thereto in customary form and covering such matters of the type customarily covered by legal opinions of such nature, which opinion shall be addressed to the underwriters;

(u) if Pubco files an Automatic Shelf Registration Statement covering any Registrable Securities, use its reasonable best efforts to remain a WKSI (and not become an ineligible issuer (as defined in Rule 405)) during the period during which such Automatic Shelf Registration Statement is required to remain effective;

(v) if Pubco does not pay the filing fee covering the Registrable Securities at the time an Automatic Shelf Registration Statement is filed, pay such fee at such time or times as the Registrable Securities are to be sold; and

- 13 -

(w) subject to the terms of <u>Section 2(c)</u> and <u>Section 2(d)</u>, if an Automatic Shelf Registration Statement has been outstanding for at least three (3) years, at the end of the third year, refile a new Automatic Shelf Registration Statement covering the Registrable Securities, and, if at any time when Pubco is required to re-evaluate its WKSI status Pubco determines that it is not a WKSI, use its reasonable best efforts to refile the registration statement on Form S-3 and keep such registration statement effective (including by filing a new Resale Shelf Registration or Shelf Registration, if necessary) during the period throughout which such registration statement is required to be kept effective.

6 . <u>Termination of Rights</u>. Notwithstanding anything contained herein to the contrary, the right of any Investor to include Registrable Securities in any Demand Registration or any Piggyback Registration shall terminate on such date that such Investor (together with its Affiliates) beneficially owns less than 1% of the outstanding Common Stock on an as-converted basis and may sell all of the Registrable Securities owned by such Investor pursuant to Rule 144 of the Securities Act without any restrictions as to volume or the manner of sale or otherwise; provided, however, that with respect to any Investor whose rights have terminated pursuant to this <u>Section 6</u>, if following such a termination, such Investor loses the ability to sell all of its Registrable Securities pursuant to Rule 144 of the Securities act without any restrictions as to volume or the manner of sale or otherwise due to a change in interpretive guidance by the Commission, then such Investor's right to include Registrable Securities in any Demand Registration or any Piggyback Registration shall be reinstated until such time as the Investor is once again able to sell all of its Registrable Securities Act without any restrictions as to volume or the manner of sale or otherwise.

7. <u>Registration Expenses</u>.

(a) All expenses incident to Pubco's performance of or compliance with this Agreement, including, without limitation, all registration, qualification and filing fees, listing fees, fees and expenses of compliance with securities or blue sky laws, stock exchange rules and filings, printing expenses, messenger and delivery expenses, fees and disbursements of custodians, and fees and disbursements of counsel for Pubco and all independent certified public accountants, underwriters (excluding underwriting discounts and commissions) and other Persons retained by Pubco (all such expenses being herein called "<u>Registration Expenses</u>"), shall be borne by Pubco as provided in this Agreement and, for the avoidance of doubt, Pubco also shall pay all of its internal expenses (including, without limitation, all salaries and expenses of its officers and employees performing legal or accounting duties), the expense of any annual audit or quarterly review, the expense of any liability insurance and the expenses and fees for listing the securities to be registered on each securities exchange on which similar securities issued by Pubco are then listed. Each Person that sells securities pursuant to a Demand Registration, a Takedown Demand or Piggyback Registration hereunder shall bear and pay all underwriting discounts and commissions and transfer taxes applicable to the securities sold for such Person's account.

(b) Pubco shall reimburse the holders of Registrable Securities included in such registration for the reasonable fees and disbursements of one counsel chosen by the Applicable Approving Party and one local counsel (if necessary) for each applicable jurisdiction and chosen by the applicable holder of Registrable Securities, in each case, for the purpose of rendering a legal opinion on behalf of such holders in connection with any underwritten Demand Registration, takedown offering or Piggyback Registration.

- 14 -

8. Additional Payments Under Certain Circumstances.

(a) [Payments ("<u>Additional Payments</u>") with respect to the shares of Common Stock included in the Registrable Securities shall be assessed as follows if the following event occurs (such event being herein called a "<u>Registration Default</u>"): the Existing Resale Shelf Registration Statement ceases to be effective prior to the expiration of the Effectiveness Period (unless and except to the extent that another Registration Statement covering the applicable Registrable Securities is effective during the Effectiveness Period).

Additional Payments shall accrue on the applicable Registrable Securities for each such day from and including the date on which any such (b) Registration Default occurs to but excluding the date on which all such Registration Defaults have been cured at a rate of \$0.05 per share (subject to proportionate adjustment in the event of any stock split, reverse stock split or other recapitalization) per month or portion thereof (on a 30/360 basis); provided, however, that the Company's obligation to pay Additional Payments extends only to any shares of Common Stock included in the Registrable Securities that are affected by the Registration Default; and provided further that Additional Payments shall in no event accrue on account of any Registrable Securities during any period that such Registrable Securities may not be sold pursuant to the terms of the Lock-Up Agreements or any other applicable lock-up arrangements to which the applicable Investor is party; provided, however, that notwithstanding anything to the contrary herein, no Additional Payments were accrued or are payable for any Registration Default in the 180 day period following November 8, 2019, and each Investor waives any entitlement thereto. Other than the obligation of payment of any Additional Payments in accordance with the terms hereof, the Company will have no other liabilities for monetary damages with respect to its registration obligations. With respect to each Investor, the Company's obligations to pay Additional Payments remain in effect only so long as the applicable shares of Common Stock held by the Investor are Registrable Securities. Notwithstanding anything to the contrary contained herein, (i) in no event shall the aggregate of all Additional Payments payable by the Company hereunder on account of any share of Common Stock exceed \$0.50 per share (subject to proportionate adjustment in the event of any stock split, reverse stock split or other recapitalization), (ii) no Additional Payments shall accrue during any Suspension Period, (iii) a Registration Default shall be deemed not to have occurred and be continuing, and no Additional Payments shall accrue as a result thereof, if the Registration Default (x) relates to any information supplied or failed to be supplied by an Investor in relation to any Registration Statement or the related Prospectus or (y) arises due to the filing by Pubco of any post-effective amendment to the Existing Resale Shelf Registration Statement in connection with Pubco's obligation to comply with Section 1(b) or Section 1(c) (but only until such post-effective amendment is declared effective by the Commission), and (iv) no Additional Payments shall accrue or be payable to any OEP Party or in respect of the Registrable Securities issued pursuant to the Investment Agreement. No Additional Payments shall be payable (i) if as of the relevant Registration Default, the Registrable Securities may be sold by the Investors without volume or manner of sale restrictions under Rule 144, as determined by counsel to the Company pursuant to a written opinion letter to such effect, addressed and reasonably acceptable to the Company's transfer agent or (ii) with respect to any period after the expiration of the Effectiveness Period (it being understood that this clause shall not relieve the Company of any Additional Payments accruing prior to the expiration of the Effectiveness Period).

- 15 -

(c) Any amounts of Additional Payments pursuant to this Section 8 will be payable in cash in arrears on the last day of each month following the date on which a Registration Default occurs. The amount of Additional Payments will be determined on the basis of a 360-day year comprised of twelve 30-day months, and the actual number of days on which Additional Payments accrued during such period.]¹

9. <u>Indemnification</u>.

Pubco agrees to (i) indemnify and hold harmless, to the fullest extent permitted by law, each Investor and their respective officers, directors, members, (a) partners, agents, affiliates and employees and each Person who controls such Investor (within the meaning of the Securities Act or the Exchange Act) against all losses, claims, actions, damages, liabilities and expenses caused by (A) any untrue or alleged untrue statement of material fact contained in any registration statement, prospectus or preliminary prospectus or any amendment thereof or supplement thereto or any omission or alleged omission of a material fact required to be stated therein or necessary to make the statements therein not misleading, or (B) any violation or alleged violation by Pubco of the Securities Act or any other similar federal or state securities laws or any rule or regulation promulgated thereunder applicable to Pubco and relating to action or inaction required of Pubco in connection with any such registration, qualification or compliance, and (ii) pay to each Investor and their respective officers, directors, members, partners, agents, affiliates and employees and each Person who controls such Investor (within the meaning of the Securities Act or the Exchange Act), as incurred, any legal and any other expenses reasonably incurred in connection with investigating, preparing or defending any such claim, loss, damage, liability or action, except insofar as the same are caused by or contained in any information furnished in writing to Pubco or any managing underwriter by such Investor expressly for use therein; provided, however, that the indemnity agreement contained in this Section 9 shall not apply to amounts paid in settlement of any such claim, loss, damage, liability or action if such settlement is effected without the consent of Pubco (which consent shall not be unreasonably withheld, conditioned or delayed), nor shall Pubco be liable in any such case for any such claim, loss, damage, liability or action to the extent that it solely arises out of or is based upon an untrue statement of any material fact contained in the registration statement or omission to state therein a material fact required to be stated therein or necessary to make the statements therein not misleading, in each case to the extent that such untrue statement or alleged untrue statement or omission or alleged omission was made in the registration statement, in reliance upon and in conformity with written information furnished by such Investor expressly for use in connection with such registration statement. In connection with an underwritten offering, Pubco shall indemnify any underwriters or deemed underwriters, their officers and directors and each Person who controls such underwriters (within the meaning of the Securities Act or the Exchange Act) to the same extent as provided above with respect to the indemnification of the holders of Registrable Securities.

¹ Note to Draft: Amendments from the Prior Agreement to only be made if Deerfield consents to this Agreement, otherwise this Section 8 shall read as set forth in the Prior



(b) In connection with any registration statement in which a holder of Registrable Securities is participating, each such holder shall furnish to Pubco in writing such information relating to such holder as Pubco reasonably requests for use in connection with any such registration statement or prospectus and, to the extent permitted by law, shall indemnify Pubco, its officers, directors, employees, agents and representatives and each Person who controls Pubco (within the meaning of the Securities Act) against any losses, claims, damages, liabilities and expenses resulting from any untrue or alleged untrue statement of material fact contained in the registration statement, prospectus or preliminary prospectus or any amendment thereof or supplement thereto or any omission or alleged omission of a material fact required to be stated therein or necessary to make the statements therein not misleading, but only to the extent that such untrue statement or omission is contained in any information so furnished in writing by such holder; provided that the obligation to indemnify shall be individual, not joint and several, for each holder and shall be limited to the net amount of proceeds actually received by such holder from the sale of Registrable Securities pursuant to such registration statement.

(c) Any Person entitled to indemnification hereunder shall (i) give prompt written notice to the indemnifying party of any claim with respect to which it seeks indemnification (provided that the failure to give prompt notice shall not impair any Person's right to indemnification hereunder to the extent such failure has not materially prejudiced the indemnifying party) and (ii) unless in such indemnified party's reasonable judgment a conflict of interest between such indemnified and indemnifying parties may exist with respect to such claim, permit such indemnifying party to assume the defense of such claim with counsel reasonably satisfactory to the indemnified party. If such defense is assumed, the indemnifying party shall not be subject to any liability for any settlement made by the indemnified party without its consent (but such consent shall not be unreasonably withheld, conditioned or delayed). An indemnifying party who is not entitled to, or elects not to, assume the defense of a claim shall not be obligated to pay the fees and expenses of more than one counsel (as well as one local counsel for each applicable jurisdiction) for all parties indemnified party and any other of such indemnified party a conflict of interest may exist between such indemnified party and any other of such indemnified parties with respect to such claim. In such instance, the conflicted indemnified parties shall have a right to retain one separate counsel, closen by the holders of a majority of the Registrable Securities included in the registration, at the expense of the indemnifying party. No indemnifying party. No inclemnifying party, in the defense of such claim or litigation, shall, except with the consent of each indemnified party, consent to the entry of any judgment or enter into any settlement which does not include as an unconditional term thereof the giving by the claimant or plaintiff to such indemnified party of a release from all liability in respect to such claim or litigation.

- 17 -

Each party hereto agrees that, if for any reason the indemnification provisions contemplated by Sections 9(a) or 9(b) are unavailable to or insufficient to (d) hold harmless an indemnified party in respect of any losses, claims, damages, liabilities or expenses (or actions in respect thereof) referred to therein, then each indemnifying party, in lieu of indemnifying such indemnified party, shall contribute to the amount paid or payable by such indemnified party as a result of such losses, claims, damages, liabilities or expenses (or actions in respect thereof) in such proportion as is appropriate to reflect the relative fault of the indemnifying party and the indemnified party in connection with the actions which resulted in such losses, claims, damages, liabilities or expenses, as well as any other relevant equitable considerations. The relative fault of such indemnifying party and indemnified party shall be determined by reference to, among other things, whether any action in question, including any untrue or alleged untrue statement of a material fact or omission or alleged omission to state a material fact, relates to information supplied by such indemnifying party or indemnified party, and the parties' relative intent, knowledge, access to information and opportunity to correct or prevent such statement or omission. The parties hereto agree that it would not be just or equitable if contribution pursuant to this Section 9(d) were determined by pro rata allocation (even if the holders or any underwriters or all of them were treated as one entity for such purpose) or by any other method of allocation which does not take account of the equitable considerations referred to in this Section 9(d). The amount paid or payable by an indemnified party as a result of the losses, claims, damages, liabilities or expenses (or actions in respect thereof) referred to above shall be deemed to include any legal or other fees or expenses reasonably incurred by such indemnified party in connection with investigating or, except as provided in Section 9(c), defending any such action or claim. No Person guilty of fraudulent misrepresentation (within the meaning of Section 11(f) of the Securities Act) shall be entitled to contribution from any Person who was not guilty of such fraudulent misrepresentation. The sellers' obligations in this Section 9(d) to contribute shall be several in proportion to the amount of securities registered by them and not joint and shall be limited to an amount equal to the net proceeds actually received by such seller from the sale of Registrable Securities effected pursuant to such registration.

(e) The indemnification and contribution provided for under this Agreement shall remain in full force and effect regardless of any investigation made by or on behalf of the indemnified party or any officer, director or controlling Person of such indemnified party and shall survive the transfer of Registrable Securities and the termination or expiration of this Agreement.

10. Participation in Underwritten Registrations. No Person may participate in any registration hereunder which is underwritten unless such Person (a) agrees to sell such Person's securities on the basis provided in any underwriting arrangements approved by the Person or Persons entitled hereunder to approve such arrangements (including, without limitation, pursuant to any over-allotment or "green shoe" option requested by the underwriters; provided that no holder of Registrable Securities shall be required to sell more than the number of Registrable Securities such holder has requested to include) and (b) completes and executes all questionnaires, powers of attorney, custody agreements, stock powers, indemnities, underwritten registration shall be required to make any representations or warranties to Pubco or the underwriters (other than representations and warranties regarding such holder, such holder's title to the securities, such Person's authority to sell such securities and such holder's intended method of distribution) or to underake any indemnification obligations to Pubco or the underwriters with respect thereto that are materially more burdensome than those provided in <u>Section 9</u>. Each holder of Registrable Securities shall execute and deliver such other agreements as may be reasonably requested by Pubco and the lead managing underwriter(s) that are consistent with such holder's obligations under <u>Section 4</u> and this <u>Section 10</u> or that are necessary to give further effect thereto. To the extent that any such agreement is entered into pursuant to, and consistent with, <u>Section 4</u> and this <u>Section 10</u>, the respective rights and obligations created under such agreement shall supersede the respective rights and obligations of the holders, Pubco and the underwriters created pursuant to this <u>Section 10</u>.

- 18 -

11. Other Agreements. Pubco shall file all reports required to be filed by it under the Securities Act and the Exchange Act and the rules and regulations adopted by the Commission thereunder and shall take such further action as the Investors may reasonably request, all to the extent required to enable such Persons to sell securities pursuant to (a) Rule 144 adopted by the Commission under the Securities Act (as such rule may be amended from time to time) or any similar rule or regulation hereafter adopted by the Commission or (b) a registration statement on Form S-1 or any similar registration form hereafter adopted by the Commission. Upon request, Pubco shall deliver to the Investors a written statement as to whether it has complied with such requirements. Pubco shall at all times use its reasonable best efforts to cause the securities so registered to continue to be listed on one or more of the New York Stock Exchange, the American Stock Exchange and the Nasdaq Stock Market. Pubco shall use its best efforts to facilitate and expedite transfers of Registrable Securities pursuant to Rule 144, which efforts shall include timely notice to its transfer agent to expedite such transfers of Registrable Securities.

12. <u>Definitions</u>.

(a) "Affiliate" means, with respect to any specified Person, any other Person that, at the time of determination, directly or indirectly through one or more intermediaries, controls, is controlled by or is under common control with such specified Person.

(b) "<u>Applicable Approving Party</u>" means the holders of a majority of the Registrable Securities participating in the applicable offering or, in the case of a Short-Form Registration effected pursuant to <u>Section 2(c)</u>, the holders of a majority of the type of Registrable Securities that initiated such Short-Form Registration.

(c) "<u>as-converted basis</u>" means, for purposes of computing beneficial ownership, such number of shares of Common Stock calculated on a basis assuming all shares of Series A Preferred Stock had been converted by the holders thereof pursuant to Section [6] of the Certificate of Designations classifying the Series A Preferred Stock, but disregarding any restrictions or limitations upon the conversion of such Series Preferred Stock[, including those limitations specified in Section [8(f)] thereof].

(d) "Block Trade" means any non-marketed underwritten takedown offering taking the form of a bought deal or block sale to a financial institution.

(e) "<u>Business Day</u>" means any day that is not a Saturday or Sunday or a legal holiday in the state in which Pubco's chief executive office is located or in New York, NY.

(f) "<u>Closing</u>" means the closing of the sale of Series A Preferred Stock to OEP pursuant to Section [1.2] of the Investment Agreement.

(g) "<u>Commission</u>" means the U.S. Securities and Exchange Commission.

- 19 -

- (h) "<u>Common Stock</u>" means the Class A Common Stock of Pubco, par value \$0.0001 per share.
- (i) "<u>Common Unit</u>" has the meaning set forth in the LLC Agreement.
- (j) "Deerfield" means Deerfield Private Design Fund IV, L.P., a Delaware limited partnership.

(k) "Exchange Act" means the Securities Exchange Act of 1934, as amended from time to time, or any successor federal law then in force, together with all rules and regulations promulgated thereunder.

(l) "<u>FINRA</u>" means the Financial Industry Regulatory Authority.

(m) "Founder Shares" means the 6,250,000 shares of Common Stock issued to the Original Holders prior to Pubco's initial public offering.

(n) "Free-Writing Prospectus" means a free-writing prospectus, as defined in Rule 405 of the Securities Act.

(o) "<u>LLC Agreement</u>" means the Fifth Amended and Restated Limited Liability Company Agreement of the Company, dated as of or about the date hereof, by and among the Company, Pubco and the other members of the Company (as the same may be amended, supplemented or modified from time to time in accordance with the terms thereof).

(p) "Lock-Up Agreements" means those certain Lock-Up Agreements, dated as of July 8, 2019, by and among Pubco, the Company, and certain of the Persons listed on the Schedule of Investors attached hereto.

(q) <u>"Merger Agreement</u>" means the Agreement and Plan of Merger, dated as of July 8, 2019, by and between Pubco, the Company and certain other parties, as amended.

(r) "<u>Person</u>" means an individual, a partnership, a corporation, a limited liability company, an association, a joint stock company, a trust, a joint venture, an unincorporated organization and a governmental entity or any department, agency or political subdivision thereof.

(s) "<u>PIPE Shares</u>" means the 12,500,000 shares of Common Stock issued to the one of the Investors pursuant to that certain Subscription Agreement, dated as of July 8, 2019, by and between Pubco and such Investor.

(t) "Prospectus" means the prospectus included in any Registration Statement, as supplemented by any and all prospectus supplements and as amended by any and all post-effective amendments and including all material incorporated by reference in such prospectus.

(u) "Public Offering" means any sale or distribution by Pubco and/or holders of Registrable Securities to the public of Common Stock pursuant to an offering registered under the Securities Act.

- 20 -

(v) "<u>OEP Party</u>" or "<u>OEP Parties</u>" means OEP and each Affiliate of OEP to whom shares of Series A Preferred Stock or Common Stock are transferred pursuant to Section [5.3] of the Investment Agreement.

(w) "Original Holders" shall mean each of Chris Wolfe, Steven Hochberg, Dr. Mohit Kaushal, Dr. Gregory Sorensen and Dr. Susan Weaver.

(x) "Registered" and "Registration" mean a registration effected by preparing and filing a Registration Statement or similar document in compliance with the requirements of the Securities Act, and the applicable rules and regulations promulgated thereunder, and such Registration Statement becoming effective.

(y) "Registrable Securities" means (i) any Common Stock held by the Investors, (ii) any Founder Shares held by the Investors, (iii) any Private Placement Warrants (or underlying securities) held by the Investors, (iv) any PIPE Shares held by the Investors (v) any Common Stock issued to an Investor pursuant to the terms of the Merger Agreement, (vi) any Common Stock issued to OEP pursuant to the Investment Agreement (whether or not such shares of Series A Preferred Stock or Common Stock are subsequently transferred to any OEP Party) held by any OEP Party (vii) any Common Stock issued upon conversion of the Series A Preferred Stock issued to OEP pursuant to the Investment Agreement (whether or not such shares of Series A Preferred Stock or Common Stock are subsequently transferred to any OEP Party) held by any OEP Party or (viii) any Common Stock issued or issuable with respect to the securities referred to in the preceding clauses (i) through (vii) by way of a stock dividend or stock split or in connection with a combination of shares, recapitalization, merger, consolidation or other reorganization. As to any particular Registrable Securities, such securities shall cease to be Registrable Securities when they have been sold or distributed to the public pursuant to an offering registered under the Securities Act or sold to the public through a broker, dealer or market maker in compliance with Rule 144 following the date hereof or repurchased by Pubco or any of its subsidiaries. For purposes of this Agreement, a Person shall be deemed to be a holder of Registrable Securities, and the Registrable Securities shall be deemed to be in existence, whenever such Person has the right to acquire directly or indirectly such Registrable Securities (upon conversion or exercise in connection with a transfer of securities or otherwise, but disregarding any restrictions or limitations upon the conversion or exercise of such right), whether or not such acquisition has actually been effected, and such Person shall be entitled to e

(z) "<u>Registration Statement</u>" means any registration statement filed by Pubco with the Commission in compliance with the Securities Act and the rules and regulations promulgated thereunder for a public offering and sale of Common Stock or Registrable Securities, including the Prospectus included in such registration statement, amendments (including post-effective amendments) and supplements to such registration statement, and all exhibits to and all material incorporated by reference in such registration statement (other than a registration statement on Form S-8, or their successors).

- 21 -

(aa) "Rule 144", "Rule 158", "Rule 405", "Rule 415" and "Rule 430B" mean, in each case, such rule promulgated under the Securities Act (or any successor provision) by the Commission, as the same shall be amended from time to time, or any successor rule then in force.

(bb) "Securities Act of 1933, as amended from time to time, or any successor federal law then in force, together with all rules and regulations promulgated thereunder.

(cc) "<u>Shelf Participant</u>" means any holder of Registrable Securities listed as a potential selling stockholder in connection with the Existing Resale Shelf Registration Statement, OEP Resale Shelf Registration Statement or the Shelf Registration, as applicable, or any such holder that could be added to such Existing Resale Shelf Registration Statement, OEP Resale Shelf Registration Statement or Shelf Registration without the need for a post-effective amendment thereto or added by means of an automatic post-effective amendment thereto.

(dd) "<u>WKSI</u>" means a "well-known seasoned issuer" as defined under Rule 405.

13. Miscellaneous.

(a) <u>No Inconsistent Agreements</u>. Neither the Company nor Pubco shall not hereafter enter into any agreement with respect to its securities which is inconsistent with or violates or in any way impairs the rights granted to the Investors in this Agreement.

(b) Entire Agreement; Effectiveness. This Agreement constitutes the entire agreement of the parties hereto with respect to the subject matter hereof and supersedes all prior agreements, understandings, negotiations and discussions among the parties hereto, written or oral, with respect to the subject matter hereof, including without limitation, but subject to the remainder of this Section 13(b), the Prior Agreement. This Agreement shall be automatically effective as of the Closing (as defined in the Investment Agreement), without further action by any party hereto. Prior to the Closing, the Prior Agreement shall remain in effect. If the Investment Agreement is terminated for any reason, then this amendment shall be void and of no force and effect.

(c) <u>Remedies</u>. Any Person having rights under any provision of this Agreement shall be entitled to enforce such rights specifically (without posting a bond or other security), to recover damages caused by reason of any breach of any provision of this Agreement and to exercise all other rights granted by law. The parties hereto agree and acknowledge that money damages would not be an adequate remedy for any breach of the provisions of this Agreement and that, in addition to any other rights and remedies existing in its favor, any party shall be entitled to specific performance and/or other injunctive relief from any court of law or equity of competent jurisdiction (without posting any bond or other security) in order to enforce or prevent violation of the provisions of this Agreement.

(d) <u>Amendments and Waivers</u>. Except as otherwise provided herein, the provisions of this Agreement may be amended or waived only with the prior written consent of Pubco and the holders of a majority of the Registrable Securities then outstanding; <u>provided</u>, that no amendment may materially and disproportionately adversely affect the rights of any holder of Registrable Securities compared to other holders of Registrable Securities without the consent of such adversely affected holder; and <u>provided further</u>, that the definition of "<u>Effectiveness Period</u>" and <u>Section 8</u> (Additional Payments Under Certain Circumstances) may not be amended without the prior written consent of Deerfield. Any amendment or waiver effected in accordance with this <u>Section 13(d)</u> shall be binding upon each Investor, Pubco and the Company. The failure of any party to enforce any of the provisions of this Agreement shall in no way be construed as a waiver of such provisions and shall not affect the right of such party thereafter to enforce each and every provision of this Agreement in accordance with its terms.

- 22 -

(e) <u>Successors and Assigns</u>. All covenants and agreements in this Agreement by or on behalf of any of the parties hereto shall bind and inure to the benefit of the respective successors and assigns of the parties hereto whether so expressed or not. In addition, whether or not any express assignment has been made, the provisions of this Agreement which are for the benefit of purchasers or holders of Registrable Securities are also for the benefit of, and enforceable by, any subsequent holder of Registrable Securities.

(f) <u>Severability</u>. Whenever possible, each provision of this Agreement shall be interpreted in such manner as to be effective and valid under applicable law, but if any provision of this Agreement is held to be prohibited by or invalid, illegal or unenforceable in any respect under any applicable law, such provision shall be ineffective only to the extent of such prohibition, invalidity, illegality or unenforceability, without invalidating the remainder of this Agreement.

(g) <u>Counterparts</u>. This Agreement may be executed simultaneously in counterparts (including by means of telecopied, facsimile or portable data format (PDF) signature pages), any one of which need not contain the signatures of more than one party, but all such counterparts taken together shall constitute one and the same Agreement.

(h) <u>Descriptive Headings: Interpretation</u>. The descriptive headings of this Agreement are inserted for convenience only and do not constitute a part of this Agreement. The use of the word "including" herein shall mean "including without limitation."

(i) <u>Governing Law; Jurisdiction</u>. All issues and questions concerning the construction, validity, enforcement and interpretation of this Agreement and the exhibits and schedules hereto shall be governed by, and construed in accordance with, the laws of the State of Delaware, without giving effect to any choice of law or conflict of law rules or provisions (whether of the State of Delaware or any other jurisdiction) that would cause the application of the laws of any jurisdiction other than the State of Delaware. The parties hereto agree that any suit, action or proceeding seeking to enforce any provision of, or based on any matter arising out of or in connection with, this Agreement or the transactions contemplated hereby shall be brought in any Delaware Chancery Court, or if such court does not have subject matter jurisdiction, any court of the United States located in the State of Delaware. Each of the parties hereby irrevocably consents to the jurisdiction of such courts (and of the appropriate appellate courts therefrom) in any such suit, action or proceeding and irrevocably waives, to the fullest extent permitted by law, any objection that it may now or hereafter have to the laying of the venue of any such suit, action or proceeding in any such suit, action or proceeding brought in any such court has been brought in an inconvenient forum Process in any such suit, action or proceeding may be served on any party anywhere in the world, whether within or without the jurisdiction of any such court.

- 23 -

(j) Notices. All notices, demands or other communications to be given or delivered under or by reason of the provisions of this Agreement shall be in writing and shall be given (and shall be deemed to have been duly given upon receipt) by delivery in person, by or email or by registered or certified mail (postage prepaid, return receipt requested) to each Investor at the address indicated on the Schedule of Investors attached hereto and to Pubco and the Company at the addresses indicated below (or at such other address for a party as shall be specified in a notice given in accordance with this <u>Section 13(j)</u>):

if to Pubco:

AdaptHealth Corp. 780 Third Avenue New York, New York 10017 Telephone: (212) 551-1600 Attention: Chris Wolfe Email: chris.wolfe@dfbhealthcare.com

with a copy to:

Willkie Farr & Gallagher LLP 787 Seventh Avenue New York, New York 10019 Attention: Steven J. Gartner, Michael E. Brandt; and Danielle Scalzo Facsimile: (212) 728-8111 Email: sgartner@willkie.com; mbrandt@willkie.com; and DScalzo@willkie.com

if to the Company:

AdaptHealth Holdings, LLC 122 Mill Road, Suite A130 Phoenixville, Pennsylvania 19460 Attention: Luke McGee Email: luke.mcgee@adapthealth.com

with a copy to:

Willkie Farr & Gallagher LLP 787 Seventh Avenue New York, New York 10019 Attention: Steven J. Gartner; Michael E. Brandt; and Danielle Scalzo Facsimile: (212) 728-8111 Email: sgartner@willkie.com; mbrandt@willkie.com; and DScalzo@willkie.com

- 24 -

(k) <u>Mutual Waiver of Jury Trial</u>. As a specifically bargained inducement for each of the parties to enter into this Agreement (with each party having had opportunity to consult counsel), each party hereto expressly and irrevocably waives the right to trial by jury in any lawsuit or legal proceeding relating to or arising in any way from this Agreement or the transactions contemplated herein, and any lawsuit or legal proceeding relating to or arising in any way to this Agreement or the transactions contemplated herein shall be tried in a court of competent jurisdiction by a judge sitting without a jury.

(1) <u>No Strict Construction</u>. The parties hereto have participated jointly in the negotiation and drafting of this Agreement. In the event an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as if drafted jointly by the parties hereto, and no presumption or burden of proof shall arise favoring or disfavoring any party by virtue of the authorship of any of the provisions of this Agreement.

IN WITNESS WHEREOF, the parties hereto have executed this Registration Rights Agreement as of the date first written above.

COMPANY:

ADAPTHEALTH HOLDINGS LLC

By: Name:

Title:

[Signature Page to Registration Rights Agreement]

IN WITNESS WHEREOF, the parties hereto have executed this Registration Rights Agreement as of the date first written above.

PUBCO:

ADAPTHEALTH CORP.

By: Name: Title:

[Signature Page to Registration Rights Agreement]

IN WITNESS WHEREOF, the parties hereto have executed this Registration Rights Agreement as of the date first written above.

INVESTORS:

[ONE EQUITY PARTNERS VII, L.P.]

By: Name: Title:

IN WITNESS WHEREOF, the parties hereto have executed this Registration Rights Agreement as of the date first written above.

INVESTORS:

DEERFIELD PRIVA TE DESIGN FUND IV, L.P. By: Deerfield Mgmt IV, L.P., General Partner By: J.E. Flynn Capital IV, LLC, General Partner

By: Name:

Title: Authorized Signatory

2321 CAPITAL LLC

By: Name: Title:

AMPEVLLC

By: Name:

Title:

BLUEMOUNTAIN SUMMIT OPPORTUNITIES FUND II (US) L.P.

By: Name: Title:

BLUE RIVER NJ LLC

By: Name: Title:

CFCP LLC

By:
Name:
Title:

FRESH POND INVESTMENT LLC

By: Name:

Title:

JEDI ENTERPRISES, LLC

By: Name:

LBM DME HOLDINGS LLC

By: Name: Title:

Title:

MAYAID2001 LLC

By: Name: Title:

MCLARTY CAPITAL PARTNERS SBIC, L.P.

By: Name: Title:

OCEAN ROCK NJ LLC

By: Name: Title:

PLAINS CAPITAL LLC

By: Name:

Title:

QUAD CAPITAL, LLC

By: Name: Title:

QUADRANT MANAGEMENT, INC.

By: Name: Title:

VERUS EQUITY HOLDING COMPANY LLC

By: Name:

Title:

VERUS NOTE HOLDING COMPANY LLC

By: Name: Title:

WHEATFIELD LLC

By: Name: Title:

CLIFTON BAY OFFSHORE INVESTMENTS L.P.

By: Name: Title:

BLUEMOUNTAIN FOINA VEN MASTER FUND L.P.

By: Name:

Title:

BMSB L.P.

By: Name: Title:

BLUEMOUNTAIN FURSAN FUND L.P.

By: Name:

Title:

REGISTRATION RIGHTS AGREEMENT JOINDER

The undersigned is executing and delivering this Joinder pursuant to the Registration Rights Agreement dated as of (as the same may hereafter be amended, the "Registration Rights Agreement"), among AdaptHealth Holdings Corporation, a Delaware corporation, AdaptHealth Holdings LLC, a Delaware limited liability company (the "<u>Company</u>"), and the other persons named as parties therein.

By executing and delivering this Joinder to Pubco, the undersigned hereby agrees to become a party to, to be bound by, and to comply with the provisions of the Registration Rights Agreement as a holder of Registrable Securities in the same manner as if the undersigned were an original signatory to the Registration Rights Agreement.

Accordingly, the undersigned has executed and delivered this Joinder as of the day of, 20	
	INVESTOR:
	[•]
	By:
	Address for Notices:
	[•] [•] [•]
	Agreed and Accepted as of
	ADAPTHEALTH HOLDINGS LLC
	By:

PUT/CALL OPTION AND CONSENT AGREEMENT

This Put/Call Option and Consent Agreement (this "*Agreement*") is made and entered into May 25, 2020, by and between AdaptHealth Corp., a Delaware corporation ("*Pubco*"), AdaptHealth Holdings LLC, a Delaware limited liability company (the "*Company*"), and the stockholders of Pubco or noteholders of the Company, as applicable, listed on the signature pages hereto (each, a "*Holder*" and collectively, the "*Holders*"). Pubco, the Company and the Holders are each referred to as a "*Party*" and, collectively, they are sometimes referred to as the "*Parties*."

RECITALS

WHEREAS, the Company, a subsidiary of Pubco, issued (a) those certain Promissory Notes, dated November 8, 2019, in favor of each of BMSB L.P., BlueMountain Fursan Fund L.P., BlueMountain Foinaven Master Fund L.P. and BlueMountain Summit Opportunities Fund II (US) L.P. and (b) those certain Amended and Restated Promissory Notes, dated March 20, 2019, in favor of each of BlueMountain Summit Opportunities Fund II (US) L.P. and (b) those certain Amended and Restated Promissory Notes, dated March 20, 2019, in favor of each of BlueMountain Summit Opportunities Fund II (US) L.P., BlueMountain Foinaven Master Fund L.P., BlueMountain Fursan Fund L.P. and BMSB L.P. (the recipients of such Promissory Notes, collectively, the "*Noteholders*");

WHEREAS, the Company and certain other parties are party to that certain Fifth Amended and Restated Limited Liability Company Agreement of AdaptHealth LLC, dated as of November 8, 2019 (the "*LLC Agreement*");

WHEREAS, (a) BlueMountain Foinaven Master Fund L.P., BMSB L.P. and BlueMountain Fursan Fund L.P. (collectively, the "*Class A Stockholders*") collectively own 3,647,353 shares of Class A Common Stock, \$0.0001 par value per share ("*Class A Common Stock*"), of Pubco and (b) BlueMountain Summit Opportunities Fund II (US) LP (together with the Class A Stockholders, the "*Stockholders*) own 150,581 Common Units of the Company and 150,581 Class B Common Stock of Pubco (collectively, the "*Consideration Units*", and together with the Class A Common Stock, the "*Interests*");

WHEREAS, Pubco, the Company and the Stockholders are party to that certain letter agreement with respect to certain limitations on transfer of the Interests, dated July 8, 2019 (the "Lock Up Agreement");

WHEREAS, Pubco, the Company and the Stockholders are party to that letter agreement with respect to certain board designation rights, dated November 8, 2019 (the "Board Designation Letter");

WHEREAS, Pubco and the Company intend to enter into that certain Stock Purchase Agreement and Agreement and Plan of Merger (the "*Purchase Agreement*"), by and among Pubco, the Company, Eleanor Merger Sub LLC ("*Merger Sub*"), Solara Holdings, LLC ("*Solara*") and LCP Solara Blocker Seller, LLC, in its capacity as Blocker Seller and the Representative (in each case as defined therein), pursuant to which Pubco will purchase 100% of the issued and outstanding equity interests of LCP Solara Blocker Corp. (the "*Blocker Purchase*") and Solara will merge with and into Merger Sub, with Solara as the surviving company (the "*Merger*"); WHEREAS, in connection with the entry into the Purchase Agreement, Pubco intends to enter into that certain Investment Agreement (the "*Investment Agreement*") on or about the date hereof with OEP AHCO Investment Holdings, LLC, a Delaware limited liability company (the "*Purchaser*"), and, solely for purposes of Section 3.10 thereof, One Equity Partners VII, L.P., a Delaware limited partnership, pursuant to which Pubco will issue to the Purchaser shares of Class A Common Stock and shares of a new series of preferred stock designated as "Series A Convertible Preferred Stock," par value \$0.0001 per share, of Pubco ("*Series A Preferred Stock*");

WHEREAS, pursuant to Section 8(f)(i) of the Certificate of Designation, in the form attached to Annex I of the Investment Agreement, to be filed with the State of Delaware setting forth the rights, terms and conditions of the Series A Preferred Stock (the "*Certificate of Designation*"), the Series A Preferred Stock will be convertible into Class A Common Stock in an amount up to the Share Cap (as defined in the Certificate of Designation), provided that the Purchaser may, subject to certain conditions and limitations set forth in the Certificate of Designation, request the conversion of Series A Preferred Stock in excess of the Share Cap in exchange for cash rather than shares of Class A Common Stock, in which case Pubco will be required to repurchase such shares of Series A Preferred Stock offered for conversion in excess of the Share Cap on the terms and conditions set forth in such Section 8(f)(i);

WHEREAS, the terms of Sections 2.9(g) and 2.10 of the Fifth Amended and Restated Liability Company Agreement of the Company require, among other things, that the Company undertake all actions to maintain at all times a one-to-one ratio between the number of Common Units owned by Pubco and the number of outstanding shares of Common Stock of Pubco;

WHEREAS, Section 3(g)(ii) of each of the BM Notes prohibits the repurchase by the Company of any of its equity interests without the prior written consent of the respective Noteholder;

WHEREAS, Section 3(g)(viii) of each of the BM Notes generally prohibits the acquisition by the Company (including through any of its subsidiaries or affiliates) of any material portion of the assets or stock of any person involving consideration in excess of \$175 million per annun; and

WHEREAS, the Parties desire to grant certain put rights, call rights and consents related to the aforementioned agreements.

NOW THEREFORE, for and in consideration of the Transactions contemplated in the Purchase Agreement and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties agree as follows:

AGREEMENT

ARTICLE I DEFINITIONS

Capitalized terms used but not defined herein shall have the respective meanings ascribed to such terms in the LLC Agreement.

(a) "Business Day" means any day except Saturday, Sunday and any day which shall be a legal holiday or a day on which banking institutions in the State of New York or State of Pennsylvania generally are authorized or required by law or other governmental action to close.

(b) "*Current Market Price*" per share of Class A Common Stock, as of any date of determination, means the volume-weighted average price per share of Class A Common Stock for the thirty (30) consecutive full Trading Day period ending on, and including, the Trading Day immediately preceding such day, appropriately adjusted to take into account the occurrence during such period of any stock split, reverse stock split, stock dividend, reorganization or similar event with regard to the Class A Common Stock, as reported by NASDAQ (or, if NASDAQ ceases to publish such price, any successor service reasonably chosen by Pubco), or if such information is unavailable, the market price of one (1) share of Class A Common Stock on such Trading Day determined, using a volume-weighted average method, as reasonably determined by the Board of Directors of Pubco.

(c) "*Lien*" means any mortgage, deed of trust, pledge, hypothecation, security interest, encumbrance, claim, escheat, encroachment, lien, charge of any kind, option, easement, purchase right, right of first refusal, right of pre-emption, conditional sale agreement, covenant, condition or other similar restriction (including restrictions on transfer) or any agreement to create any of the foregoing, other than restrictions under securities laws, the LLC Agreement, the Lock Up Agreement or the Voting Agreements.

(d) "*Market Disruption Event*" means, with respect to any date, the occurrence or existence, during the one-half hour period ending at the scheduled close of trading on such date on the principal U.S. national or regional securities exchange or other market on which the Class A Common Stock is listed for trading or trades (or for purposes of determining the Current Market Price per share of Class A Common Stock, any period or periods aggregating one half-hour or longer during the regular trading session on the relevant day), of any material suspension or limitation imposed on trading (by reason of movements in price exceeding limits permitted by the relevant exchange or otherwise) in the Class A Common Stock or in any options contracts or futures contracts relating to the Class A Common Stock.

(e) "NASDAQ" means the NASDAQ Stock Market (or its successor).

(f) "Trading Day" means a day on which the NASDAQ is open for the transaction of business and on which there has not occurred a Market Disruption Event.

(g) "Voting Agreements" mean those certain Voting Agreements, dated as of the date hereof, by and between each of the Stockholders, on the one hand, and the Stockholders, on the other hand.

ARTICLE II CALL OPTION

Section 2.1 <u>Call Option</u>. Subject to <u>Section 2.3</u> and <u>Section 5.1</u> hereof, on or after the consummation of the Closing (as defined in the Purchase Agreement) of the Blocker Purchase and the Merger until 6:00 p.m. Eastern Time on October 31, 2020 (the "*Option Period*"), Pubco shall have the right, but not the obligation, to exercise a one-time option to purchase up to (i) 1,898,967 of the Interests (as adjusted to take into account the occurrence during such period of any stock split, reverse stock split, stock dividend, reorganization or similar event) held by the Stockholders *minus* (ii) the number of Put Interests subject to a Put Option Exercise Notice previously delivered by the Stockholders (subject to the last sentence of Section 5.1) (the "*Call Option*") at a price per share of Class A Common Stock or Consideration Unit of \$15.76 (the "*Call Option Exercise Price*").

Section 2.2 <u>Call Option Exercise Notice</u>. In order to exercise the Call Option, Pubco shall send by written notice during the Option Period a completed irrevocable exercise notice in the applicable form attached hereto as Exhibit A to the Stockholders (the "*Call Option Exercise Notice*") specifying the number of Interests to be purchased by Pubco pursuant to such exercise of the Call Option (such interests, the "*Called Interests*").

Section 2.3 <u>Call Option Mechanics</u>. In connection with the exercise of the Call Option, the Stockholders shall determine, in their sole discretion, the allocation of the Called Interests among the Stockholders and their respective shares of Class A Common Stock and Consideration Units.

ARTICLE III PUT OPTION

Section 3.1 <u>Put Option</u>. Subject to <u>Section 5.1</u> hereof, during the Option Period, the Stockholders shall have the right, but not the obligation, to exercise a one-time option to cause Pubco to purchase up to (i) 1,898,967 of the Interests (as adjusted to take into account the occurrence during such period of any stock split, reverse stock split, stock dividend, reorganization or similar event) held by the Stockholders *minus* (ii) the number of Called Interests subject to a Call Option Exercise Notice previously delivered by Pubco (subject to the last sentence of Section 5.1) (the "*Put Option*") at a price per share of Class A Common Stock or Consideration Unit equal to the greater of (x) \$14.50 and (y) 85% of the Current Market Price on the date the Put Exercise Notice is delivered (the "*Put Option Exercise Price*").

Section 3.2 <u>Put Option Exercise Notice</u>. In order to exercise the Put Option, the Stockholders shall send, by written notice during the Option Period a completed irrevocable exercise notice in the applicable formattached hereto as Exhibit A to the Stockholders (the "*Put Option Exercise Notice*") specifying the number of Interests to be purchased by Pubco pursuant to such exercise of the Put Option (such Interests, the "*Put Interests*").

ARTICLE IV CONSENT TO TRANSACTIONS AND PIK ELECTION

Section 4.1 <u>Consent to Repurchases</u>. Each of the Noteholders hereby waives the restrictions set forth in Section 3(g)(ii) of each of the BM Notes to permit, and hereby consents and for all other purposes under the BM Note to, the repurchase and/or redemption by the Company of a corresponding number of units in the Company held by Pubco upon the repurchase from time to time by Pubco of shares of Series A Preferred Stock delivered for conversion by the Purchaser in excess of the Share Cap as required by the Certificate of Designation, to the extent necessary to permit the Company to comply with the terms of Section 2.9(g) and 2.10 of the LLC Agreement to maintain a one-to-one ratio between the number of Common Units owned by Pubco and the number of outstanding shares of Common Stock of Pubco.

Section 4.2 <u>Consent to Acquisition</u>. Each of the undersigned Noteholders hereby waives the restrictions set forth in Section 3(g)(viii), of each of the BM Notes to permit, and hereby consents for all other purposes under the BM Note to, the entry into and the consummation of the transactions contemplated by the Purchase Agreement, including, without limitation, the Blocker Purchase and the Merger. For the avoidance of doubt, this letter shall constitute a waiver and consent with respect to such transactions contemplated by the Purchase Agreement for the purposes of Section 3(e) of each of the BM Notes.

Section 4.3 Option to Pay PIK in Cash. Pubco hereby irrevocably elects, pursuant to the last sentence of Section 2(b)(ii) of each BM Note, to pay in cash the PIK Interest (as defined in each such BM Notes) payable on any date following the consummation of the Closing (as defined in the Purchase Agreement) of the Blocker Purchase and the Merger.

Section 4.4 <u>No Other Waivers</u>. Except as expressly waived or consented to hereby, all terms, conditions and restrictions set forth in the BM Notes shall remain in full force and effect in accordance with their terms.

ARTICLE V COVENANTS AND AGREEMENTS

Section 5.1 <u>Call/Put Option Procedures</u>. Any Put Option or Call Option shall be settled, and the proceeds paid to the relevant Stockholders, no later than fifteen (15) Business Days following the delivery of the Put Option Exercise Notice or Call Option Exercise Notice, as applicable (such date of payment, the "*Payment Date*"). The Interests over which the Put Option or Call Option is exercised shall be delivered on the Payment Date to Pubco or its designee free and clear of all Liens, together with such stock powers, stock certificates or evidence of transfer as reasonably requested by Pubco, in exchange for the Put Option Exercise Price or Call Option Exercise Price, as applicable, payable by wire transfer of immediately available funds to the account or accounts designated by the Stockholders to Pubco at least two (2) Business Days prior to the Payment Date. If the Company delivers a Call Option Exercise Notice within two (2) calendar days of the delivered and received first, regardless of the actual time when it was received relative to the Put Option Exercise Notice, and the Put Option Exercise Notice will be treated as if it was delivered and received first, regardless of the actual time when it was received relative to the Put Option Exercise Notice, and the Put Option Exercise Notice will be null and void with regard to the number of Interests covered by the Call Option Exercise Notice.

Section 5.2 <u>Restrictions on Disposition</u>. Subject to Transfers in accordance with <u>Section 5.4</u> or the call rights or put rights set forth in <u>Articles II</u> and III hereto, and without limiting the obligations of the Stockholders pursuant to the terms of the Voting Agreements, no Stockholder shall sell, assign, transfer, give, encumber, pledge or in any other way dispose of any of the Interests (any such act is referred to herein as a "*Transfer*"), other than Interests held by the Stockholders in excess of the number of Interests that are subject to the Call Option and Put Option. Subject to the terms of this Agreement, the Stockholders shall be entitled to exercise all rights of ownership of their Interests.

Section 5.3 <u>Transfer Matters.</u> With respect to the BM Notes, the LLC Agreement and the Lock Up Agreement, Pubco and the Company hereby consent to the future transfer of any of the BM Notes or the Interests to any of the Stockholder's or Noteholder's, as the case may be, affiliates, equityholders or other fund(s) or entities managed by BlueMountain Capital Management, LLC, subject to the other provisions of such agreements. Pubco and the Company further acknowledge and agree that the rights under the Board Designation Letter can be transferred to a transferee of the BM Notes pursuant to the foregoing and Section 7(h) of each BM Note.

Section 5.4 Expressly Permitted Transfers.

(a) From the date hereof through the earlier to occur of (i) the termination of this Agreement and (ii) the expiration of the Option Period at 6:00 p.m. on October 31, 2020 (the earlier to occur, the "*Transfer Restriction Expiration Time*", a Stockholder may Transfer any of its Interests or any interest therein (A) to any of Stockholder's Affiliates or equityholders, or other funds or entities managed by BlueMountain Capital Management, LLC or any of its Affiliates, in each case, that execute a joinder to this Agreement in a form reasonably acceptable to Pubco, or (B) to any other person provided that Pubco has consented in writing to such Transfer, which consent can be withheld at the sole discretion of Pubco (any such transfere pursuant to this <u>Section 5.3(a)</u> shall be referred to herein as a "*Permitted Transferee*"). Any Transfer made pursuant to this <u>Section 5.3(a)</u> shall be effective only if such Permitted Transferee shall agree in writing to be bound by the terms and conditions of this Agreement.

(b) In the event a Transfer of any Interests has taken place in violation of the provisions of this Agreement, such Transfer shall be void and of no effect, and the voting and economic rights of such Interests on any matter whatsoever shall remain vested in the transferor.

Section 5.5 Stock Legend.

(a) From the date hereof until the Transfer Restriction Expiration Time, any stock certificates representing the Interests (other than any certificates representing shares of Class A Common Stock held through one or more global stock certificates by The Depository Trust & Clearing Corporation or its nominee) shall be revised to contain the following legend:

THIS CERTIFICATE IS TRANSFERABLE ONLY UPON COMPLIANCE WITH AND SUBJECT TO THE PROVISIONS OF THE PUT/CALL OPTION AGREEMENT DATED AS OF MAY 25, 2020, BY AND AMONG ADAPTHEALTH CORP. (THE "COMPANY"), THE STOCKHOLDERS SIGNATORY THERETO AND CERTAIN OTHER PARTIES THERETO. ANY TRANSFER WITHOUT SUCH COMPLIANCE AND APPROVAL SHALL BE VOID AND OF NO EFFECT. A COPY OF THE PUT/CALL OPTION AGREEMENT IS ON FILE IN THE OFFICE OF THE SECRETARY OF THE COMPANY AT ITS PRINCIPAL PLACE OF BUSINESS. THE COMPANY WILL FURNISH A COPY OF SUCH AGREEMENT TO THE RECORD HOLDER OF THIS CERTIFICATE, WITHOUT CHARGE, UPON WRITTEN REQUEST TO THE COMPANY AT ITS PRINCIPAL PLACE OF BUSINESS OR REGISTERED OFFICE.

(b) Pubco and the Company shall cause the legend set forth in <u>Section 5.5(a)</u> hereof to be removed from any and all stock certificates or book-entries representing any of the Interests no later than the Business Day immediately following the Transfer Restriction Expiration Time.

Section 5.6 <u>Purchase and Sale of Common Stock Upon Bankruptcy or Involuntary Transfer</u>. In the event a Stockholder (i) voluntarily or involuntarily files for bankruptcy in any court of competent jurisdiction, or (ii) has its Interests Transferred by operation of law or otherwise involuntarily, unless provided to the contrary in a separate agreement between Pubco and such Stockholder, upon written notice to such Stockholder, Pubco shall have the continuing option to exercise its Call Option with respect to the Interests beneficially owned by such Stockholder in accordance with <u>Article II</u> hereto.

ARTICLE VI REPRESENTATIONS AND WARRANTIES

Section 6.1 <u>Mutual Representations and Warranties</u>. Each Party represents and warrants to the other parties hereto as follows:

(a) it has all requisite power and authority to enter into this Agreement and to consummate the transactions contemplated hereby.

(b) the execution and delivery of this Agreement and the consummation of the transactions contemplated hereby have been duly authorized by all necessary action on the part of such Party and no other proceedings on the part of such Party are necessary therefor.

(c) this Agreement has been duly and validly executed and delivered by it and is valid, binding, and enforceable in accordance with its terms (subject to (i) the effects of bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium or other similar laws affecting creditors' rights generally and (ii) general equitable principles (whether considered in a proceeding in equity or at law)).

(d) the execution, delivery and performance by such Party of this Agreement does not and will not violate the organizational documents of such Party or any applicable law or any other agreement to which it is bound.

Section 6.2 <u>Pubco Representations and Warranties</u>. Prior to the entry into this Agreement, the board of directors of Pubco has approved, for all purposes of Section 16 of the Exchange Act and Rule 16b-3 thereunder, the acquisition, disposition and exercise by BlueMountain Capital Management, LLC ("BMCM") and each of the Stockholders of the Call Option and the Put Option, as well as the acquisition and disposition of any shares of Class A Common Stock or Class B Common Stock by BMCM or any of the Stockholders in connection with the acquisition, disposition or exercise of the Call Option or the Put Option, in each case on the terms and conditions set forth in this Agreement.

ARTICLE VII MISCELLANEOUS

Section 7.1 <u>Further Assurances</u>. The Parties agree (a) to furnish upon request to each other such further information, (b) to execute and deliver to each other such other documents, and (c) to do such other acts and things, all as the other Party may reasonably request for the purpose of carrying out the intent of this Agreement and the documents referred to in this Agreement.

Section 7.2 <u>Termination</u>. This Agreement shall terminate upon the date on which the Purchase Agreement is terminated in accordance with its terms prior to the Closing (as defined therein).

Section 7.3 <u>Amendments and Waivers</u>. No amendment or waiver of any provision of this Agreement shall be valid unless in writing and signed by the Party to be charged with such amendment or waiver. No waiver by any Party of any default, misrepresentation, or breach of warranty or covenant hereunder, whether intentional or not, shall be deemed to extend to any prior or subsequent default, misrepresentation, or breach of warranty or covenant hereunder or affect in any way any rights arising by virtue of any prior or subsequent such occurrence.

Section 7.4 Notices. All notices permitted or required to be given pursuant to this Agreement shall be given (and will be deemed to have been duly given, if given) by hand delivery, courier service, facsimile, email, or mailed by registered or certified mail, postage prepaid, return receipt requested:

If to the Holders:

BlueMountain Capital Management, LLC 280 Park Avenue, 12th Floor New York, NY 10017 Attn: General Counsel Email: legalnotices@bmcm.com

with a copy (which shall not constitute notice) to:

King & Spalding, LLP 1180 Peachtree Street Atlanta, GA 30309 Attn: Raymond E. Baltz, Jr. Email: rbaltz@kslaw.com

If to the Company or Pubco:

AdaptHealth Corp. 220 West Germantown Pike Suite 250 Plymouth Meeting, PA 19462 Attention: General Counsel E-mail: cjoyce@adapthealth.com

with a copy (which shall not constitute notice) to:

Willkie Farr & Gallagher LLP 787 Seventh Avenue New York, NY 10019 Attention: Steven J. Gartner Michael E. Brandt Danielle Scalzo E-mail: sgartner@willkie.com mbrandt@willkie.com dscalzo@willkie.com Facsimile: 212-728-9962

Notice given by personal delivery, courier service or mail shall be effective upon actual receipt. Notice given by facsimile shall be confirmed by appropriate answer back and shall be effective upon actual receipt if receipt is received during the recipient's normal business hours, or at the beginning of the recipient's next Business Day after if not received during the recipient's normal business hours. Any party may change any address to which notice is to be given to it by giving notice as provided above of such change of address.

Section 7.5 Entire Agreement. This Agreement supersedes all prior agreements between the Parties with respect to its subject matter and constitutes (along with the documents referred to in this Agreement) a complete and exclusive statement of the terms of the agreement between the Parties with respect to its subject matter.

Section 7.6 <u>Assignments, Successors, and No Third-Party Rights</u>. Subject to <u>Section 5.4</u>, no Party may assign any of its rights under this Agreement without the prior consent of the other Parties. Subject to the preceding sentence, this Agreement will apply to, be binding in all respects upon, and inure to the benefit of the successors and permitted assigns of the Parties. Nothing expressed or referred to in this Agreement will be construed to give any person or entity other than the Parties to this Agreement any legal or equitable right, remedy or claim under or with respect to this Agreement or any provision of this Agreement.

Section 7.7 Specific Performance. The Parties recognize and agree that immediate irreparable damages for which there is not adequate remedy at law would occur in the event that the provisions of this Agreement are not performed in accordance with the specific terms hereof or are otherwise breached. It is accordingly agreed that in the event of a failure by a Party to perform his, her or its obligations under this Agreement, each of the other Parties shall be entitled to specific performance through injunctive relief, without the necessity of posting a bond, to prevent breaches of the provisions and to enforce specifically the provisions of this Agreement, in addition to any other remedy to which such Party may be entitled, at law or in equity.

Section 7.8 <u>Consent To Jurisdiction; Governing Law</u>. This Agreement will be governed by and construed in accordance with the laws of the State of Delaware. The Parties hereby irrevocably and unconditionally consent to submit to the exclusive jurisdiction of the state and federal courts located in the State of Delaware for any actions, suits or proceedings arising out of or relating to this Agreement and the transactions contemplated hereby. The Parties hereby irrevocably and unconditionally consent to the jurisdiction of such courts (and of the appropriate appellate courts therefrom) in any such action, suit or proceeding and irrevocably waive, to the fullest extent permitted by law, any objection that they may now or hereafter have to the laying of the venue of any such action, suit or proceeding in any such court or that any such action, suit or proceeding may be served on any Party anywhere in the world, whether within or without the jurisdiction of any such court. Without limiting the foregoing, each Party agrees that service of process on such Party as provided in Section 7.8 shall be deemed effective service of process on such Party.

Section 7.9 <u>WAIVER OF JURY TRIAL</u> EACH OF THE PARTIES HERETO HEREBY IRREVOCABLY WAIVES ANY AND ALL RIGHT TO TRIAL BY JURY IN ANY LEGAL PROCEEDING ARISING OUT OF OR RELATED TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY.

Section 7.10 <u>Severability</u>. Any term of this Agreement which would be invalid or unenforceable as written shall be deemed limited in scope and/or duration to the extent necessary to render it enforceable. The determination of any court that any provision is invalid or unenforceable shall not affect the validity or enforceability of the remaining terms and provisions hereof or the validity of the offending term or provision in any other situation or in any other jurisdiction.

Section 7.11 <u>Construction</u>. Wherever the context so permits, the use of words in this Agreement in the masculine, feminine or neuter gender shall be construed to include all of such genders. All references to articles, sections, subsections, or subparagraphs are to provisions of this Agreement unless the context dictates otherwise.

Section 7.12 <u>Successors</u>. All of the terms, agreements, covenants and conditions of this Agreement are binding upon, and inure to the benefit of and are enforceable by, the Parties and their respective successors and permitted assigns. Neither this Agreement, nor any of the rights, interests or obligations hereunder shall be assigned by any of the Parties (whether by operation of law or otherwise) without the prior written consent of the other Parties;.

Section 7.13 <u>Delivery by PDF and Facsimile</u>. This Agreement and any amendments hereto, to the extent signed and delivered by means of portable document format ("**PDF**") or a facsimile machine, shall be treated in all manner and respects as an original contract and shall be considered to have the same binding legal effects as if it were the original signed version thereof delivered in person. At the request of any Party hereto, each other Party hereto shall re-execute original forms thereof and deliver them to all other Parties. No Party hereto shall raise the use of PDF or a facsimile machine to deliver a signature or the fact that any signature or this Agreement was transmitted or communicated through the use of PDF or a facsimile machine as a defense to the formation of a contract and each such Party forever waives any such defense.

[Signature Pages Follow]

IN WITNESS WHEREOF, the Parties have executed this Agreement as of the date first above written.

ADAPTHEALTH CORP.

By: /s/ Luke McGee Name: Luke McGee Title: Authorized Signatory

ADAPTHEALTH HOLDINGS LLC

By: /s/ Luke McGee

Name: Luke McGee Title: Chief Executive Officer

BLUEMOUNTAIN FOINA VEN MASTER FUND L.P.

By: /s/ Richard Horne Name: Richard Horne Title: Deputy General Counsel, Tax

BMSB L.P.

By: /s/ Richard Horne Name: Richard Horne Title: Deputy General Counsel, Tax

BLUEMOUNTAIN FURSAN FUND L.P.

By: /s/ Richard Horne Name: Richard Horne Title: Deputy General Counsel, Tax

BLUE MOUNTAIN SUMMIT OPPORTUNITIES FUND II (US) L.P.

By: /s/ Richard Horne Name: Richard Horne Title: Deputy General Counsel, Tax

If exercise of Put Option:

NOTICE OF EXERCISE OF PUT OPTION

To AdaptHealth Corp.:

Date:

The undersigned Stockholders hereby irrevocably exercise their option to cause Pubco to purchase ______ of the Interests held by the Stockholders, in accordance with Article III of the Put/Call Option and Consent Agreement dated May 25, 2020. The Interests to be sold to Pubco shall allocated as follows:

Stockholder	Class A Common Shares	Consideration Units
BMSB L.P.		
BlueMountain Fursan Fund L.P.		
BlueMountain Foinaven Master Fund L.P.		
BlueMountain Summit Opportunities Fund II (US) L.P.		

Stockholders:

By:	
Name:	
Title:	



If exercise of Call Option:

NOTICE OF EXERCISE OF CALL OPTION

To: Stockholders

Date: _____

AdaptHealth Corp., a Delaware corporation, hereby irrevocably gives notice to the Stockholders of the exercise of its option to purchase ______ of the Interests held by the Stockholders at a price per Interest of \$15.76 and an aggregate price of \$______, in accordance with Article II of the Put/Call Option and Consent Agreement dated May 25, 2020.

ADAPTHEALTH CORP. By:

Ladies and Gentlemen:

Reference is made to (i) that certain Investment Agreement (the "Third Party Investment Agreement"), dated as of the date hereof, between AdaptHealth Corp. (the "Company"), OEP AHCO Investment Holdings, LLC ("OEP Vehicle") and, for the purposes of Section 3.10 thereof, One Equity Partners VII, L.P. ("OEP Fund" and, together with OEP Vehicle, "One Equity"), (ii) the form of Certificate of Designations (as defined in the Third Party Investment Agreement) attached as an exhibit to the Third Party Investment Agreement and (iii) the Voting Agreement (the "Voting Agreement"), dated as of the date hereof, between the Company and Deerfield Private Design Fund IV, L.P. ("DPDIV"). For good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, DPDIV, Deerfield Partners, L.P. ("Deerfield Partners", and together with DPDIV, "Deerfield") and the Company hereby agree as follows:

1. Exchange. From and after the date hereof, each of DPDIV and the Company shall use their reasonable best efforts to, as soon as practicable after the date hereof and in any event within thirty (30) days of the date hereof, (i) negotiate and mutually agree to a Series B-1 Certificate of Designation, Preferences and Rights on substantially the terms and conditions set forth in the summary of terms attached hereto as Exhibit A (the "Series B-1 Certificate of Designation" and the shares of capital stock issuable thereunder, "Series B-1 Preferred Stock"), (ii) upon agreement thereof, file the Series B-1 Certificate of Designation with the Secretary of State of the State of Delaware, (iii) negotiate and enter into an agreement (the "Exchange Agreement") to be entered into between the Company and DPDIV providing for the exchange of all but 4.5% of the Class A Common Stock, par value \$0.0001 per share (the "Class A Common Stock"), beneficially owned by it (the "Exchange Shares") as of the date of the closing of such exchange for Series B-1 Preferred Stock, and (iv) consummate the transactions contemplated by the Exchange Agreement (the "Exchange"). The Company acknowledges and agrees that, to the knowledge of the Company, (i) for purposes of Rule 144 under the Securities Act, DPDIV's holding period for the shares of Series B1 Preferred Stock issued to it in the Exchange, and any shares of Class A Common Stock issued upon the conversion thereof, shall be deemed to have commenced on the date DPDIV acquired the Exchange Shares from the Company or an affiliate of the Company (or such earlier date as may be permitted pursuant to Rule 144 under the Securities Act); and (ii) the consummation of the Exchange, the issuance of shares of Series B-1 Preferred Stock pursuant to the Exchange and the issuance of shares of Class A Common Stock upon the conversion of the Series B-1 Preferred Stock in full (without giving effect to the conversion of any Series B-2 Preferred Stock into Series B-1 Preferred Stock and otherwise without regard to any limitation on the conversion thereof) shall not require the approval of stockholders under the Certificate of Incorporation or Bylaws, the DGCL, Nasdaq listing rules or otherwise. The Exchange Agreement will provide for the irrevocable surrender of DPDIV's voting rights with respect to the Exchange Shares (including its voting rights in respect of the Company's annual meeting of stockholders). The Company shall use its reasonable best efforts not to fix a record date for any meeting of its stockholders with respect to which a record date has not already been set until the Exchange has become effective, unless the Exchange shall not have become effective as a result of Deerfield's willful breach of this Letter Agreement. No later than the consummation of the Exchange, Section 4(a) of the Registration Rights Agreement (as defined in the Third Party Investment Agreement) shall be amended, modified or waived to provide that:

(i) Deerfield shall not be required to enter into lock-up agreements pursuant to such Section 4(a) on more than two (2) occasions (including any lock-up agreement entered into prior to the execution of such amendment).

(ii) The lock-up agreements to which Deerfield enters into pursuant to such Section 4(a) shall be for a period of not more than sixty (60) days.

(iii) The obligation of Deerfield to enter into lock-up agreements pursuant to such Section 4(a) shall terminate on November 8, 2021.

(iv) Deerfield shall not be required to enter into a lock-up agreement pursuant to such Section 4(a) within six (6) months following the expiration of a previous lock-up agreement pursuant to such Section 4(a).

The Company further acknowledges and agrees that, even prior to the execution of such amendment, modification or waiver to the Registration Rights Agreement, Deerfield shall not be obligated to enter into a lock-up in respect of a period of more than 60 days.

Sale of Shares. Each of Deerfield and the Company shall use their reasonable best efforts to, as soon as practicable after the date hereof and in any event within thirty (30) 2. days of the date hereof, (i) negotiate and mutually agree to a Series B-2 Certificate of Designation, Preferences and Rights (the "Series B-2 Certificate of Designation" and the shares of capital stock issuable thereunder, "Series B-2 Preferred Stock"), which shall be on terms and conditions that are substantially equivalent to those set forth Series A Certificate of Designation (as defined in the Third Party Investment Agreement (as defined below)) and otherwise as agreed between Deerfield and the Company, provided that the shares of Series B-2 Preferred Stock shall be convertible into Series B-1 Preferred Stock (and otherwise be consistent with the terms described below) and (ii) negotiate and enter into an investment agreement (the "Deerfield Investment Agreement") among the Company and Deerfield Partners that (A) provides for the purchase by Deerfield Partners from the Company, and issuance and sale by the Company to Deerfield Partners, of an aggregate of \$35,000,000 of Series B-2 Preferred Stock and at the same purchase price per share as the price per share to be paid by One Equity for a share of Series A Preferred Stock; and (B) otherwise contains representations, warranties, covenants, closing conditions and agreements that are substantially equivalent to those contained in the Third Party Investment Agreement (with appropriate modifications) and otherwise as agreed between Deerfield and the Company, except that (I) it shall provide for the purchase of Series B-2 Preferred Stock (and no Class A Common Stock), (II) it shall provide that the Series B-2 Preferred Stock and any shares of capital stock (including Class A Common Stock issuable upon conversion of Series B-1 Preferred Stock) directly or indirectly issuable upon conversion thereof (collectively, the "Purchased Securities") will be subject to the restrictions on transfer set forth in Section 5.3 of the Third Party Investment Agreement only for a period of 60 days following the closing under the Deerfield Investment Agreement (which restrictions shall not, for the avoidance of doubt, apply to any securities beneficially owned by Deerfield other than the Purchased Securities), (III) it shall include confidentiality provisions and procedures with respect to Company information, (IV) it shall not provide for board observer or board designation rights or any right of first offer and (V) it shall condition the closing on the consumnation of the Exhange, unless the Exchange shall not have become effective as a result of Deerfield's willful breach of this Letter Agreement. The shares of Series B-2 Preferred Stock shall be convertible into shares of Series B-1 Preferred Stock at a rate equal to the result of the conversion rate of the Series A Preferred Stock, divided by the conversion rate of the Series B-1 Preferred Stock.

3. Voting Agreement. The parties hereto acknowledge and agree that they are entering into this Letter Agreement as an inducement to DPDIV's willingness to enter in the Voting Agreement. Without limiting Deerfield's rights to pursue any other remedies available to it hereunder, at law or in equity including, without limitation, the right to pursue a decree of specific performance and/or injunctive relief (which rights are hereby acknowledged), in the event that the parties hereto fail to enter into the Exchange Agreement and consummate the Exchange in accordance with Section 1 hereof, or to enter into the Deerfield Investment Agreement in accordance with Section 2 hereof, in each case other than as a result of Deerfield's breach of this Letter Agreement, then the Voting Agreement shall automatically terminate and be of no further force or effect. In the event that DPDIV materially breaches its obligations under the Voting Agreement, then DPDIV's rights and the Company's obligations under this Letter Agreement shall automatically terminate and be of no further force or effect. The Company hereby confirms that (x) it and/or One Equity have entered into voting agreements with stockholders of the Company representing a majority of the voting power of the outstanding common stock of the Company in respect of the approval of such single matter as shall be necessary to approve the issuance of such number of shares to permit the conversion in full into Class A Common Stock of the Series A Preferred Stock to be issued to One Equity pursuant to the terms of the Third Party Investment Agreement under applicable Nasdaq rules and (y) agrees that it will use its reasonable best efforts to provide for the Company to enter into voting agreements (consistent with the voting agreements entered into with One Equity) with stockholders of the Company representing a majority of the voting power of the outstanding common stock of the Company in respect of the approval of such single matter (less the number of shares with respect to which Deerfield can exercise voting power after giving effect to the Exchange) as shall be necessary to approve the issuance of such number of shares to permit the conversion in full into Class A Common Stock of the the Series B-1 Preferred Stock issuable upon conversion of the Series B-2 Preferred Stock to be issued to Deerfield under the Deerfield Investment Agreement under applicable Nasdaq rules; provided that the entry into such voting agreements shall be a condition to Deerfield's obligation to enter into the Deerfield Investment Agreement.

4. No Conflicts.

- 4. 1. Deerfield represents and warrants to the Company that neither the execution and delivery by Deerfield of this Letter Agreement, nor the consummation of the transactions contemplated hereby, nor compliance by Deerfield with any of the provisions hereof or thereof will (a) violate or conflict with the organizational documents of Deerfield, (b) conflict with or violate any law applicable to Deerfield or by which any of its properties or assets is bound or subject or (c) result in any breach of, or constitute a default (or event which, with the giving of notice or lapse of time or both, would constitute a default) under, or give to any person any rights of termination, acceleration or cancellation of or result in the creation of any lien on any of the assets or properties of Deerfield, any note, bond, mortgage, indenture, deed of trust, license, lease, agreement or other instrument or obligation to which Deerfield or any of its subsidiaries is a party or by which any of them or any of their respective properties or assets is bound or subject, except, in the case of clauses (b) and (c), for any such conflicts, violations, breaches, de-faults, terminations, accelerations, cancellations or creations as, individually or in the aggregate, would not reasonably be expected to have a material impairment or material delay in the ability of Deerfield to perform its material obligations under this Agreement or to consummate the transactions contemplated by this Agreement.
- 4.2. The Company represents and warrants to Deerfield that neither the execution and delivery by the Company of this Letter Agreement, nor the consummation of the transactions contemplated hereby, nor compliance by the Company with any of the provisions hereof or thereof will (a) violate or conflict with the organizational documents of the Company, (b) conflict with or violate any law applicable to the Company or by which any of its properties or assets is bound or subject or (c) result in any breach of, or constitute a default (or event which, with the giving of notice or lapse of time or both, would constitute a default) under, or give to any person any rights of termination, acceleration or cancellation of or result in the creation of any lien on any of the assets or properties of the Company, any note, bond, mortgage, indenture, deed of trust, license, lease, agreement or other instrument or obligation to which the Company or any of its subsidiaries is a party or by which any of them or any of their respective properties or cascets is bound or subject, except, in the case of clauses (b) and (c), for any such conflicts, violations, breaches, de-faults, terminations, accelerations or creations as, individually or in the aggregate, would not reasonably be expected to have a Company Material Adverse Effect (as defined in the Third Party Investment Agreement). The execution and delivery of this Letter Agreement and the issuance (directly or indirectly) of securities as contemplated hereby is not, and will not be, subject to, or trigger, any preemptive rights, rights of first offer, notice rights, approval/consent rights, voting rights, review rights or similar rights of any third party and will not trigger any anti-dilution rights, except for rights of the type contemplated in the Third Party Investment. No consents of One Equity, other than those that have already been obtained, are necessary for the consummation of the transactions contemplated by this Letter Agreement in accordance with the terms hereof.</

5. Expenses. The Company shall, regardless of whether any transaction contemplated hereby is consummated, reimburse Deerfield for its reasonable and documented out-of-pocket third-party costs and expenses (including legal fees) incurred in connection with due diligence, the negotiation and preparation of this Letter Agreement, the Exchange Agreement and the Deerfield Investment Agreement, including the negotiation and preparation of a draft investment agreement and certificates of designation in connection with Deerfield's proposed investment in Series A Preferred Stock prior to the date hereof, and any other agreement or transaction contemplated hereby or thereby and undertaking of the transactions contemplated pursuant to this Letter Agreement; provided, that such reimbursement obligation shall not exceed \$250,000 in the aggregate. Any such reimbursement shall be made promptly following submission of invoices in respect of the costs and expenses at or following the first to occur of (x) the closing of the transactions contemplated by the Deerfield Investment Agreement, the (y) consummation of the transactions contemplated by the Acquisition Agreement (as defined in the Third Party Investment Agreement) and (z) the termination of the Acquisition Agreement (as defined in the Third Party Investment Agreement).

6. Miscellaneous.

- 6.1. The execution and delivery of this letter agreement (the "Letter Agreement") by the Company and Deerfield is binding on and enforceable against the Company and Deerfield. This Letter Agreement supersedes any other agreement, whether written or oral, that may have been made or entered into by the parties hereto relating solely to the matters contemplated hereby. In the event of any inconsistency between the terms of this Letter Agreement and any other prior agreement relating to the matters addressed herein, the parties agree that the terms of this Letter Agreement shall control. This Letter Agreement may be amended, modified, superseded, or canceled, and any of the terms, representations, warranties or covenants hereof may be waived, only by written instrument executed by all of the parties hereto or, in the case of a waiver, by the party waiving compliance. This Letter Agreement may be executed in two or more counterparts, each of which shall be an original, but all of which together shall constitute one and the same instrument. This Letter Agreement may not be assigned by any party without the written consent of the other party, and shall be binding upon, inure to the benefit of, and may be enforced by, each of the parties to this Letter Agreement and its successors and permitted assigns. This Letter Agreement shall be governed by and construed, interpreted and enforced in accordance with the laws of the State of Delaware, without giving effect to the principles of conflicts of law thereof. Each provision of this Letter Agreement shall be considered separable, and if, for any reason, any provision reprovisions hereof are determined to be invalid, illegal or unenforceable, such invalidity, illegality or unenforceability shall attach only to such provision and shall not in any manner affect or render illegal, invalid or unenforceable any other provision of this Letter Agreement, and this Letter Agreement shall be carried out as if any such illegal, invalid or unenforceable provision of this Letter Agree
- 6.2. This Letter Agreement shall terminate on the one-year anniversary of the date hereof; provided, however, that no such termination shall relieve any party of any breach hereof prior to such termination.
- 6.3. This Letter Agreement, including any schedules, amendments, modifications, waivers, or notifications relating thereto may be executed and delivered by facsimile, electronic mail, or other electronic means. Any such facsimile, electronic mail transmission, or communication via such electronic means shall constitute the final agreement of the parties and conclusive proof of such agreement, and shall be deemed to be in writing and to have the same effect as if signed manually.

[REMAINDER OF PAGE INTENTIONALLY LEFT BLANK; SIGNATURE PAGE FOLLOWS]

Very Truly Yours,

ADAPTHEALTH CORP.

By: /s/ Luke McGee Name: Luke McGee Title: Authorized Signatory

[Signature page to the Side Letter]

Very Truly Yours,

ADAPTHEALTH CORP.

By:			
Name:			
Title:			

Accepted as of the date first above written:

DEERFIELD PRIVATE DESIGN FUND IV, L.P.

By: Deerfield Mgmt IV, L.P., its General Partner By: J.E. Flynn Capital IV, LLC, its General Partner

By: /s/ David Clark Name: David Clark Title: Authorized Signatory

DEERFIELD PARTNERS, L.P.

By: Deerfield Mgmt, L.P., its General Partner By: J.E. Flynn Capital, LLC, its General Partner

By: <u>/s/ David Clark</u> Name: David Clark Title: Authorized Signatory

EXHIBIT A

Series B-1 Convertible Preferred Stock AdaptHealth Corp.

Subject	Summary
Par Value	\$0.0001 per share
Dividends	Pro rata on an as-converted basis with the Class A Common Stock. No other dividends.
Voting Rights	No voting rights, except for the following or as otherwise required by law
	Approval of a majority of the outstanding shares of Series B-1 is required to:
	• Alter rights, powers, preferences, etc. of Series B-1
	Increase authorized shares of Series B-1
	 Amend charter or bylaws in a manner adverse to the rights (other than voting rights) of the Series B-1 relative to the rights of the holders of common stock
	Amend the Series B-1 Certificate of Designation
Liquidation Rank	Senior to all common stock, junior to Series A Preferred and to any other series of preferred stock designated as senior to the Series B-1 Preferred Stock
Liquidation Preference	Preference: \$0.0001 per share, plus any declared but unpaid dividends
	Following payment of the preference, participates ratably with the common stock on an as-converted basis
Optional Conversion	Convertible at the Holder's option at any time into [100] shares of Class A Common Stock per share of Series B-1 Preferred Stock, subject to the Blocker/Beneficial Ownership Limitation
	Class A Common Stock must be delivered within the standard settlement period (currently two (2) trading days) after delivery of a conversion notice, consistent with the provisions in the Series A Preferred Certificate of Designations
Blocker/Beneficial Ownership Limitation	4.9% of the total number of shares of common stock then outstanding

Subject	Summary
Fractional Shares	No fractional shares of Class A Common Stock will be issued upon conversion – fractional shares will be rounded up to the next whole share
Adjustments to Conversion Rate	 In the case of stock dividends, subdivisions of stock, combinations of stock, reclassifications of stock, in each case, with respect to Class A Common Stock, the conversion rate will be multiplied by the following fraction: A/B A = # of shares of Class A Common Stock outstanding immediately after such event B = # of shares of Class A Common Stock outstanding immediately after such event
Fundamental Transactions	In the case of mergers, consolidations, sales of substantially all assets, tender or exchange offers or other fundamental transactions involving the Class A Common Stock, holders of Series B-1 are entitled to receive consideration they would have been entitled to receive if converted to Class A Common Stock immediately prior to such transaction.